

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

THYMIDINE NUCLEOTIDES IN MUSLEIC ACID BIOSTNTHESIS

by

Hans Johan Grav

Thesis presented for the degree of Doctor of Philosophy, The University of Glasgow.

March, 1964

ProQuest Number: 10647359

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10647359

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

Aginion Ledgements

I should like to express my sincere gratitude to Professor J. N. Davidson, F.R.S. for providing the facilities for this research and to Dr. R. M. S. Smellie for his invaluable advice, constant encouragement and patience as supervisor.

I asknowledge with pleasure the helpful discussions with Dr. H. M. Keir, Dr. R. Y. Thomson and Dr. R. H. Burdon and I should like, further to extend my warmest thanks to all members of this department for all useful advice and suggestions so freely given. I should especially like to thank Mr. R. Bason for proof reading and for making many helpful suggestions in the preparation of this manuscript.

I am also indepted to Mr. R. Gallander for the preparation of Figures 17, 22, 23, 24, 25, 26, 27, 32, 34, 35, 37, 38 and 40, to Miss M. Hood for typing and to Mr. G. Russell who rendered expert technical assistance from time to time.

Finally, I must acknowledge with thanks the receipt of a Medical Research Council Assistantship during the early period of this work and the subsequent grant from The British Empire Ocnoer Campaign.

ABBREVIATIONS

The following abbrevia	tions will be employed in this thesis: -
RNA	ribonucleio acid
e-RNA	"soluble" ribonucleic acid
DNA	decryribonueleic acid
UR	uridine
OR	oytidine
AR	adenosine
(RR	guenosine
UMP	uridine 5'-monophosphate
מש	uridine 5'-diphosphete
UTP	uridine 5'-triphosphate
OMP	cytidine 5'-monophosphate
CDP	cytidine 5'-diphosphate
CTP	cytidine 5' -triphosphete
ANTP	edenosiae 5' -monophosphate
S-MP	adenylosuccinic acid
ADP	adenosine 5'-diphosphate
ATP	adenosine 5' -triphosphate
XXP	xenthosine 5' -monophosphete
ОМР	guanosine 5' -monophosphate
ODP	guanosine 5' -diphosphate
OTP	guanosine 5' -triphosphate
IMP	incsine 5'-monophosphete

1

5-He-3HP	5-methyloytidine 5'-monophosphete
IDP	incaine 5'-diphosphete
ITP	incoine 5'-triphosphate
UdR	2'-deczywridine
COR	2'-decaycytidine
AdR	2' -deoxyadenceine
GdR	8' -decxyguenosi ne
AUMP	decxyuridine 5'-monophosphete
OUDP	decxyuridine 5°-diphosphato
AUTP	deoxyuridine 5'-triphesphete
dowe; decay aytidy inte	deoxycytidine 5'-monophosphete
AGDP	deoxycytidine 5'-diphosphate
dotp	decxycytidine 5'-triphosphate
5-He-00119	5-methyldeoxycytidine 5'-monophosphate
5-MeOH-dOMP	5-methylhydroxydeoxydytidine 5'-monophosph
5-NoOH-49DP	5-methylhydroxydeoxycytidine 5'-diphosphet
5-NOOH -40TP	5-methylhydroxydeoxycytidine 5'-triphosphe
dAMP, decayadenylate	decayed encaine 5' -monophesphete
AADP	deoxyndenesine 5' -diphosphate
ATP	deoxyedenosime 5' -triphosphete
dow, decaygunaylate	decayguanosine 5' -monophosphete
GODP	decrygumacine 5' -diphosphate
40TP	decrygumnosine 5' -triphosphate
TaR , shymidine	2'-decayribose thymine
THP, THP-5'	thymidime 5'-monophosphete
THP-3'	thymidine 3'-monophosphete

A STATE OF A	
TDP	thymidine 5'-diphosphate
TTP	thymidime 5'-triphosphete
FRPP	5-phosphoribosyl-l-pyrophosphete
AIGAR	4-emino-5-imidasole carboxamide ribonucleo
PAIGAR	5-fermaino-4-imidasole carboxamide ribo- nucleotide
PRA	5-phosphoribosylemine
GAR	glycinemide ribonucleotide
IGAR	formylglycinamide ribonucleotide
AIR	5-eminoimidesole ribonucleotide
(^a n) Tar	thymidine labelled with tritium
(^S H) GAR	decuy cytidine lebelled with tritium
(^{3R} P) TMP	thymidine 5'-monophosphete labelled with a redicective phosphorus stom
OFF	2-cymoethylphosphete
DOG	di ayclohexylcorbodiimide
DCU	dicyclohexylures
MAD	ni cotinamide -adenine dinucleotide
NADE	reduced nicotinemide-edenine dinucleotide
NADP	micotinamide-adeniae dinucleotide phosphate
NADPH	reduced micotinemide-edenime dimucleotide phosphete
Pi	inorganic orthophosphate
P-P.	inorganic pyrophosphate
tris	2-amino-2-hydroxymethylpropensel, 3-diol
BDTA	sthylenediaminetetra-acetate
A A A A A A A A A A A A A A A A A A A	

Enzymes: Trivial enzyme nomenclature was used throughout this thesis. The term <u>thymidiae end thymidylate kinnses</u> is employed to describe the set of phosphokineses required to phosphorylate thymidine to thymidine-5'-triphosphate. The term <u>thymidylate kinnses(s)</u> refers to phosphokineses empable of phosphorylating thymidine 5'-monophosphate to the products thymidine 5'-di- and thymidine 5'-monophosphate. The term <u>TMP kinnse</u> refers exclusively to the phosphokinese cetalysing the formation of thymidi 5'-diphosphate from thymidine 5'-monophosphate. (For other definitions, see text, Section 3.1).

CONTENTS

ABBREVIATIONS

	The second se	
		I
CHAPTER I.	GINERAL INTRODUCTION	1
1.1. Che	mical and physical structure of the nucleic acids	2
(m)	RNA	8
(b)	DNA	3
1.2. Bic	synthesis of the nucleic scids	7
(a)	Biosynthesis of the purine ribonucleoside mono- phosphates	8
(b)	Biosynthesis of the pyrimidine ribonucleoside monophosphates	13
(0,	Biosynthesis of the pentose moleties	19
(a)	Formation of nucleoside di- and triphosphates	24
(•)	Biosynthesis of ribonucleic acids	26
(1,	Biosynthesis of deoxyribonucleic acids	29
1.3. Ont	abolism of the nucleotides and their substituents	34
1.4. Gor	trol of DMA biomynthemis	38
CHAPTER II	STUDIES ON THE FORMATICN OF THYMIDINE 5'-TRIPHOS- PHATE WITH ENLYMES FROM LANDSCHUTZ ASCITES TUMOUR ORLLS	
2.1. Int	roduction	44
(=)	The existence of cellular pools of free nucleotides	44
(b.	Internucleotide metabolism	46
(0.	Formation of nucleotides by preformed pathways	51
(a	Thymine ribonucleomide and thymidine 5'-diphosphate coenzymes	54
(-	Formation of thymidine aucleotide phosphates	57

2.2.	Expe	rimental	61
1	2.2.1.	Preparation of grude cell-free extracts from Iands chuts as gites tumour cells.	61
8	8.2.2.	Chemical properation of (32P) -labelled thymidine 5'-phosphates	62
1		(a) Preparation of (32p) -labelled thymidine 5'-mono- phosphete	68
		(b) Preparation of thymidine 5'-triphosphate labelled with (³² P) in the proximal position and labelled with (³² P) in the two terminal positions	68
		(c) Preparation of unlabelled thymidine 5'-di- phosphete	72
		(d) Preparation of thymidine 5'-diphosphete labelled with (32p) in the proximal position and labelled with (32p) in the terminal position	73
	2.2.3.	General analytical methods	75
	۹	(a) Batimation of protein	75
	٩.	(b) Shrometography on paper	75
	£,	(c) Location of free or bound phosphate on paper chrometograms	76
	1000	(d) Location of radioactivity on paper chromatograms	77
	4.	(e) Spectrophotometric measurements	77
	(†	(f) Techniques used for the determination of radioactivity	78
;	2.2.4.	Procedures used for the assay of kinese activities	79
		(a) Assny of the kineses catalysing the formation of thymidine 5'-triphosphate from thymidine	79
	•	(b) Determination of specific activities	83
-	2.2.5.	Buffer solutions	85

- (a) Tris buffer 85
- (b) Phosphete buffer 85

2.2.6.	Preparation of adsorbants for column chromatography	85
	(s) Preparation of charcoal adsorbent	85
	(b) Pretreatment of Dower-1-G1" for column chrometo- graphy	87
- Andrews	(c) Preparations of columns of Dower-50-Na ⁺	87
	(d) Preparation of SOTSOLA cellulose for column chromstography	88
8.2.7.	Materials	89
2.3. Reau	lts	90
8.3.1.	Optimal conditions for the formation in vitro of TTP from thymidine by cell-free extracts of Landschutz ascites tumour cells	90
2.3.2.	The mechanism of formation of TTP from thymidine by enzymes from cell-free extracts of Landschutz ascites tumour cells	96
CHAPTER III	• PRELIMINARY ENZYMOLOGICAL STUDIES ON THE THYMIDINE AND THYMIDYLATE KINASES OF LANDSCHUTZ ASCITES TUMOUR CELLS	
3.1. Intr	oduction	105
Gene	ral properties of the nucleoside and nucleotide hosphokineses	105
3.2. Expe	rimental	110
3,2,1.	General methods	110
3.2.2.	Biological material	110
	(a) Extracts from Landschutz escites tumour cells	110
	(b) Extracts from normal rat liver	111
3.2.3.	Procedures used for the assny of enzymes	111
	(a) Thymidine kinese	111

vii

a survey a	(b) Radiosotivity assay for THP kinase	114
A PLAN	(c) Spectrophotometric assay for THP kinase	117
See.	(d) Asony for decrycytidine and decrycytidylete kineses	119
	(e) Phosphatese aseays	182
3.2.4.	Enzyme fractionation techniques	124
	(a) Fractionation with dilute sortic soid	184
	(b) Fractionations using Cy -elumina gel	126
	(c) The use of gel filtration techniques in enzyme fractionations	127
	(d) Enzyme fractionations using ammonium sulphate	131
	(e) Enzyme fractionstions using a cetone	132
.3. Resu	lts	133
3.3.1.	Biological factors affecting the activity of the thymidine and thymidylate kinase system from cell-free extracts of Landschutz ascites tumour	1.00
	00110.	133
3.3.8.	Experiments with the thymidine kinese from cell-free extracts of Landschutz escites tumour cells	137
3.3.3.	Stability of the thymidine and thymidylate kinases of cell-free extracts of Landschutz ascites tumour cells	142
3.3.4.	Experiments with the TMP kinese of cell-free extracts of Lendschutz ascites tumour cells	146
3.3.5.	Partial purification of thymidine 5'-monophosphate kinase (TMP kinase) from cell-free extracts of Landschutz ascites tumour cells	156
	(a) Nethod I	157
	(b) Nethod II	161

SHAFTER IV. GENERAL DISCUSSION

- 4.1. The mechanism of formation of thymidine 5'-triphosphate from thymidine and thymidine 5'-monophosphate by enzymes from Iandschutz escites tumour cells 166
- 4.2. Intracellular location of the thymidine and thymidylate kinases
- 4.3. The possible involvement of the thymidine and thymidylete kinases in the control and regulation of DNA synthesis 192

SUMMARY

210

185

ix

"True windom is not to be gained by measuring out the boundless world..... It is rather to be attained by the thorough investigation of any individual thing for thus we seek to arrive at a full knowledge and understanding of its true and peculiar nature."

Schopenhauer.

CHAPTER I

GENERAL INTRODUCTION

In the 17th and 18th centuries any competent scientist had reasonable expectation that he might adequately cover the whole range of scientific knowledge and experience within a lifetime. The ever widening scope and growing effort expended on science throughout the 19th and into the 20th century has resulted in massive proliferation and subdivision of scientific disciplines so that his present-day counterpart may only aspire to mastery of an infinitesimal part of the scientific knowledge available to him. Hevertheless, the acquisition of a unified vision of the natural processes thus made more difficult for the individual remains as desirable to-day as it was 200 years ago, but the restrictions imposed by rigid and marrowly based specialization allied to failure of communication between practitioners of related sciences has too often led to a frustration of these efforts. In retrospect, it is not difficult to perceive that the remedy lay in the removal of artificially created interdisciplinary barriers thus giving due recognition to the fundamental principle of the interdependence of all scientific disciplines. Nowhere have the advantages of co-operation and concerted effort been better illustrated than within the biological sciences and, in particular, in the progress of the research on the nucleic acids over the last decade. Indeed, such vest progress had been inconceivable without the perticipation of groups of scientists having at their command a wide array of disciplines ranging from physics and chemistry through biochemistry to genetics, all seeking differing primary objectives but simultaneously channelling their efforts into a quest for the true nature of "the genetic material".

- 1 -

Many aspects of the transmission of heredity remain obscure, but a momentary culmination of these various researches was attained with the identification of the two types of nucleic acid DNA and RNA with, respectively, the cell carrier of genetic information and the agent responsible for translation of this information into the regulation of cell metabolism.

The occurrence of these high molecular weight polymers in all living tissues including the viruses is now well established as is their close association with protein, and a vest body of evidence indicates that their precise structure is of signal importance in the expression of the biological function of the nucleic acids.

1. 1. Chemical and physical structure of the nucleic acids

(e) RNA

RMA consists of the nucleoside 5'-monophosphetes of edenine, cytosine guanine and uracil joined together in long chains through the 5'-phosphete group on one nucleotide to the 3'-hydroxyl group on the neighbouring nucleotide. In addition, certain ribonucleic acids such as 5-RM contain a relatively high proportion of a derivative of uracil, 5-ribofuranosyluracil (pseudouridine) (Gohn, 1960). Various ribonucleosides derived from methylated purines and pyrimidines, including thymine, 5-methyloytosine, 1-methyledenine, 2-methyledenine, 6-methyleminopurine, 6-dimethyleminopurine 1-methylguanine, 6-hydroxy-8-methyleminopurine and 6-hydroxy-8-dimethyleminopurine, occur in small emounts (Adler, Weissman and Gutman, 1956;

- 2 -

Dunn, Smith and Spehr, 1960; Dunn, 1959, 1961/. Recently, the occurrence of inosine and 1-methylinosine has been reported in yeast S-RNA (Hall, 1963). The type of linkage and the identity of the nucleotides were determined by studies on the products of elkeline and enzymic hydrolysis of ribonucleic acids extracted from various becterial and mammalian sources (John, 1950, 1951; John and Volkin, 1953; Whitfield, Heppel and Markham, 1955). Early studies on RNA had suggested that it was a tetranucleotide since the methods then in use gave rise to fragments of this order of complexity. This hypothesis was soon abandoned when, with the edvent of more precise analytical techniques, such as carometography. the base composition of RNAs from different sources was observed to vary very widely. Furthermore, with the introduction of milder methods of extraction it became possible to isolate RMAs containing up to several hundred nucleotide units. Recent studies have also shown that the cell content of RNA is heterogeneous and RNAs with molecular weights ranging from 10,000 to 20,000 for 3-RNA (Lemecnik, Stephenson and Hecht, 1958) to complex polymers of 10⁶ or more have been described.

The secondary structure of RMA remains somewhat obscure and is probably related to the specific function of the different types of RMA. According to Doty (Doty et el., 1959; Doty, 1961) it appears that the larger molecules behave as irregularly coiled, relatively compact, single polymeric chains involving up to one-half of the molecule.

(b) DNA

DNA has recently been identified as the genetic substance (Avery,

- 3 -

MacLeod and McGarty , 1944; Hershey, 1953; Meedle, 1957; and the expression of its functions has received an admirable definition by Kornberg (1962) in the following words: "In its role DNA must have two functions; it must contain information, in chemical code, to direct the development of the cell according to its inheritance, and it must be reproducible in exact replice for the transmission of this inheritance to future generations".

In its chemical composition DNA differs from RNA in that it contains thymine in place of uracil and 2-decxy-D-ribose in place of D-ribose. Moreover, in addition to the four major bases, certain others occur in small amounts. The most widely distributed of the unusual bases is 5-methylcytosine which is particularly abundant in wheat germ DMA (Wyett, 1950, 1951 and also is present in DNAs from celf thymus, fish and insect sources. T-even bacteriophages have been found to contain 5-hydroxymethyl cytosine where it replaces cytosine (Wyett and John, 1952; Kornberg, 4immerman, Kornberg and Josse, 1959); 6-methylaminopurine has been discovered as a constituent of DNA extracted from certain strains of <u>Bacherichis coli</u> (Dunn and Smith, 1958).

The DNA molecule is built up from deoxyribonucleotide units linked together in a manner similar to that existing in RNA. In contrast to RNA, analysis of samples of DNA from various sources have revealed that certain regularities in its base composition generally apply. It was found that the purine content always equalled the pyrimidine content, and while smong the purines the adenine content may differ considerably from the guanine, and among the pyrimidines, the thymine from the cytosine, there is always

. . .

equivalence among the bases with an amino group in the 6-position in the ring to the bases with a keto group in the 6-position (Ghargaff, 1950, 1951, 1955).

That DNA is a very large molecule became evident after the introduction of mild methods of extraction, and with the aid of recently perfected physical techniques, such as light sometering and ultracentrifugation, it has become possible to obtain a measure of its molecular weight. The values so obtained have varied between $2 \ge 10^6$ and $16 \ge 10^6$ for DNAs from different sources and it is possible that even molecules of this size are degraded forms of still more complex molecules (Hershey and Bungi, 1960; Levinthal and Davison, 1961). For the T2 becteriophage DNA, molecular weights of about 100 x 10^6 have been quoted and it was suggested that a molecule of this order of complexity represented the total quote of DNA in each phage particle (Rubenstein, Thomas and Hershey, 1961).

Although the DNA has been regarded as the least variable of all cell components and the total DNA content has been utilized to calculate the total number of cells in a given portion of tissue (Davidson and Leslie, 1950), a considerable body of evidence is available to indicate that within a single cell nucleus there is physical heterogeneity of DNA molecules. Chromatographic fractionations of DNA on SCTEOLA cellulose (Bendich <u>et al.</u>, 1958), have indicated that the various fractions so obtained exhibit base ratios that differ widely from those predicted by the Chargeff rules; similar results were obtained in studies using the ultracentrifuge (Butler, Laurence, Robins and Shooter, 1959).

- 5 -

On the basis of storeochemical models, X-ray crystallogrephic mensurements acoumulated by Wilkins and collaborators (Wilkins, 1957) and chemical data, Wat on and Grick (1953a.L. proposed a structure for DNA in which two polynucleotide chains are wound about a common axis in a heliasl menner. The physical measurements had indicated that DN/ exists in the form of long stiff rods, 10,000 or more nucleotide units in length, and Watson and Drick further proposed that the decaypentose rings linked by 3', 5'-phosphate residues formed the backbone of the chain with the purine and pyrimidine rings emerging at right angles from the main axis of the chain on the inside. To explain the exceptional stability of the DRA molecule it was suggested that the purine and pyrimidine bases on one strand are bonded to the pyrimidine and purine bases on the complementary strand by hydrogen bonds. X-ray measurements have indicated that the space between opposing strands in the model agrees with the calculated value for a purine hydrogen bonded to a pyrimidine, thus accounting for the equivalence of the purines to the pyrimidines and the hydrogen bonding of edenine to thymine and cytosine to guanine. Fig. 1 shows how the nucleotide units are linked together in DNA and also indicates that the two complementary strands are of opposite polarity; Fig. 2 represents a spacefilling molecular model of DNA.

Gollapse of the rigid helical structure to a random coil form may be brought about when DNA is in aqueous solution under the influence of such factors as temperature, pH, ionic strength and dielectric constant of the medium or by agents such as urea that are known to favour dissociation of

- 6 -

Fig. 1



Fig. 1. Schematic representation of a section of the DNA molecule.

(Adapted from Kornberg, A. (1960) Science, 131, 1503)



Fig. 2. Three-dimensional representation of a portion of the DMA molecule.

(after Foughelman, M., Langridge, R., Soeds, W. E., Stokes, A. H Wilson, H. R., Hooper, C. W., Wilkins, M. H. T., Bercley, R. K and Hamilton, L. D. (1955) Nature, Lond., <u>175</u>, 834) hydrogen bonds. This phenomenon is referred to as denaturation. The predice structural changes induced in the DNA molecule by denaturation are not well understood and conflicting views have been expressed as to the molecular weight of denatured DNA. It was suggested (Doty, Marmur, Bigner and Schildkraut, 1960) that DNA is converted to single stranded material upon denaturation since the data from sedimentation and diffusion measurements had indicated that the molecular weight had been halved. On the other hand, Rice and Doty (1957) and Gavalieri, Deutsch and Rosenberg (1961) failed to detect differences in the molecular weight of DNA before and after denaturation when the progress of denaturation was followed by lightscattering techniques. The available evidence seems to suggest that there is a degree of strand separation associated with the denaturation process while some hydrogen bonds remain unbroken.

The association of DNA with the chromosomes when cells are undergoing mitosis is supported by an impressive body of evidence, but the secondary structure of DNA or the precise arrangement of the polynucleotide double helix in morphologically recognizable structures remains an unsolved problem (Taylor, 1963; Thomas, 1963). However, Cairns (1963) provides evidence which suggests that the circular "chromosome" of <u>E. coli</u> (Kellenberger, 1960) consists of a single DNA unit with a molecular weight of $4 \ge 10^9$.

1. 2. Biosynthesis of the Nucleic Acids

Self-replication is a unique property of the living cell and since its hereditary message has been found to reside in its polynucleotide

- 7 -

complement, the cell must possess efficient mechanisms for the precise duplication of this polynucleotide material. While the individual stages of this process may not be easily discernible in the intect cell, studies on broken cells and cell-free systems have shown that the general characteristics of the biogenesis of the polynucleotides makes it convenient to consider the process as proceeding through the following four main phases: - (1) An initial phase involving the assembly of the purine and pyrimidine ring systems, the biosynthesis of the carbohydrate moieties. and the convergence of the synthetic pathways of the purime and pyrimidine beses on the one hand and the pentose moistles on the other in the formation of nucleoside monophosphetes; (2) the conversion of ribonucleotides to deoxyribonucleotides; (3) the phosphorylation of ribo- and deoxyribonucleos: monophosphates to the corresponding di- and triphosphates; (4) polymerisatic of ribonucleotides and decxyribonucleotides to polynucleotide material in presence of an appropriate polynucleotide primer.

Although the pheses outlined above appear to apply universally irrespective of the origin of the cells, the precise sequence of anabolic events is a property of the individual cell types and, in addition to <u>de</u> <u>novo</u> biogenesis, some cells can utilise preformed (see Section 2. 1. (c)) purine and pyrimidine bases and can thus dispense with one or more of the oblightory stages in the formation of the polynucleotides.

(a) Biosynthesis of purine ribonucleoside monophosphates

Although it has been known for a long time that many organisms are capable of producing purines de novo from small molecules, the

- 8 -

establishment of the complete sequence of events leading to the assembly of the purine ring has come about primarily as a result of the afforts of Buchenan and Greenberg and their colleagues over the last 15 years. The initial attack on the problem of purine biosynthesis was carried out by administering small molecules labelled (13C) to live pigeons and examining the distribution of this isotope in the excreted unic said. This was achieved by selective degradation of the recovered uric soid in such a way that each individual carbon atom could be assayed for its isotope content (Buchenan, Sonne and Delluva, 1948). On the basis of these preliminary studies with labelled precursors of unic soid, it was shown that carbons 2 and 8 were derived from formate, carbon 6 came from GOg and that the ultimate precursor of carbon 4 was 0, of glycine (Sonne, Buchanan and Delluve, 1948; Elwyn and Sprinson, 1950). Simultaneous studies on the incorporation of mitrogenous precursors into uric acid had revealed that nitrogen 7 was derived from the anino group of glycine (Shemin and Rittenberg, 1947). In further investigations it was found that mitrogens 3 and 9 arose from the smide group of glutamine and that either glutamic acid or aspertic acid donated the nitrogen in position 1 of the purine ring structure (Sonne, Lin and Buchanan, 1953, 1956). Buchanan et al. (1957) recently demonstrated that aspartic acid was the specific donor of nitrogen atom 1.

At a very much earlier date it had been shown that lack of the enzyme xanthine oxidase prevented the formation of unic acid in pigeon liver slices and that the product which accumulated was instead hypoxanthine (Edson,

- 9 -

Krebs and Model, 1936), and this observation was used by Greenberg (1948, 1950) to demonstrate that the same small molecules which provided the carbon and nitrogen atoms in uric acid were also precursors for hypoxanthine Analogous studies with memmels and migroorganisms had indicated that an identical assortment of small molecules was the precursor pool for the purine rings of the guanine and adenine of the nucleic acids (Heinrich and Wilson, 1950), and the conclusion was, therefore, inescepable that the verious purines had a common precursor in their biosynthesis from small Greenberg (1951a) studied the incorporation of 14G-formate and fragmonts. 14000 into hypoxenthine with the use of pigeon liver homogenetes and made the important discovery that hypoxenthine was not a primary product but was derived from the ribonucleotide of hypoganthine - inosinic acid. This compound was well established as a demmination product of edenylic acid and although there was little evidence to indicate that it was a precursor of AMP, IMP attracted much attention and has since remained a foorl point in purine metebolism.

In the search for possible intermediates in purine biosynthesis, an observation made by Statton and Fox (1945) proved partiment. They found that when <u>Recharichia coli</u> was grown in presence of sulphonamides, a diszotisable amine could be isolated from the culture medium and this amine was later identified by Shive <u>et al.(1947</u>) as 5-amino-4-imidazole carboramide The probable relationship of this compound to purine synthesis was immediated recognized since it is equivalent to hypoxenthine with carbon 2 missing. Although this material was shown to be incorporated into the nucleic acids

- 10 -

(Miller, Gurin and Wilson, 1950), experiments conducted with pigeon liver extracts (Greenberg, 1951b; Shulmann and Buchaman, 1952) indicated that the true intermediate in hypoxanthine synthesis was not 4-emino-D-imidezole carboxamide itself but rather its ribonucleotide derivative AIGAR. Further in vitro studies using the pigeon liver system revealed that AIGAR was converted to 5-formamino-4-imidezole carboxamide ribonucleotide (PAIGAR) (Greenberg, 1954) by the transfer of a formyl group from H¹⁰-formyltetrahydrofolic moid (Hartman and Buchaman, 1959). This reaction is followed by Fing closure to yield incoine 5'-monophosphate (IMP) (Warren, Flake and Buchaman, 1957). There now remained the elucidation of the pathway leading to the production of AIGAR and again the pigeon liver extract proved an invaluable tool in that it represented a source from which all the enzymes of purine blosynthesis could be obtained in a soluble form and in high activity.

The discovery that the phosphoribosyl derivatives were involved in purine biosynthesis (Goldthweit, Peebody and Greenberg, 1954) had led to a series of investigations into the intermediates involved in the synthesis of these compounds. From such studies an enzymic mechanism was discovered which involved ribose-5-phosphate and ATP to give 5-phosphoribosyl-1-pyrophosphate (PRPP) in a reaction first described by Kornberg, Lieberman and Simma (1954, 1955a, b). In what is now considered the first step proper in purine synthesis <u>de novo</u>, PRPP was shown to be utilised by a purified pigeon liver system with glutamine in the formation of 5-phosphoribosylamine (PRA) (Goldthweit, 1956), which in turn condensed with glycine to give

- 11 -

glycinamide ribonucleotide (GAR). The next step was shown to be the formylation of GAR to yield formylglycinamide ribonucleotide (FGAR) (Goldthwait, Peabody and Greenberg, 1956; Warren and Flaks, 1956) by the intervention of N⁵, M¹⁰ -anhydroformyltetrahydrofolic soid as donor of the formyl group (Hartman and Buchanan, 1959). The stage which follows involves the reaction of FGAR with glutamine and ATP to form irreversibly formylglycinsmidine ribonucleotide (Levenberg and Buchanan, 1957b). However, with less purified enzyme preparations, the appearance of this compound is transient and the product which then accumulates was identified as 5-eminoimidazole ribonucleotide (AIR) (Levenberg and Buchanan, 1957a). Lukens and Buchanan (1957) discovered that the next step in purine biosynthesis was the reaction of AIR with GOg (or HGOg) to yield 5-emino-4imidazolecarboxylic acid ribonuclectide followed by reaction with aspertic acid and ATP to give 5-amino-4-imidazole-[N-succiny1,-carboxamide ribonucleotide, which in turn is cleaved to fumarate and AICAR (Miller, Lukens and Buchanan, 1957). Thus a complete reaction sequence in the enzymic formation of IMP from small precursors has been established using the system derived from pigeon liver, and although information from other sources is not impressive, all the available evidence points to a similar sequence of events in all cells which possess the ability to synthesise purines de novo. The probable reaction mechanism is illustrated in Fig. 3.

Experiments on bone merrow extracts demonstrated that IMP was a precursor of both GMP and AMP (Abrams and Bently, 1955e) of the RNA in that tissue, and further in vitro studies with bone marrow (Abrams and Bently, 195



Fig. 3. The enzymic synthesis of inosinic acid de novo.

(from Davidson, J. N. (1960) "The Biochemistry of the Nucleic Acids", 4th Ed., London: Methuen and Go. Ltd.; and with pigeon liver (Jarter and Cohen, 1956; Lieberman, 1956; have established that the initial remotion in the conversion of I P to AMP involves the condensation of IMP with L-aspertic sold in presence of GTP as co-factor. The ademylosuccinic sold thus formed is then ensymically eleaved to give, reversibly, AMP and fumarate.

The first step in the enzymic conversion of IMP to GMP involves the oxidation of IMP by an MAD-dependent enzyme to give XMP. This compound is subsequently aminated by the amide nitrogen of glutamine in the formation of GMP; an ATP requirement was demonstrated for systems from bone marrow (Bently and Abrams, 1956) and from pigeon liver (Lagerkvist, 1955). The relevant conversions are shown in Fig. 4.

(b) Biosynthesis of pyrimidine ribonucleoside monophosphetes

On eccount of the clear structurel similarity between the purine and pyrimidine ring systems it was thought for a long time that there was an intimate blochemical relationship between their respective blosynthetic pathways and that the purines were predursors of the pyrimidines or <u>vide</u> <u>versa</u> since both types of compound could be built up from similar types of small molecules. Thus it was shown that ammonis, 30_2 and aspertic acid were incorporated into the pyrimidines of avian liver polynucleotides. However, when methods became available for specific degredation of biosynthetically labelled pyrimidines, it appeared that nitrogen atom 1 was derived from ammonis (legertwist, 1953), derbon 2 from 30_2 (Heinrich and Wilson, 1950) and the remainder of the pyrimidine ring was derived from aspertic acid, <u>i.e.</u> N-3, 0-4, 3-5 and 3-6 (Reichard and Legertwist, 1953).



Fig. 4. Ensymic synthesis of adenylic and guanylic acids from incainic acid.

(from Davidson, J. N. (1960) "The Biochemistry of the Nucleic Acids", 4th Ed., London: Methuem and Go. Ltd.)
More striking evidence for the existence of a unique pathway for pyrimidine synthesis de novo was furnished by studies on mutant strains of Neurospora which require, for growth, the nucleoside uridine; some of these mutants were found to utilize the pyrimidine orotic acid in place of uridine (Mitchel Houlahan and Nyc, 1948). Similarly, orotic acid is a growth factor for Lactobacillus bulgaris 09 and when 143-labelled orotic acid was provided in the medium, radioectivity could be recovered in the UMP and GMP of the becterial nucleic acids while the AMP and GMP remained unlabelled (Wright et al., 1951). In addition to protic acid, 5,6-dihydroorotic acid was shown to support growth in this organism and labelled ureidosuccinic acid was utilised for pyrimidine synthesis. Experiments on extracts from yeast (Edmonds, Delluve and Wilson, 1952), ret liver (Hurlbert and Potter, 1952) and various mammalian tissues (Weed, 1951) similarly appeared to implicate ureidosuccinic acid and orotic acid in a biosynthetic pathway leading to uridine 5'-monophosphete (UMP).

In bacterial proparations the preliminary step in this sequence was shown to be the interaction of MM_3 and CO_g in presence of ATP to form carbanyl phosphete while in rat liver systems this compound was apparently produced vie an alternative route requiring the presence of an acyl-L-glutemi acid (Jones, Spector and Lipmann, 1955; Reichard and Hanshoff, 1956). In what is now regarded as the initial step in pyrimidine biosynthesis <u>de</u> <u>novo</u> carbanyl phosphete reacts with aspartic acid to yield ureidosuccinic acid (or carbanyl-L-aspertic moid) in a reaction which is essentially irreversible (Reichard, 1954, 1957). Lieborman and Kornberg (1954)

- 14 -

demonstrated that cell-free extracts of <u>Aymobacterium croticum</u> were capable of catalysing the interconversion of ureidosuccinic acid and crotic acid through the intermediate 5,6-dihydrocrotic acid; ring closure was effected by the ensyme dihydrocrotage followed by exidation of the ensuing dihydroorotic acid under the influence of a flevin nucleotide linked dehydrogenase (Friedman and Vennesland, 1958). The two ensymes also appeared to be present in rat liver (Wu and Wilson, 1956) and in extracts of <u>E.coli</u> (Yates and Pardee, 1956a) where both reactions were shown to be freely reversible.

The formation of the first nucleotide in the pyrimidine sequence, orotidine 5'-phosphete was accomplished by the reaction of orotic acid with PMPP in presence of the highly specific orotidylic acid pyrophosphorylase, thus the enzyme purified from yeast was without action on other pyrimidines (Lieberman, Kornberg and Simms, 1955). Orotidine 5'-phosphate was in turn converted to uridine 5'-monophosphate (DMP), irreversibly, by orotidine 5'-phosphate decarboxylase (Hurlbert and Reichard, 1955). The probable complete pathway is shown in Fig. 5.

Up to the present no pethway of <u>de novo</u> pyrimidine synthesis has emerged other than that leading to the production of UMP; all the other pyrimidines being formed either directly or indirectly from UMP and, as far as is known, UMP may be said to perform a role analogous to that played by IMP in purine mucleotide metabolism. Thus UMP is a precursor of the UMP and, ultimately, of the TMP of the nucleic soids.

Detailed pathways leading from UMP to the other pyrimidine nucleoside monophosphates have been elucidated in recent years and several detailed

- 15 -



Fig. 5. The ensymic synthesis of unidylic acid de novo.

(After Birnie, G. D. (1959) Ph.D. Theats,

The University, Glasgow)

studies have emphasized that a number of alternative routes may operate to effect these nucleotide interconversions (see Section 2. 1. (b)). The precise conditions under which cells can utilize one route in preference to others remain uncertain, neither has the question of possible species preference in the selection of such pathways been satisfactorily answered (see reviews by Grosbic, 1960 and Bessman, 1963).

That a pathway existed to convert UMP to GMP was indicated by various metabolic studies in which the extent of labelling of the nucleic acid pyrimidines was followed as a function of time after the injection of variou pyrimidine precursors. Incorporation of 140-labelled orotic acid into the nuclear RNA of rat liver and rat liver slices (Hecht and Potter, 1956, 1958) showed that the ratio of the specific activity of GMP to the specific activi of UNP from the same samples eventually approached unity. Thus it would appear that part of the UMP formed by the de novo pathway is converted to SMP before being incorporated into the RNA. The exact pethway for the conversion of UMP to GMP was investigated by Lieberman (1956) with extracts of R. coli and by Kammen and Hurlbert (1959) using a call-free supernatant fraction from Novikoff hepatoma. Both systems appeared to catalyse a direct amination at the level of UDP or UTP in the presence of ATP, amination at the triphosphate level being favoured by both sets of data, but the systems differed in that the E.coli extract required ammonia for the amination while the mammalian system utilised glutamine as donor of the amino group.

The biosynthesis of the thymidine derivatives has attracted much attention in recent years. The presence of a methyl group in the 5-position

- 16 -

of the thymine ring creeted special problems of biosynthesis, and further interest was stimulated by the observation (Reichard and Estborn, 1951) that the thymidine nucleotides appeared to function almost exclusively as precursors of DNA.

Early studies on whole cell systems had pointed to the role of uracil and cytosine derivatives as precursors of DNA thymine; thus E.coli cells were found capable of incorporating orotic acid into DNA thymine residues (Reichard, 1949) and ure oil was utilised in the synthesis of DNA thymine by thymine-less mutents of E. coli (Green and Johen, 1957). Uridine (Ames and Megasanik, 1957; and cytidine (Rose and Schweigert, 1953) have likewise been shown to be incorporated intact into DNA thysine residues. From these and similar studies it was suggested that the de novo conversion of urecil derivatives to thymine derivatives was conducted entirely on the nucleotide level without at any stage implicating free pyrimidine bases. This conclusion was confirmed by the demonstration that 143-labelled decryoytidine (CdR) appeared in both cytosine and thymine residues of DNA when injected into rate (Reichard and Estborn, 1951). In experiments designed to illuminat the nature and origin of the one-cerbon unit which constitutes the difference between the urecil and thymine ring systems, the B-carbon of serine as well as the a-carbon of glycine were found to be incorporated extensively into the 5-methyl groups of DNA thymine in the adult ret (Elwyn and Sprinson, 1954 In addition, formate and formaldehyde could also serve as precursors in rat tisaues (Totter, 1954). The evidence from such incorporation experiments pointed to a folic acid derivative as the donor of the one-carbon unit in the

- 17 -

formation of thymidylic soid; more specifically, there were reasons for believing N^D, N¹⁰-methylenetetrahydrofolic acid to be the active intermediate in this reaction. This derivative was reported to be the product of non-enzymic interaction of formaldehyde and tetrahydrofolic acid and of serine hydroxymethylese action on werine (Jaenicke, 1956). Independent evidence to this effect was derived from the fact that low concentrations of the folig acid antimotabolite aminopterin blocked the utilisation of deoryuridine for thymine synthesis (Friedkin and Roberts, 1956). Evidence pertaining to the mature of the one-carbon acceptor in thymidylate synthesis have come from comperisons of incorporation rates of verious possible preguraors into polynucleotides. Such experiments have revealed that deoxyuridine like thymidine was utilised elaost exclusively for UNA synthesis while decryuridine itself was not incorporated into the DNA of regenerating ret liver (Reichard, 1955). Friedkin and Hornberg (1957), using cell-free extracts of E.goli, described the formation of thymidine 5' -triphosphate (TTP) from deoxyuridine 5'-phosphate (dUMP) in presence of MD, M10-methylenotetrahydrofolic soid, ATP, Mg and thymidylete kinese, thus confirming the role of decryuridine derivatives as the immediate precursors of TMP. The missing links in one possible route leading from UMP to TMP were provided by the discoveries of enzyme systems which catalyse the reduction of CDP to dODP (Reicherd and Rutberg, 1960) and of the specific desminases responsible for the conversion of dGMP to dUMP (Scareno, 1958; Maley and Maley, 1959) (see Section 2. 1. (b)).

- 18 -

(c) Blosynthesis of the pentose moleties of the nucleic acids

D-ribose (Levens and Jacobs, 1909) and 2-decxy-D-ribose (Levens, Mikeska and Mori, 1930) were early identified as the cerbohydrate components of RNA and DNA, respectively. However, the intracellular formation of these compounds remained obscure until after the Embden-Neyerhof scheme of anaerobic glycolysis had received general recognition. At that time, an alternative pathway of glucose exidation (Warburg, Christian and Griese, 1935; Dickens, 1938) which led to the formation of a pentose phosphate with 6-phosphogluconic acid as one of the intermediates had been discovered.

Actual investigation of the aerobic biosynthesis of pentoses was initiated about 1950 when Gohen and Horecker undertook intensive studies of the "hexose-monophosphate shunt" and identified the pentose intermediates. Preliminery studies led to the conclusion that D-ribulose-D-phosphate occupie a key position in the formation of D-ribose-5-phosphate from glucose (Scott and Sohen, 1951; Horsoker, 1951). The first reaction involved was shown to be the oxidation of D-glucose-6-phosphate by NADP in presence of the widely distributed glucose-6-phosphate dehydrogenase to fora C-phospho-D-gluconie acid (Glock and Malean, 1951; Horacker and Smyrniotia, 1953). Further studies indicated that yeast, bectaria (Scott and Schen, 1953) and mammalian tissues (Seegmiller and Horacker, 1952) contain the enzyme 6-phosphogluconic acid dehydrogeness which in presence of NADP detalyses the oxidative deerrboxylation of 6-phosphogluconic acid and leads to the formation of D-ribulose-5-phosphete: the precise mechanism of this reaction is still obscure but it seems clear that a decarboxylation occurs at carbon atom 1

- 19 -

and a dehydrogenation at carbon atom 3 (Glock, 1955). The conversion of ribulose-5-phosphate to D-ribose-5-phosphate is reversibly catalysed by the widely distributed enzyme ribose-5-phosphate isomerase (Horecker and Smyrniotis, 1951; Horecker, Smyrniotis and Seegmiller, 1951).

While these studies were in progress, a number of investigations pointed to a series of anacrobic reactions in which pentose phosphates were reconverted to herose phosphates and triose phosphates. These reactions were shown to be reversible and to involve the intermediate formation of both tetrose and heptulose intermediates. Horocker et al. (1953) implicated the thismine pyrophosphete dependent enzyme transketoless in these reactions and showed that it catalysed the cleavage of xylulose-5-phosphete by the transfer of a ketol group to a suitable acceptor aldehyde; moreover, it was shown that ribose-5-phosphate qualified as a reaction partner giving glyceraldehyde-3-phosphate and sedoheptulose-7-phosphate (Horacker and Smyrniotis, 1952). Confirmation of the connection between the merobic and anaerobic pathways of pentose phosphate biosynthesis was provided by the demonstration that animal tissues and micro-organisms contain an enzyme which cetalyses the isomerisation of ribulose-5-phosphete and xylulose-5-phosphete (Hurwitz and Horecker, 1956; Dickons and Williamson, 1956). In the presence of glyceraldehyde-3-phosphate, the heptulose phosphate was shown to be cleave by the enzyme transaldolese in such a manner that its dihydroxyscetone moiety was transferred to the triose phosphate with the formation of frustose-6phosphate and erythrose-4-phosphate (Horecker and Smyrniotis, 1955). Fructose-6-phosphate may finally be converted to glucose-6-phosphate under

- 20 -

the influence of phosphohexcisomersse, thus completing the pentose phosphate pathway.

The precise interactions between the transaldolase and transketolase systems and the stoichiometry of the various reactions involved have been further elucidated by Horecker and his collaborators (Horecker <u>et al.</u>, 1956, but knowledge of the relative importance of the various elternative pathways in the production and reutilisation of ribose-5-phosphate is still rudiments and current investigations (Ghosh and Barnstein, 1963) indicate that in almost no instance is one pathway utilised to the total exclusion of others.

Although ribose-5-phosphate may be considered as the ultimate product in the pentose phosphate pathway, this sugar requires further modification to complete the connection between <u>de novo</u> synthesis and the ribose of the nucleotides and the nucleic acids. Formation of ribose-1-phosphate has been shown to occur in two ways. It can be formed reversibly from ribose-5-phosphate by the action of phosphoribomutase, an enzyme which has been demonstrated in muscle (Guerino and Sable, 1955), or it can arise by the action of the enzyme nucleoside phosphorylace (Kaleker, 1947) which catalys a reaction of the type:

Base-ribonucleoside + inorganic phosphate Base + ribose-lphosphate However, as pointed out by Kalakar (1947), there are reasons for believing that the reaction favours utilisation rather than formation of ribose-l-phosphate (see Section 2. 1. (c)). In <u>E.coli</u> (Long, 1955) and in rat liver (Agranoff and Brady, 1956) ribose-5-phosphate can also be formed from D-ribose by the enzyme ribokinase in presence of Mg⁺⁺ and ATP.

- 21 -

The discovery by Kornberg, Lieberman and Simms (1954, 1955 a, b) of the reversible reaction leading to the formation of 3-phosphoribosyl-1-pyrophosphate (PRPP) from ribose-5-phosphets has proved to be of widespread significance in the biosynthesis <u>de novo</u> of purine (Section 1. 2. (a)) and pyrimidine (Section 1. 8. (b)) nucleotides and also at the level of the utilization of the preformed bases (Section 2. 1. (a)). The reaction involves transfer of the terminel pyrophosphete group of ATP to the carbon 1 of the ribose-5-phosphate, in shigh position the pyrophosphete unit apparently essumes the a-configuration (Romy, Romy and Buchaman, 1955). Some of the relevant reactions in the pertonse phosphete pathway are illustrated in Fig. 6.

A pethway for the formation of 2-decayribose, not involving the intermediate formation of ribose or nucleoside derivatives, was described by Racker (1951, 1952), who demonstrated a condensation of glyceraldehyde-3phosphate and acetaldehyde to produce 2-decayribose-5-phosphate. The enzyme ontalysing this reaction was termed decayriboseldeless and was reported to occur in <u>F. coli</u> (Racker, 1951) and in ret liver (Boxer and Shohk, 1953). Further utilisation of the decayribose-5-phosphate so formed could be viewalised through the formation of decayribose-1-phosphate by a phosphodecayribonutese described by Manson and Lempen (1951) and the coupling of this remation with the nucleoside phosphorylese of Friedkin and Kalcker (1950 to form decayribonucleosides in the following manner:

Deoxyribese-l-phosphate + purine ____ purine deoxyribeside + inorganic phosphate

Similar phosphoryleses, present in ret liver, react with the pyrimidine thymine and uracil, though apparently not with cytosine (De Verdier and

- 22 -



Fig. 6. Some remotions in the formation de novo and the metabolism of ribose phosphates.

The symbol P represents ester linked orthophosphate.

(adepted from Fruton, J. S. and Simmonds, S. (1956) "General Biochemistry", 2nd Bd., Gh. 21, New York: John Wiley and Some Ltd.) Potter, 1960). Finally, in vivo studies on some growing bacterial (Lenning and Sohen, 1955; David and Jaymond, 1958) and mammalian (Ghosh and Bernstein, 1963) cell populations have been interpreted as indicating a direct reduction of ribose to decayribose.

The significence of the Recker pathway for the formation of the decryribosyl molety of the nucleic acids remains doubtful, particularly since the equilibrium in the transaldolese reaction is unfavourable towards synthesis (Allende and Racker, 1963) and there are strong reasons for believing the pathway involving decayribose phosphates to be exclusively concerned with the degradation of the pentose molecule. Moreover, Kornberg (1957) failed to find any trace of a decayriboayl derivative analogous to PRPP, the entire intermediate in ribonucleotide biosynthesis. The most powerful evidence against the participation of de novo formed decayribose in DNA synthesis has been the large number of reports that a direct reduction of ribose to decayribose on occur while in ribosidic linkage in mucleosides or nucleotides. This conclusion is supported by two main lines of evidence; the first of which concerns experiments performed with ribonucleosides or ribonucleotides labelled with 140 in both the nitrogenous base and in the pentose mojety. Experiments of this type have shown that in microorganisms and in mammalian tissues such precursors were incorporated into the DNA without apparent alteration in the relative specific activities of sucar and base (Rose and Schweigert, 1953; Roll, Weinfeld and Carroll, 1956). The second line of evidence concerns attempts by enzyme studies in isolated systems to determine the precise level at which such reductions occur.

- 83 -

Reichard (1958) presented evidence in favour of the reduction of both cytidine and unidine at the ribonucleotide level with cell-free extracts from chick embryos, and in subsequent studies with purified enzymes from <u>E.coli</u> and from chick embryos it was demonstrated that the reduction of cytidine and unidine phosphates occurs preferentially at the diphosphate level (Reichard, Beldesten and Rutberg, 1961; Bertani, Häggmark and Reichard, 1961 The actual mechanism of the reduction isstill a matter of some dispute, but the <u>E.coli</u> enzyme appears to require the presence of Mg^{**} and ATP with reduced lippic acid as cofactor (More and Reichard, 1963). The question whether the action of this enzyme is restricted to the reduction of pyrimidin ribonucleotides is as yet unresolved, but it appears likely that the purine decoxyribonucleotides may be produced via a similar direct reduction mechanism (MacNutt, 1958); thus Reichard (1960) has presented evidence for the preferred reduction of GDP to dGDP in chick embryo extracts.

(d) Production of nucleoside di- and triphosphates

The primary products of the <u>de novo</u> synthesis of purine and pyrimidine nucleotides are in each case the nucleoside momphosphates; however, all the known systems that produce polynucleotides utilize as ultimate precursors either the nucleoside diphosphates or the nucleoside triphosphates and it is therefore necessary to consider the reactions that convert the monophosphates to their enabolically more useful polyphosphate derivatives.

The presence in cells of the most important of the nucleoside polyphosphates - ATF - can largely be accounted for by such well-known sources as exidative phosphorylation and by transfer of phosphate from 1, 3-diphosphe

- 24 -

glycerste and phospho(enol)pyruvate to ADP. However, the free ribo- and decryribonucleoside polyphosphates of the purine and pyrimidime series have been discovered in tissues (see Section 2. 1. (a)) and the means by which nucleoside polyphosphates other than ATP secure their phosphate have attracted considerable attention in resent years.

The findings of such studies have pointed to a series of transphosphorylations which can account for the distribution of phosphate from ATP to all the nucleotides, and in general, two classes of enzymes, which have been termed nucleoside monophosphate kinases (I) and nucleoside diphosphate kinases (II), may be considered as responsible for the legitimate de novo formation of nucleoside di- and triphosphates. Such reactions involve the transfer of a terminal phosphate from ATP and a requirement for magnesium ions has been demonstrated.

(I) nucleoside monophosphete + ATP ____ nucleoside diphosphete + ADP
(II) nucleoside diphosphete + ATP ____ nucleoside triphosphete + ADP

Muscle edenylate kinase, or myokinase, specific for adenine nucleotides was the first nucleoside monophosphate kinase to be discovered (Solowick and Kalckar, 1943; Kalckar, 1943) and was shown to catalyse the reaction:-

AMP + ATP ____ ADP + ADP

In the years following this discovery a large number of nucleotide phosphokinases were described exhibiting various specificities toward nucleotide acceptors; thus an enzyme partially purified from yeast (Strominge Heppel and Maxwell, 1954; 1959) catalysed the phosphorylation of AMP, GMP and UMP to the corresponding diphosphates, and a nucleoside diphosphokinese has been investigated in yeast, muscle and intestinal muccas (Berg and Joklik, 1954) where it was shown to convert UDP and IDP to UTP and ITP, respectively, in presence of ATP. The incorporation of deckyribonucleoside monophosphates into DNA indicates that such tissues are capable of phosphorylating the monophosphates to the triphosphate level, presumably under the influence of deckyribonucleotide kinages. A number of these have been described.

At present there is considerable evidence that many tissues enclose separate kinase systems specific for each of the purine and pyrimidine nucleoside monophosphates although the ability of such enzymes to distinguish between ribosyl and deoxyribosyl derivatives of a particular base romains a matter of some dispute (Bessman, 1963). In either onse, the kinases that act on the thymidine nucleotides would appear to be placed in a special position in that the corresponding ribosyl derivatives have not been detected (For a more detailed discussion of the phosphokinases, see Section 3. 1.).

(.) Biosynthesis of ribonucleic acids

A number of systems have been discovered which catalyse the formation of polyribonucleotides from ribonucleotidesprecursors (Grunberg-Hansgo, 1962; Smellie, 1963). Of the enzyme systems which have so far been described, the great majority appears to be involved in three general types of remotion; the first forms polynucleotides from ribonucleoside 5'-diphosphates with the elimination of inorganic phosphate, the second type, utilizing ribonucleoside 5'-triphosphates, adds a limited number of nucleotide units to existing RNA chains, the third type which also uses the 5'-triphosphates results in

- 26 -

incorporation of nucleotide residues into internucleotide linkage in polyribonucleotides.

The enzyme which promotes the incorporation of ribonucleoside 5' -diphosphetes into polynucleotide material, polynucleotide phosphorylase, wes discovered in 1955 by Grunberg-Manago and Ochoa in extracts from Asotobester vinelandii and upon subsequent pertial purification it was shown to catalyse the formation of high molecular weight polynucleotides indistinguishable from naturally occurring RNA (Singer, Heppel, Ochoe and W11, 1959) when examined under the sorutiny of chemical and enzymic methods of analysis. The ensyme appears to be widely distributed in bacteria and in plants (Grunberg-Manago, Ortig and Ochos, 1956; Littauer and Mornberg, 1957; Brummond, Steehelin and Ochos, 1957), while the evidence for its occurrence in animal tissues is unimpressive. Extensively purified enzymes from A. vinelandii (Singer, Heppel and Hilmoe, 1960) and from E. coli (Litteuer and Kornberg, 1957, have been shown to be dependent for activity on magnesium ions. Such preparations are capable of catalysing the polymerisation of ribonucleoside diphosphates in the absence of RNA primer (Beers, 1957) and they have been employed extensively for the preparation of artificial polynucleotides (Ochos and Heppel, 1957). The function of the polynucleotide phosphorylase in bacterial cells has not yet been established.

Enzymes from a variety of sources have been shown to add nucleotide units on to the ends of existing polynucleotide chains. This reaction is now relatively well understood and appears to involve the addition of a

- 27 -

few ribonucleceide 5'-phosphoryl units derived from the corresponding triphosphates on to the 3'-hydroxyl ends of pre-existing ribonucleic acids of rather low molecular weight. The original observation was made by Heidelberger et al. (1956) who discovered that P-labelled AMP was incorporated intect into the rat liver cytoplasm, and in subsequent experiments it was shown that in RNA so labelled the terminal AMP residue wes preferentially linked through a cytidyl-3', 5'-cytidyl grouping to the main polynucleotide chein (Hecht, Zameonik, Stephenson and Scott, 1958; Harbers and Heidelberger, 1959). Similar systems have been shown to exist in several other biological sources (Pater on and LePage, 1957; Herbert, 1958; Hurwitz, Bresler and Kaye, 1959). Important contributions to the understanding of the cellular function of these systems have been made by Lameonik and his collaborators (Hosgland et el., 1960; Hecht et. al., 1959) who have shown that the end unit containing AMP provides a functional grouping in soluble RNA (s-RNA) necessary for its role as carrier of activated amino acida.

A contrasting type of reaction involves the incorporation of ribonucleoside triphosphetes into the interior of polynucleotide chains. Genellakis (1957a) using a ret liver system, observed that a UMP unit derived from UTP could be incorporated into a non-terminal position, thus providing indication of extensive polynucleotide synthesis. Similarly, a system from the soluble fraction of chick embryos which incorporates AMP residues from ATP into RNA has been described (Shung and Wahler, 1959). Subsequent experiments on this system (Shung, Mahler and Enrione, 1960)

- 28 -

have indicated that the incorporation of adenylate residues is stimulated by the addition of GMP, UMP and GMP. The presence of systems of this type has further been demonstrated in pigeon liver microsomes (Straus. and Goldwasser, 1961), ascites careinome (Burdon and Smellie, 1961) and in <u>Microsoccus lysodeikticus</u> (Nekamoto and Weiss, 1962).

While such systems show requirement for various types of RNA primer, other systems have been described dependent on added DNA and the presence of all four ribonucleoside 5'-triphosphates for optimal activity. Although such enzymes appear to be widely distributed among bacteria (Hurwits, Breeler and Diringer, 1960; Furth, Hurwits and Goldmann, 1961; Ochoa, Burma, Kröger and Weill, 1961; Stevens, 1961) reports from animal sources are relatively few and include only a particulate fraction from onlf thymus nuclei (Abrega, Edmonds and Biswas, 1962) and a system derived from ascites tumour nuclei (Burdon and Smellie, 1962).

In the besterial systems, at least, it appears that the added DNA plays an important part in the polymerisation, since the base ratios of the products show a close resemblance to those of the added DNA (Neiss and Nekamoto, 1961a) as do the respective hearest neighbour frequences (Neiss and Nekamoto, 1961b). The generally accepted interpretation of these results is that such systems provide a mechanism for the transfer of genetic information from DNA to a complementary strand of RNA.

(f) Biosynthesis of deoxyribonucleic acids

DNA synthesis appears to be a process almost exclusively associated with cell proliferation and independent of the rate of cellular metabolism.

- 29 -

Although the strict limits of definition would tend to confine such synthesis to the ultimate polymerisation step, it seems probable that at its inception the point of diversion from the nonspecific pool of nucleotide precursors is the reduction of ribosyl to decxyribosyl derivatives; thereafter the process proceeds as directed through the specific sequence of reactions that leads to the formation of the DNA molecule (Ganellakis, 1962; Bessman, 1963,

Enzyme systems which cetalyse the formation of polydeoxyribonucleotides appear to occur very widely and the existence of such enzymes in all cells capable of mitosis seems probable. Such enzymes have shown a strict requirement for all the main four deoxyribonucleoside triphosphates, Mg" and highly polymerised primer DNA. DNA nucleotidyltransferace (DNA polymerase) was first discovered in extracts of E. coli by Kornberg, Lehman and Simms in 1956 and, subsequently, enzyme systems performing a similar role have been reported in tissue extracts from a veriety of animal, plant and bacterial sources (Bollum and Potter, 1958; Harford and Kornberg, 1958; Smellie, Keir and Davidson, 1959; Manstevinos and Ganellakis, 1959; Smellie, et al., 1960). Of ell such systems, only the nuclear enzyme from oalf thymus (Krakow, Coutsogeorgopoulos and Ganellakis, 1961; Keir and Smith, 1963) which incorporates a few ribo- and deoxyribonucleotides on to the ends of existing DMA chains appears to be markedly different in properties from those ascribed to the E. coli enzyme by Kornberg and his associates. The unique properties of the latter enzyme were intensively investigated on preparations purified several thousand-fold over the original E. coli extract (Lohman, Bessman, Simms and Kornberg, 1958) and it was shown that the general

- 30 -

characteristics of the reaction and the requirement for net synthesis of DMA (Bessman, Lehman, Simms and Kornberg, 1958) could be expressed as follows:



The mechanism of the remation catalysed by the <u>E.coli</u> DNA polymerase is viewed by Kornberg (1960) as the formation of a 3', 5'-diester linkage between two nucleotide units by an enzyme catalysed nucleophilic attack on the nucleotidyl phosphorus atom of the deoxyribonucleoside 5'-triphosphate by the terminal 3'-hydroxyl at the growing and of the new polynucleotide chain with the elimination of inorganic pyrophosphate (Fig. 7).

However, it was seen that when one of the substrates was cmitted from the remation mixture a small but significant number of deoxyribonucleotides were added on to the nucleoside ends of some DNA chains (Adler <u>et al.</u>, 1958) whereas the presence of the full complement of deoxyribonucleoside triphosphates resulted in a ratio of product to primer of 20:1. The end product so formed proved to be indistinguishable from high molecular weight, double stranded DNA as isolated from natural sources (Schachman <u>et al.</u>, 1958) A somewhet similar situation appears to obtain when the <u>3.coli</u> enzyme is incubated with either TTP and dATP or with dCTP and dGTP in the absence of primer DNA resulting in the formation of high molecular weight d-AT or d-OG polymers (Schachman <u>et al.</u>, 1960). Subsequent product enalysis has shown that the d-AT polymer so formed is composed exclusively of dAMP and TMP



Fig. 7. Postulated mechanism for extending a DNA chain.

(adapted from Kornberg, A. (1960) Science, 131, 1503)

residues arranged in a strictly elternating sequence on both strands with adenine hydrogen-bonded to thymine as in the primed reactions. In the oase of the polymor composed of dGMP and dGMP residues, the reaction again proceeds exponentially in the absence of primer DNA but the product formed differs from the d-AT polymer in that it is made up of separate strands containing dGMP or dGMP (Josse, Meiser and Kornberg, 1961).

Subsequent investigations have focussed ettention on the role played by the DNA primer in these reactions, and the findings not only helped to elucidate the machanism of DNA replication in <u>E.coli</u> but they provided powerful independent support for the predictions following from the setson and Crick (1953b) concept of the structure of DNA. Thus it appeared that the enzyme DNA polymerase receives directions from a template in the form of the DNA primer in such a way that it matches the particular purime or pyrimidime substrate which will form a hydrogen-bonded pair with a base on the template, the directions being dictated by the relationship between adenine and thymine and between guanine and cytosine. The net result appears to be that DNA functions as a template directing the synthesis of exact replices of itself.

These conclusions were based on several lines of enzymic evidence (Kornberg, 1962); firstly, the double stranded nature of the product, secondly, it was seen that the incorporation of purine and pyrimidine analogues into DNA followed the pattern predicted by the hydrogen-bonding ompabilities of these unnatural bases, thirdly, the chemical base composition of the primer DNA and the enzymic product were identical, fourthly, a

- 32 -

procedure developed for the determination of the 16 possible merest neighbour base sequences showed that different DNAs direct the synthesis of unique and non-random patterns of such sequences and that the two complementary strands were of opposite polarity (Josse, Keiser and Kornberg, 1961), and finally, the requirement of TTP, dGTP, dGTP, dATP and DNA primer for DNA synthesis. The evidence is clearly consistent with the contention that the DNA polymerese provides an appropriate mechanism for the replication of DNA.

Much less information is evaluable on the replication of DNA in enimal systems, but a DNA polymerose purified from calf thymus (Bollum, 1959, 1963) has been shown to be similar to the <u>E.coli</u> enzyme in many important respects. Partial purification of a DNA polymerose from ascites tumour cells has also been reported (Keir, 1962).

As to the physical state of DNA for optimal priming capacity, it was observed that when DNA is bested to 100° for 10 minutes it becomes twice as effective as a primer for the <u>E.coli</u> enzyme (Lehman, 1959) and, even more strikingly, when <u>E.coli</u> is infected with the bacteriophage T2 the native, double stranded DHA is relatively inert as a primer unless it is modified (Aposhian and Kornberg, 1962). Similarly, the single stranded DNA of the bacteriophage $\phi X174$ has proved to be an excellent primer for the <u>E.coli</u> polymerase (Lehman, 1959). It seems probable that although complete strand separation is not required for priming activity, part of the chain, at least, must be single stranded and available for base pairing with incoming precursor molecules.

- 33 -

It seems probable also that the replication process, in bacteria at least, proceeds sequentially in a predstermined direction starting from one end of the DNA chain (Negata, 1963; Yoshikawa and Sucoka, 1963).

In addition to the in vitro studies described above, a number of investigations on whole cells have lent support to the Watson-Grick concept of replication. The experiment of Meselson and Stahl (1958) showed clearly that when K. coli divides, each daughter cell receives two DNA units, one old (or "conserved") unit and one newly synthesised unit. This was interpreted as indicating "semi-conservative" replication. However, in subsequen experiments with this organism Cavalieri and Rosenberg (1961, 1962) suggested that the conserved unit was not a single strand as could be expected from the in vitro studies, but a double helix: thus each doughter cell was presumed to receive one new and one conserved double helix joined together as a pair. It is somewhat difficult to reconcile this finding with the known behaviour of the K.coli polymerase unless some mechanism of strand unwinding, separation and recombination is visualised prior to aitosis. In contrast, examination of the pH melting behaviour of hybrid DNA where 5-bromouracil was employed to label one subunit has revealed that the bonds which hold the subunits together bye specific and consistent with the properties expected of the hydrogen bonds of two strends of a single double helix (Shooter and Beldwin, 1962; Beldwin and Shooter, 1963), thereby suggesting that the replicating subunits are single polynucleotide strands.

1. 3.. Ostabolies of the nucleotides and their substituents

In addition to the enzyme systems that cetalyse the formation of

- 34 -

polynucleotides there appears to exist, in constant competition with the anabolic forces, a large number of catabolic systems operating at all levels of the synthetic pathways. The equilibris between the opposed forces may be modified in response to the immediate requirement of the cell with the involvement of physical as well as biological changes in the environment. Separate sellular locations of the enzyme systems of anabolic and catabolic function can be visualized to account for specifically directed synthesis and degradation of the polynucleotides since the equilibris of both nucleotide kineses and polymerases favour synthesis.

Gurrently, there is reason to believe that the polynucleotides are initially degraded by ribonucleases and decxyribonucleases, of which there are many types, to nucleoside monophosphates with an unknown number of oligonucleotide intermediate products. Little is known of the precise siturtions in which these degradations manifest themselves and hence the function of the enzyme systems which ontelyse them, but some are undoubtedly involved in the breakdown of nucleic acid fragments so that the ensuing nucleotide units may be reutilised for polynucleotide synthesis. The ultimate products of nuclease action appear to be nucleoside monophosphates with the phosphate group either in the 5'- or the 3'-position thereby reemphasising the role played by the nucleoside monophosphates as pivotal points in nucleotide metabolism.

The dephosphorylation of nucleoside 5'- or 3'-monophosphetes and the nucleoside 5'-polyphosphetes is catalysed by a number of phospheteses of low specificity abundant in many tissues. Such enzymes as intestinal phosphetese

- 35 -

exhibit low specificity towards both the nucleoside part and the number of phosphate groups. In such cases, the hydrolysis of polyphosphates probably occurs in a stepwise fashion, $\underline{g} \cdot \underline{g} \cdot \underline{g} \cdot \underline{h}$ ADP \rightarrow AMP \rightarrow edenosine. In addition, several types of enzyme are known with relatively selective action; thus the mammalian 5'-nucleotideses when purified are capable of hydrolysing only the nucleoside 5'-monophosphates while no action can be detected on 3'-monophosphates or polyphosphates (Heppel and Hilmos, 1951). 3'-Nucleotideses have also been reported in extracts of certain leaves and plant seeds (Heppel and Hilmos, 1953). The nucleosides, ultimate products of phosphatese action on nucleotides, are further split to give the free base and pentose moisties, presumably by the reactions cetalysed by (a) nucleoside phosphorylese or (b) by the irreversible hydrolysis effected by nucleoside hydrolase.

- 36 -

(Schmidt, 1955). Further degredation does not appear to occur in man and other primetes and in birds, while in other mammals ellantoin is the chief expretion product. The conversion of uric acid to ellentoin has been studied in detail in rat liver (Ganellakis and Cohen, 1955; Ganellakis, Tuttle and Ochen, 1955). An appraisal of the purine nucleotide interconversions relevant to the degredation of purines is shown in Fig. 8.

The cetabolism of the pyrimidine bases appears to be channelled exclusively through uracil and thymine, cytosine not being metabolised vigorously in either direction, and the present evidence suggests that cytosine derivatives are catabolised via uracil derivatives in the way shown in Fig. 8 (Potter, 1960; Lichtenstein, Barnum and Cohen, 1960).

The degradation of uracil and thymine in meanslian systems (Fink, Sline, Henderson and Fink, 1956; Gamellakis, 1956, 1957b) has been shown to involve an initial reduction step with the formation of the 5; 6-dihydroderivatives followed by ring opening to give the appropriate uraido-acid. Cleavage of the latter compounds produced β -amino acids, OO_g and NH_3 . The possibility that the amino acids so formed may obtain entry to the altric acid cycle has been visualised by Kupiecki and Goon (1957). In contrast to the pattern obtaining in meanslian cells, the ontebolism of thymine and uracil in selected bacteris appears to involve an initial oxidation leading to the production of 5-methylbarbituric acid and barbituric acid, respectively (Heyeishi and Kornberg, 1958; Wang and Lampen, 1958). The pathway of further degradation in bacterial systems has remained obscure until recently (Bigge and Doumas, 1963) when it was shown that 5-methylbarbiturate was



Fig. 8. The major pathways in the catabolism of purine and pyrimidine nucleotides.

Thick arrows indicate exit points to anabolic and ontabolic pathways. Thick half-arrows indicate the bias of the system when notively ontabolising. Broken arrows represent postulated reactions or reactions of doubtful significance in higher organisms.

(edepted from Potter, V. R. (1960) "Nucleic Acid Outlines", Vol. I, Ch. VIII, Minnempolis: Burgess Publishing Company) hydrolytically cleaved to upon and methylmalonate thus affording a possible alternative entry to the citric acid cycle.

1. 4. The control of DNA synthesis

The unique molecular properties and nicely circumscribed functions of DNA as the cellular store of gonetic information has resulted in a number of important conclusions related to the nature of events in the life cycle of the cell. The property of exceptional metabolic stability, alone, predicts the existence of efficient mechanisms directed towards the protection and preservation of the cellular complements of DNA so as to ensure the viability of such progeny as receive its hereditary message. Moreover, it seems clear that the process of DHA replication, an essential prerequisite to this latter function, must be subject to an extremely rigorous set of cellular controls. Thus it is postulated that DHA synthesis must be so regulated that it provides exact replicas, identical with respect to composition, base sequence and amount to the original cellular quote of DNA. In addition to this quantitative type of control, several lines of evidence point to the existence of control mechanisms directly related to temporal events in the life cycle of the cell. This conclusion is partly based on the fact that demonstrable DNA synthesis seem to be confined to cell populations actively engaged in cell division while such synthesis appeared unrelated to the rate of general intracellular metabolism.

By the use of several recently developed techniques, such as autoradiography, quantitative microphotospectrometry and synchronous culture

techniques, it has become possible to determine to what extent intracellular DNA synthesis is subject to control; moreover, such studies have made it evident that in animal, plant and some basterial cells the process of DNA synthesis is restricted to a discrete period in their life cycle. The evidence from studies of this type clearly points to the existence of a DNA synthesis cycle associated with the cell division cycle and the evidence clearly admits the possibility that the DNA cycle onn be subdivided into certain defined periods (Howerd and Pelc, 1953). In cells of higher organisms, at least, there appears to exist a period following mitoeis during which no DNA synthesis takes place (G,), a subsequent period during which DNA is actively synthesised resulting in a doubling of the nuclear DNA content (S), a second period during which no DNA synthesis can be detected (G,) and, finally, the period of mitosis (M) (Lajthe, Oliver and Bilis, 1954; Prescott, 1960; Vendrely and Vendrely, 1956; Lark and Masipe, 1956; Hotte and Stern, 1961; Terasima and Tolmach, 1963,. In terms of the conventional concept of the cell division cycle, the 3 period coincides with the latter part of interphase (See Fig. 9). The duration of these periods very from system to system, but is often remarkably constant from cell to cell in the same population. It is also possible to obtain variation in the duration of periods of the DNA cycle by various chemical and physical treatments without disturbing the sequence of events (Lark, 1963). Thus it seems clear that a temporal control mechanism exists to allow DNA replication at a stege prior to the morphologically observed preparations for mitosis and since cellular DNA synthesis appears to be a discontinuous process, this control mechanism must be capable of deciding when to initiate

- 39 -



Fig. 9. A representation of the DNA synthetic cycle in relation to the cell division cycle.

> In this representation, the cycle is presumed to commence at the end of mitosis (0). The early period of the cycle (G_1) in which no DNA synthesis takes place is followed by a period of DNA synthesis (S). When the replication of the nuclear DNA has been completed, there follows a period of properation for mitosis (G_2), the cycle being concluded by the period of mitosis (N).

(Adapted from Stenners, G. P. and Till, J. E. (1960) Biochim. biophys. Acts. 37, 406.
such synthesis. Recent evidence suggests that the importance of this type of control may lie in its ability to impose a decision which, once taken, irrevocably commits the cell to proceed towards mitosis or, alternatively, towards specialization (Defendi and Wanson, 1963).

In the search for the molecular basis of such regulatory mechanisms certain types of cellular control system have been discovered which, spart from their intrinsic significence, provide a series of basic models applicable to several different kinds of met-bolic processes. These disoveries originated from studies on the fundamental interrelationship between DNA, RNA and protein; more particularly, the way in which information stored in the DNA genome is transcribed onto a strand of RNA, transferred via the "messenger" RNA to ribosomes and finally "translated" by synthesis of specific proteins (Astrachan and Volkin, 1958; Brenner, Jacob and Meselson, 1961). Studies of this type on the production of certain bacterial enzymes led Jacob and Wonod (1961) to propose a general hypothesis of the istracellular regulation of protein synthesis. In their view the DMA genome is divisible into several structural genes, each responsible for controlling the formation of an appropriate transpript in the form of measunger RNA which subsequently leads to the production of a specified protein. It is further assumed that transcription is initiated only at certain points or segments on the DNA strand. These points are termed operator genes, each of which may direct the function of several structural genes. In addition, the genome is assumed to involve a gene known as the regulator which has a nucleotide sequence similar to that of the operator. The regulator is

- 40 -

presumed to function by forming an RNA transcript, termed the repressor, which tends to associate reversibly with the operator thus blocking the initiation of transcription with resulting inhibition of the protein forming mechanism. Such a situation would seem to imply that the system is normally repressed. The function of the repressor is, however, presumed to be affected by the presence of certain small molecules, called effectors, so as to either impede or facilitate the combination repressor-operator. In cause where the presence of the effector leads to active protein synthesis, Monod, Jacob and Gros (1962) postulate that a repressor so modified can no longer associate with the operator and the system therefore permits transcription to take place. In other systems only the effector modified repressor is functional and the presence of the effector inhibits transcription. These two types of effector are termed inducers and metabolic repressors, respectively.

The major outlines of this scheme is now firmly based on experimental evidence, at least as far as bacterial systems are concerned, while the actual molecular mechanisms involved remain purely speculative. Superficially, it would appear that the type of control outlined above is more directly concerned with the biological expression of DNA function than with the biogenesis of DNA; however, since such a system would be responsible for regulating the production of the enzymes required for DNA synthesis, it seems possible that the two mechanisms are intimately connected.

This view is supported by current knowledge of the cellular control of enzyme catalysed reactions. A number of investigations have focussed

- 41 -

attention on a special type of regulatory mechanism, termed feedback control, which involves cases where the reaction substrate either activates the enzyme or induces the formation of new enzyme by its influence on the enzyme forming system (positive feedback) or, similarly, where a product either inhibits the initial enzyme involved in the reaction (negative feedback inhibition) or represses the formation of the enzyme (feedback repression) (Potter and Auerbach, 1959; Potter, 1962). Induction and repression are therefore essentially positive and negative feedback mechanisms. Feedback inhibition normally refers to the situation in which accumulation of the end metabolite causes a specific inhibition of enzymes cetalysing an early step in the metabolic sequence leading to its formation. The net result of this type of regulation is that excessive accumulation of a particular and product will result in decreased activity of the overall metabolic pathway leading to its synthesis. It is thought (Davidson, 1962) that this type of control mechanism may play a significant role in regulating the formation of the deoxyribonucleoside triphosphates, the immediate presursors of DNA. Negative feedback control slao appears to influence the operation of the pathways for de novo synthesis of the purine and pyrimidine nucleotides. Thus both ANP and OMP were found to inhibit the initial enzyme in purine synthesis, PRPP eminotrensferese (Wyngearden and Ashton, 1959; Magasanik and Karibian, 1960; Henderson, 1962); similarly, CMP (Yates and Pardee, 1956b), CTP (Gerhart and Pardee, 1962) and pyrimidine deoxyribonucleosides (Bresnick, 1963; Ennis and Lubin, 1963) have been found to exert negetive feedback control over the action of aspartete

- 48 -

transcorbamylase.

The control of cellular proliferation in mammalian tissues is currently being investigated in terms of the basic blochemical processes leading to the synthesis of DNA and the means by which critical steps in the blosynthetic sequence may be regulated by interaction of products. Although some relationships between enzymic reactions examined in growing and non-growing tissues have been observed, there is yet no indication of the critical point at which control is initiated or a mechanism by which it is operated. However, a considerable body of evidence appears consistent with the view that part of the regulatory mechanism, at least. is associated with the synthesis of deoxyribonucleoside triphosphates and there are some indications that the synthetic system responsible for the formation of thymidine triphosphate (TTP) may be rate limiting in the production of substrates for the DNA polymerase. Under these dircumstances. it seemed of value to re-examine some aspects of the enzymic reactions leading to the formation of TTP from thymidine and TMP using extracts from rapidly proliferating mammalian cells.

- 43 -

GHAPTER II

"STUDIES ON THE FORMATION OF THYMIDINE 5'-TRIPHOSPHATE WITH ENZYMES FROM LANDSCHUTZ ASCITES TUMOUR CELLS."

2.1. Introduction

(a) The existence of cellular pools of free nucleotides

The introduction and development in the early 1950s of chrometographic techniques made it possible to isolate and determine extremely small quantities of cellular components. The classical experiments of Cohn (1950, 1951) introducing column ion-exchange chrometography to the field of nucleic acids led to the identification and separation of the major nucleotide components of these polymers. In the following few years. these techniques were further developed by others and applied not only to nucleic acid digests but also to dilute acid extracts of cellular material. As a result of such studies, it became apparent that, is addition to the already well documented adenine nucleotides (Smbden and Limmerman, 1927). there existed in many cells cytoplesmic pools composed of a large variety of free nucleotides. Thus the 5'-mono-, the 5'-di- and the 5'-triphosphetes of sytidine, guenosine and wridine were found to occur in the acid soluble extracts of rat tissues (Schmitz, Hurlbert and Potter, 1954; Hurlbert, Schmitz, Brumm and Potter, 1954) and in extracts of other types of tissue (Schmitz, 1954; Bergkvist, 1956; Ballio, Gesinovi and Serlupi-Grescenzi, 1956). In the course of these investigations there appeared a number of other hitherto unsuspected nucleotides, the majority of which were coenzymes, so that by 1958, Henderson and LePage could report over 110 separate nucleotides isolated from biological sources. Measureable quantities of the pyrimidine deoxyribonucleoside 5' -mono, 5' -di- and 5' -triphosphetes were found in rat thymus tissue (Potter, Schlesinger, Beuttner-Janusch and Thompson, 1957; and in extracts of other rat tissue (Schneider, 1955;

- 44 -

Ord and Stocken, 1958) while no trace of purine deoxyribonucleotides could be detected. Determination of the pool size of a purine deoxyribonucleotide has been recorded by LePage (1957) who found small quantities of dATP in extracts of Flexner-Jobling carcinoma.

Potter (1960) considers that the nucleotides so far discovered follow certain rules and fall into a limited number of categories. In general, it appears that every nucleotide that occurs as a major component of nucleic moids will also be found as the 54monophosphate, 5'-diphosphate and the 5'-triphosphate in the sold soluble fraction, and that there is also likely to occur a series of compounds that can be considered as derivatives of these three forms and are likely to possess coenzyme function.

It seems clear that the presence of pools of free nucleotides in verious parts of the cell may facilitate the synthesis of nucleic acids. Sells, therefore, need not depend solely on <u>de novo</u> synthesis of nucleotides to sustein polymerisation but can command a store of nucleotide material from which nucleic acid precursors may be directed to the site of synthesis. However, it seems apparent that while material from such a nucleotide pool could be expected to support relatively extensive RMA synthesis, the cellular decoxyribonucleotide pool, frequently of the order of 0.1% of the total nucleotide content of the DNA, is very small in comparison and extensive <u>de</u> <u>novo</u> synthesis or conversion of ribonucleotides to decoxyribonucleotides is required to sustain the process of DNA replication. The measure of control of DNA synthesis exercised by the size of the decoxyribonucleotide pool hes been intensively investigated on synchronously growing cells of various types.

- 45 -

Stern (1961), using a system from <u>Trillium erectum</u>, has shown that the decxyribonucleotides of guanine, adenine, sytosine and thymine appear in the fluid surrounding the misrospores shortly before DNA synthesis commences. A similar situation appears to exist in developing amphibian embryos (Kuriki and Okazaki, 1959). However, in some rapidly dividing synchronous cell populations such as regenerating rat liver no significant accumulation of decxyribonucleotides is observed (Rotherham and Schneider, 1958) so that the ovidence for control of DNA synthesis by the size of the decxyribonucleotide pool remains unconvincing.

(b) Internucleotide metaboliam

The intracellular pools of nucleotides (see above) are composed of nucleotides in various states of phosphorylation, nucleosides and, sometimes, of free puring and pyrimidine bases; all must be expeble, when required, of transformation into a form suitable for incorporation into the nucleic acids. Present evidence suggests that these conversions occur in a non-random fachion and that the cell can direct nucleotide material entering the pool via the <u>de novo</u> sequences through certain specified pathways leading finally to the nucleoside 5'-triphosphates.

The existence of kineses cepable of phosphorylating AMP, GMP, dAMP, dGMP, UMP, GMP, dGMP and TMP (see Sections 1. 8 (d). and 3. 1.) to their respective triphosphates have now been described and it is generally presumed that the reactions occur in a stepwise manner involving the intermediate formation of nucleocide diphosphates. In addition to such sequential phosphorylations, several pathways for the interconversion of nucleotides

- 46 -

have been described, the primary function of which is to ensure the production of distal nucleic acid precursors at the prescribed site and at the required time. Excessive production of any one precursor appears to result in an increase in the catabolic activity relative to that precursor (Roth, 1963).

Wuch less is known about the interconversions occurring among the purine than pyrimidine nucleotides. Particularly as regards the formation of the purine decxyribonucleotides present knowledge is scenty and the scheme presented in Fig. 10 must therefore be regarded as somewhat tentative. However, the formation of the two RNA precursors ATP and GTP proceeds through well documented routes from IMP itself formed de novo, and interconversion between AMP and GMP through the cycles where the two nucleotides are reconverted to IMP has been demonstrated in many tissues (Guarino and Yuregir. 1959; Mager and Magasanik, 1960). Several investigations have pointed to the occurrence of deoxyndenosine kinases capable of forming dATP from AdR (Klenow and Lichtler, 1957; Maley and Maloy, 1960), but no indication emerged from such studies as to the way in which decayadenosine derivatives are derived from the ribonucleotide pocl. In a recent report, Larsson (1963) has shown that purified enzyme preparations from 3. coli and from Novikoff hepatome which catalyse the reduction of GDP, UDP and GDP, also reduce adenosine ribonucleotides to deoxyribonucleotides, presumably at the ADP level, but the evidence seems to suggest that ADP is not a primary substrate for the reduction system so that interconversion of adenine and AdR under the influence of a purine decryribonucleoside phosphorylass remains the only well established pathway (DeVerdier and Gould, 1963). Direct reduction of guanesine ribonucleotides to decxyribonucleotides at the diphosphate level

- 47 -

Fig. 10. The metabolism of purime ribonucleotides and deoxyribonucleotides in relation to the synthesis of nucleic solds.

> s-AMP = adenylosuccinic acid (6-(succinylamino)-9-(ribofuranceyl 5'-monophosphate) purine).



Fig. 10.

has been demonstrated with extracts of chick embryos and preparations from <u>E.coli</u> (Reichard, 1961) and decxyribonucleoside phosphorylases ontalysing the conversion of guanine to GdR have been reported. Finally, Maley and Maley (1961a) have found evidence of a direct conversion in chick embryo mince of dAMP to dGMP. At present, there is little indication as to which of these interconversions, if any, represent the main intracellular pathway for the formation of the purine decxyribonucleotides.

A somewhat clearer picture has emerged over recent years of the interconversions operating in the pyrimidine nucleotide pool. Several pathways appear to promote the formation of the pyrimidine deckyribonucleotides, all arising by ramifications of the route leading from UNP formed de novo, and the decxyribonucleotides so formed are subsequently phosphorylated to the level of phosphorylation required by the polymerising system (see Fig. 11). The conversion of UNP to GMP by amination at the UTP level has already been described (see pib). However, in many systems, it seems probable that the conversions of ribonucleotide to deoxyribonucleotide are critical steps and responsible for the bulk of the cellular DNA precursor meterial. naymes estalysing the reduction of GDP to dGDP have been demonstrated in a variety of bacterial (Reichard, 1962) and memmalian (Moore and Hurlbert, 1962; Abrams, Libenson and Edmonds, 1960) (see also p.24 cell extracts. Conversion of dGMP to dUMP has been shown to be prominent in several types of rapidly growing mammalian cells (Scarano, 1960; Schneider and Rotherham, 1961; Maley and Maley, 1961b) while some recent evidence points to the possibility that embryonic tissues may utilise an alternative

- 48 -

Fig. 11. The metabolism of pyrimidine ribonucleotides and deoxyribonucleotides in relation to the synthesis <u>de novo</u> of DMA.

> Several alternative pathways are illustrated, only a limite mumber of which may operate in any one type of cell, whi remetions of mainly optabolic significance have been disregarded.

> The insertion illustrates the changes in the metabolism of cytosine decxyribonucleotides resulting from infection of <u>Recherichia coli</u> with besteriophage T2.



desmination reaction. Maley and Maley (1962a) have shown that extracts of ret embryos cetalyse the incorporation of 140 from 140-labelled GdR into DNA thymine to a greater extent than into DNA cytomine and in subsequent experiments they (Maley and Maley, 1962b) have indicated that the utilisation of GdR in several organisms involve a direct demmination of GdR to UdR followed by a phosphorylation step to give dUMP. Bacterie appear to lack the ability to form dUMP from dGMP. Bessman at pl. (1958, observed that the purified E. coli DNA polymerase could incorporate dUTP into DNA in place of TTP, but since no uracil was found in the "natural" DNA it was thought that its absence was due to the lack of kineses capable of phosphorylating dUMP to dUTP. Using cell-free extracts of the same organism, Reichard and Rutberg (1960) provided evidence for a direct reduction of uridine ribonucleotides to uridine decayribonucleotides thus indicating the existence of a third pathway for the production of dUMP. The purified CDP -> dCDP reductase system from E. coli has recently been shown to act also on UDP resulting in the formation of dUDP (Bertani, Haggmark and Reichard, 1961, 1963). In the view of Haggmerk (1963) the dUDP so formed is repidly transformed by a kinase to dUTP followed by a specific pyrophosphatase ontalysed breakdown to dUMP. A series of kinases capable of phosphorylating dUMP to dUDP and dUTP has recently been reported in T2-infected R.coli, but again production of dUTP was controlled by the presence of a dUTP specific pyrophosphatese (Greenberg and Somerville, 1962). The precise sequence and mechanism of these reactions remain uncertain, but they appear to provide an efficient means of producing dUMP in systems where desminases catalysing

- 49 -

either domp -> dump or odR -> UdR are lacking.

That <u>de novo</u> formation of the thymidine nucleotides occur principally through the methylation of dUMP to TMP is now well established and the mechanism of the reaction has been elucidated in great detail by the use of highly purified enzyme properations and sophisticated analytical techniques. Sarly studies on calf thymus extracts (Blakley, 1957) suggested that UdR was a better substrate then dUMP in the reaction, while more highly purified properations (Greenberg, Nath and Humphreys, 1961) established dUMP as the acceptor of the one-carbon unit; formaldehyde and tetrahydrofolate were required as co-factors. The purification of an <u>R. coli</u> thymidylate synthetese with similar properties has been described (Webbe and Friedkin, 1962).

An alternative pathway which at one time was thought to contribute to the thymidine nucleotide pool was the direct formation of TMP from 5-Me-ddMP by demination. In experiments with see urchin eggs, Scarsno, Bonaduce and DePetrocellis (1960) showed that 5-Me-ddMP was the preferred substrate for the ddMP deminase of that tissue. Together with earlier methylation experiments (Scarsno and Maggio, 1959) which demonstrated the tetrahydrofolate dependent formation of 5-Me-ddWP from ddMP, the evidence suggested an alternative to the dDMP \rightarrow TMP pathway. However, several subsequent experiments by Maley and Maley (1961c) have failed to reveal the presence of a 5-Me-ddMP synthetese in other similar tissues and they conclude that the pathway can at beat be a very minor one in comparison with the ddMP \rightarrow TMP conversion. It is also important to note that formation of the 5-Me-ddMP of wheet germ DMA has been demonstrated

- 50 -

to occur by methylation at the polynucleotide level (Gold, Hurwitz and Anders, 1963a).

Some of the changes in the sequence and direction of nucleotide interconversions that can occur in response to viral infection are nicely demonstrated by the example of the T2 bacteriophage-infected E. coli system. In the TS phage DNA, the dOMP residues are replaced by 5-MeOH-dGMP residues and it seems clear that the latter are not the result of hydroxymethylation in situ but are instead incorporated from the corresponding 5'-triphosphate. A series of new enzymes is sequentially induced in the infected system to secure the formation of this material while excluding dCTP from being incorporated. In this new situation, dGTP is being continuously degraded to dOMP by a newly formed specific pyrophosphatase (Kornberg, Zimmerman, Kornberg and Josse, 1959). This is followed by the hydroxymethylation of dGMP by enother new enzyme which requires tetrahydrofolate (Flaks and Cohen, 1959) and the ensuing 5-MeOH-dCMP further leads to the synthesis of 5-MeOHdCTP under the influence of a new kinase system (Somerville, Ebisuzaki and Greenberg, 1959). The main alterations in internucleotide metabolism due to T2 infection of an E. coli system is shown in the insertion on Fig. 11.

(c) Formation of nucleotides by preformed pathways

Even before the discovery of the <u>de novo</u> pathways of purine and pyrimidine ribonucleotide synthesis, it was well known that many types of cell were capable of utilising exogenously supplied purine and pyrimidine bases and their nucleosides for nucleic acid synthesis. Bacteria were generally observed to be more active in this respect than higher organisms and later

- 51 -

experiments have shown that animal cells frequently cause rapid breakdown of such compounds (Brown and Roll, 1955).

Present evidence minits the existence of at least three types of enzymic reaction which together constitute the pathways for utilisation of preformed purime and pyrimidime bases. Using adamine as the example, the following reactions appear to apply:-

(a) Ribonucleoside phosphorylase

- Adenine + ribose-l-phosphate = adonosine + orthophosphate
- (b) <u>Ribonucleatide</u> pyrophosphorylase

Adenine + PRPP = edenosine 5'-monophosphete + pyrophosphete (c) Ribonucleoside phosphokinase

Adenosine + ATP adenosine 5'-monophosphate + ADP

In general, the equilibria of these reactions appear to favour degredation, but under special diroumstances when the <u>de novo</u> pathways are not effective these reactions offer an alternative means by which the cell can acquire nucleotide precursors for nucleic acid synthesis (see review by Friedkin and Kalcker, 1961).

The specificity of nucleoside phosphorylases is not yet well documented but there can be no doubt that the purine phosphorylases differ from the pyrimidine phosphorylases. Thus the purified purine ribonucleoside phosphory lase from horse liver (Korn and Buchaman, 1955) formed adenosine, guanosine and xanthosine from adenine, guanine and xanthine, respectively, but was without action on pyrimidines. Several relatively specific nucleoside phosphorylases have been reported to act on uracil and thymine without being active on purines (Friedkin and Roberts, 1954; Pontis, Degerstedt and Reichard, 1961). Ribonucleotide pyrophosphorylase activity has been demonstrated with adenine, guanine and hypoxanthine as substrates (Kornberg, Lieberman and Simms, 1950b) and a similar enzyme ontelysing the production of UMP from uracil has been reported (Kornberg, 1957); sytosine and thymine were not utilised (see also Fig. 8).

A further two reactions of a somewhat more tenuous nature utilising the free bases have been reported. One is the deoxyribosyl phosphorylese observed in extracts of rat liver (Friedkin, 1950) (d) and the other, the trans-N-deoxyribonucleosidese described by MacNutt (1958) (e) which catalyses an exchange of base moieties of deoxyribonucleosides without forming the free deoxyribose-1-phosphate.

(d) Decayribonucleoside phosphorylase

Adenine + deoxyribose-1-phosphate + = deoxyadenosine + orthophosphat

(e) Trans-N-deoxyribonucleosidese

Adenine + deoxyuridine 🖚 deoxyadenosine + uracil

The importance of reaction (e) may be to offer yet another alternative pathway for nucleotide interconversion. This was illustrated by the use of besterial extracts (Beck and Levin, 1962) where vitamin B_{12} deprived sultures showed greatly enhanced trans-N-deoxyribonucleosidase activity between deoxyadenosine and thymine, presumably because the lack of vitamin B_{12} caused an inhibition of the direct reduction of ribonucleotides to deoxyribonucleotides. DeVerdier and Potter (1960) described a similar activity in rat liver but could not resolve the relation between transference activity and that due to reaction (d). Their work was continued by Zimmerman (1962, 1963) who purified extensively the thymidine phosphorylase from rat liver and showed that the ratio of transferase to phosphorylase activity did not elter throughout the purification thus suggesting that the two activities were the property of a single enzyme.

(d) Thymine ribonucleoside and thymidine 5'-diphosphate coenzymes

A large number of investigations have rested on the assumption that the thymidine nucleotides operate exclusively as procursors of DNA. However, the recent observation that thymine occurs in two other classes of compounds has drawn attention to the possibility of eltern tive thymine utilisation in some organisms. These two types of derivative are the thymine present in S-RNA and the thymidine 5'-diphosphate coennymes. From the point of view of the thymidime derivatives available at any time for incorporation into DNA, it is clearly important to consider to what extent such alternative utilisation may effect the progress of replicative DNA synthesis.

Trace amounts of thymine was reported (Dunn, 1959) in bacterial RNA proparations and it has subsequently been found in several other types of tissue (Dunn, 1961) principally in association with g-RNA. Enzymic degradation of such preparations revealed that thymine appeared in non-terminal positions in the polyribonucleotide chain and that the nucleotide unit was in fact derived from thymine ribonucleoside. The biosynthesis of this latter compound had been reported by DeVerdier and Potter (1960) in an exchange reaction between thymine and uridine catalysed by extracts of rat

. 54 -

liver and Dunning hepstoms and thus indicating the existence of a trans-Hribonucleosidese; with liver preparations this reaction was more prominent that the phosphorylase activity. A similar type of reaction has been observed with thymine requiring mutants of <u>E. coli</u> (Manstevinos and Zamenhof, 1961). The further utilisation of thymine ribonucleoside remains a mystery since it seems clear that no kinnses analogous to the thymidine kinnses espeble of forming the thymine ribonucleoside 5'-triphosphate can be demonstrated in extracts of <u>E. coli</u>. The preformed 5'-triphosphate, on the other hand, has been shown to be incorporated into RNA specifically in place of UMP using a purified DNA dependent RNA polymerase (Kahan and Hurwitz, 1962)

Another approach to the problem was adopted by Mandel and Borek (1961) who in the course of studies on the methionine requiring suxotroph <u>E.coli</u> $K_{12}W-6$ observed that the methyl group of RNA thymine appeared to be derived from methionine without involving the intervention of dUMP or derivatives of tetrahydrofolic acid. In subsequent experiments, they (Mandel and Borek, 1963a) showed that when the methionine requiring auxotroph was grown in a medium containing ¹⁴G-methyl labelled methionine, the specific activity of the methionine used and the specific activity of the methylated components ultimately derived from the **g-ENA**, were identical, thus demonstrating a direct transfer of the methyl group from methionine to the RNA. Thus it would appear that the thymine in RNA and the same base intended for incorporation in DNA are produced by entirely different pathways which would therefore not compute for the cellular pool of thymidime nucleotides.

The origin of all the methyl groups of g-RNA from the same pool and

- 55 -

the synthesis, under certain conditions (Mandel and Borek, 1963b), of S-RNA lecking these minor components pointed to the possibility that methylation occurs in situ after synthesis of the polynucleotide. The existence of ensyme systems which cetalyze the methylation of S-RNA at the polynucleotide level has been demonstrated in bacterial extracts (Fleisener and Borek, 1962; Svensson, Boman, Briksson and Kjellin, 1963). Although a specific function has not yet been ascribed to the minor, methylated base, component of S-RNA, it now seems certain that methylation proceeds in a non-random manner and exhibite clear species specificity thus suggesting that the methylated bases may be another criterion for distinguishing S-RNA molecules (Srinivasan and Borek, 1963; Gold, Hurwitz and Anders, 1963b). Recent work has indicated that the nucleolus may be the site of S-RNA methylation (Birnstiel, Fleissner and Borek, 1963).

A further class of thymine derivative was discovered in extracts of <u>E. coli</u> in 1959 by Okazaki, Okazaki and Kuriki. The isolated compounds appeared to be thymidine-linked sugars and the first two were tentatively identified as TDP-D-glucose and TDP-D-rhamnose (Okazaki, 1959). Following the discovery of these new thymidine nucleotides, a number of additional TDP-linked sugars and aminosugars were identified from several types of becteria and plants while none could be detected in animal cells. A number of enzyme systems have recently been described catalysing the formation of thymidine-linked sugars from thymidine 5°-triphosphate and e variety of phosphorylated sugars, including a-D-glucose-l-phosphate (Kornfeld and Glaser, 1961; Pasur and Shuoy, 1961; Kaline and Avigad, 1963),

- 56 -

a-D-galactose-1-phosphate (Pazur and Anderson, 1963; Frydman, Neufeld and Hassid, 1963) and a-D-glucosamine-1-phosphate (Kornfeld and Glaser, 1962). In addition, it has been shown that interconversion of sugars may occur while linked to TDP indicating the coanzyme function of the nucleotide; thus both TDP-glucose (Okazaki, Okazaki, Strominger and Michelson, 1962) and TDPgalactose (Pazur and Anderson, 1963) are convertible to TDP-rhamnose through certain relatively well defined pathways. Almost in every case the sugars linked to thymidine are the same as those reported as linked to uridine in other organisms and it is probable that these coanzymes are in some way involved in forming the cell well lipopolysaccharides of those organisms (Okazaki, Strominger and Okazaki, 1963).

Several features of these reactions remain very doubtful but it is evident that in organisms where such TDP-linked compounds occur, there will exist alternative pathways competing for the cellular pool of thymidine nucleotides. However, no TDP coenzymes have yet been reported to occur in examplian cells and it seems clear that such cells, at least, the thymidine nucleotide pool functions specifically in the role of providing precursors for DNA synthesis.

(e) Formation of thymidine nucleotide phosphates

Cursent concepts on the synthesis of DNA prescribe: the decayribonucleoside 5'-triphosphates of adenine; guanine, cytosine and thymine as substrates in the polymerisation reaction. However, in many systems purime and pyrimidine decayribonucleosides and decayribonucleoside 5'-monophosphates have been shown to be utilised in the synthesis of polydecayribonucleotides

- 57 -

provided the requirements for ATP and Mg^{**} are fulfilled, and it seems probable that deoxyribonucleoside and deoxyribonucleotide presursors present in such systems are converted to deoxyribonucleoside triphosphates prior to incorporation into the DNA. The formation of triphosphates in such systems have been attributed to a series of phosphokineses which entelyse the transfer of phosphate from ATP to nucleoside and nucleotide substrates (see Section 1.2 (d)). In recent years, considerable interest has centred on the cellular activities of the deoxyribonucleotide kineses as possible controlling or rate regulating factors in DNA synthesis.

Kinases capable of phosphorylating dGMP and TMP to dGTP and TTP have been described in extracts of regenerating ret liver (Heaht, Potter and Herbert, 1954) and Genellakis and Manstevinos (1958) reported that such extracts also catalysed the formation of dATP and dOTP from dAMP and dOMP. However, the interesting observation was unde that while regenerating ret liver clearly could supply the full complement of decoryribonucleotide kinese activities necessary for the formation of all the four triphosphetes, normal liver appeared to be deficient or lacking in its ability to phosphorylate TMP (Manstevinos and Genellakis, 1959). The regenerating ret liver system of Bollum and Potter (1959) and cell-free supernatant fluids derived from escites tumour cells (Smellie, Keir and Davidson, 1959) have been shown to utilise thymidine in the synthesis of DMA <u>in vitro</u> thus indicating the presence of thymidine kinese, in addition to decayribonucleotide kineses (Keir and Smellie, 1959), in these tissues.

The ability of enzyme preparations from ascites tumour cells to catelys

- 58 -

the formation of TTP from TdR or TMP is very limited (Keir and Smellie, 1959; Grey et al., 1960; Ives, Morse and Potter, 1968) particularly in comparison with their capacity to form dATP, dGTP and dGTP (Keir and Smellie, 1959; Grey et al., 1960). Similarly, while extracts of non-proliferating tissues contain active enzymes catalysing the formation of dATP, dOTP and dGTP from the corresponding deoxyribonucleoside 5'-monophosphates, the activity of the kinase system responsible for the formation of TTP is invariably low (Grey et al., 1960). Resever, the universality of this type of correlation is in some doubt since the TMP kinase activity in extracts of E. <u>coli</u> was found to be independent of the growth rate of the culture (Bessman, 1959).

Fractionation experiments on the thymidine and thymidylate kinases of ascites cells led Weissman, Smellie and Paul (1960) to suggest that the formation of TTP from TdR occurred by the stepwise addition of orthophosphate forming successively TMP, TDP and TTP thus:-

TAR = THP = TDP = TTP

Howaver, from experiments on the time course of phosphorylation of TdR, Bianchi, Butler, Grathorn and Shooter (1981) have proposed that the mechanism involves the addition of orthophosphate to TdR to yield TMP followed by the addition of pyrophosphate to yield TTP directly. TDP observed in the reaction products was considered to arise from the degradation of TTP after prolonged incubation. This mechanism may be illustrated as follows:-

TAR = TP TTP TDP

- 59 -

The present investigation represents an attempt to cherecterise relatively fully the thymidine kinase and thymidylate kinase systems derived from cell-free extracts of Landschutz ascites tumour cells and to distinguish between the two reaction mechanisms postulated for the ensymic conversion of TdR and TMP to TTP.

8.2. Experimental

2.2.1. Preparation of crude cell-free extracts from Landschutz ascites

tumour cells

Orude extracts of landsbutz escites-tunour cells were prepared by the procedure of Smellie, Keir and Devidson (1959) with some minor modifications. Landschuts escites cercinome was meintained by serial transplantation in albino mice of the departmental colony and harvested 8 - 9 days after inoculation. The ascitic fluid was withdrawn from several animals under aseptic conditions and pooled in an ice-cooled flask. The fluid was centrifuged at 200 - 300 x g for 5 minutes to separate the cells from the plasma. The resulting sediment was resuspended in 5 - 10 volumes of chilled 0.1 M phosphete buffer, pH 8.1 and again centrifuged at 200 - 300 x g to separate the tunour cells from crythrocytes. This washing procedure was repeated several times until the sediment of tumour cells was free from erythrocyte contamination. The washed cells were again resuspended in phosphete buffer and centrifuged at 600 x g for 10 minutes to pack them tightly in the centrifuge tube. To this pellet of cells was added 10 - 15 volumes of ice-cold distilled water and the cells were then disrupted by gentle pessage in a Potter-type homogeniser (Potter and Elvehjelm, 1936). 3 - 4 Bases of the material were sufficient for adequate breakdown of cells. Microscopic examination with the aid of a solution of crystal violet in citric acid was used as control of the efficiency of the process. The procedure effected rupture of most cells without destroying many nuclei.

The resulting suspension was centrifuged in a Spinco Model L ultracentrifuge at 105,000 x g for 90 minutes to yield a clear extract. The extract thus prepared had a protein concentration of 3 - 6 mg. per ml. In some experiments, this crude extract was used immediately as a source of thymidize and thymidylate kinases but, in general, these enzyme activities were preserved by rapid lyophylisation of the extract and storage at - 50°.

2.2.2. Chemical preparation of (32p) - labelled thymidine 5'-phosphetes

(a) Preparation of (32p) - labelled thymidine 5"-monophosphate ((32p) TMP)

 (^{38}p) TMP was prepared by the method of Tener (1961) from thymidine and (^{38}p) - labelled 2-ayanoethylphosphete. The latter compound was either obtained commercially (The Rediochemical Gentre, Amersham) or prepared in the laboratory from carrier-free (^{38}p) - labelled orthophosphoric acid and 2-cyanoethanol in the presence of dicyclohexylcarbodiimide (DGG). An outline of the reaction is given on Fig. 12, Stage I.

Preparation of (32p) 2-cyanoethylphosphate ((32p) GEP)

A solution containing 50 ms of carrier-free (^{35}P) - labelled orthophosphete was mixed with 1 mmole of unlabelled phosphoric acid in dilute acid solution to permit randomisation of (^{35}P) . This solution was concentrated to dryness in <u>provo</u> in a retary evaporator at 40° to remove traces of hydrochloric acid. The resulting labelled phosphoric acid was dissolved in 10 ml. of anhydrous pyridine (freshly distilled and stored over calcium hydride) and 1 ml. of 2-symmothemol was added. The solution was then evaporated to an oil <u>in vacuo</u> at 40°. The residue was redissolved in a



Fig. 12. Outline of the chemical synthesis of decxyribonucleoside 5'-polyphosphetes.

> Stage I: The formation of 2-cyanosthylphosphate from 2-cyanosthanol and orthophosphate.

Stage II: The synthesis of decxyribonucleoside 5'-monophosphates by phosphorylation of decxyribonucleosides with 2-cyanoethylphosphate.

Stage III: The formation of deoxyribonucleoside 5'-triphosphetes from the corresponding 5'-monophosphetes and orthophosphete.

The symbols a, β and γ illustrate the nomenolature employed in describing radioactive isomers of decayribonucleoside 5'-polyphosphates. further 10 ml. of anhydrous pyridine and again concentrated to an oil.

This oily meterial was dissolved in 5 ml. anhydrous pyridine and 2.1 gm. of disyclohexyloarbodiimide (DGG) was added. The reactants were carefully mixed and the securely stoppered reaction flask left overnight at room temperature.

Distilled water (5 ml.) was added to terminate the reaction and the resulting mixture heated on a boiling water-bath for 30 minutes. The contents of the flack was then taken to dryness in vacuo at 40° and 10 ml. of distilled water and 10 ml. of seturated barium hydroxide added to the residue. After standing for 10 minutes at room temperature, the pH of the solution was adjusted to 7.5 with glacial agetic agid. The mixture was then filtered at the pump to remove the precipitate containing dicyclohexylures (DCU) and barium phosphate. The precipitate on the filter was thoroughly washed with distilled water and the washings added to the bulk of the solution. Two volumes of 95% ethenol was added to the solution to precipitate the (32P) barium 2-cyanosthylphosphate. The solution was stirred and allowed to crystellise for one hour at 0°. The ORP crystels so obtained were collected by centrifugation and the ethenol supermetant was discarded. The crystals were rediscolved in 5 ml. of distilled water with the addition of a few drops of dilute acetic acid. The resulting solution was neutralised with saturated barium hydroxide and any trace of undissolved material removed by contrifugation. The GSP was reprecipitated with two volumes of ethanol and the mixture transferred to a pre-weighed centrifuge tube. The GEP was then washed with ethanol, scetone and finally other and allowed to dry at room temperature. The amount of GEP so

prepared (Be salt, monohydrate) as determined by reweighing of the test tube was 200 mg. This represents a yield of 65 per cent.

The barium selt of GEP must be converted to the free acid prior to its use in the synthesis of TMP. This was achieved by rediscolving the orystals in a small volume of distilled water (10 ml.) using sectic acid to assist solution. This solution was then passed through a column of Dower-50-H⁴ (10 a 1 cm., dia.) (for preparation, see Section 2.2.6 (c)) and the column washed with water until most of the redioactivity was in the effluent. This effluent now contained the free acid of GEP contaminated with some acetic acid. The solution was finally taken to dryness in vacuo at 40⁶ to remove the soutic acid and the residue discolved in pyridine.

Preparation of (38p) THP

The formation of TMP from thymidine and 2-oyencethylphosphate is outlized on Fig. 12, Stage II. For the preparation of (³²P) TMP, the following reaction mixture was set up:

Thymidine (2 mmoles) was dissolved in pyridine and 1 mmole of (^{32}P) GEP (prepared as described above) in pyridine was added. When only smaller amounts of 2-symmoethylphosphate were available, the molar proportions of thymidine : GEP of 2 : 1 were maintained. The reactants were thoroughly mixed and the solution concentrated to an oil in <u>vacuo</u> at 40°. Anhydrous pyridine (10 ml.) was then added and the solution again concentrated to an oil <u>in vacuo</u>. This procedure was repeated one more time and the residue was dissolved in 10 ml. of anhydrous pyridine to which was added 680 mg. of DGG. The flask was shaken until all DGC had dissolved, then securely stoppered and left for 18 hours at room temperature.

The reaction was terminated by the addition of 5 ml. of distilled water and, after 30 minutes at room temperature, 10 ml. of concentrated annonium hydroxide (specific gravity 0.880) was added and the solution heated at 60° for one hour. The resulting mixture was concentrated to dryness in yeauo at 40° and the residue extracted with 10 ml. of distilled water. Removal of the precipitated DGU was effected by filtration at the water pump. The DGU was washed carefully on the filter with several small volumes of distilled water, the washings being combined with the original filtrate.

At this stage a sample of the filtrate was taken for descending chromatography socording to the method described elsewhere (see Section 2.2.3 (b)) to assess the relative proportions of the thymidine, thymidine 3'-monophosphate and thymidine 5'-monophosphate present. An approximate assessment of the thymidine 5'-monophosphate produced could be obtained by elution of all the ultraviolet light absorbing spots by capillary flow in water and measuring their respective extinctions at 267 mm (pH 2); the percentage of TMP-5' relative to the total extinction could thus be estimated.

The combined filtrate and washings (see above) was then run on to a column of Dower-1-G1" (SO x 2 cm., die.) and subjected to gradient elution. Removal of unreacted nucleoside was effected by washing repeatedly with distilled water until the extinction at 267 mµ in the effluent approached zero. THP was then recovered from the column by repeated washings with

- 65 -

0.05 N hydrochloric acid. Some radiosctivity was retained at the top of the column after this treatment and this was discarded since it probably contained a mixture of radioactive orthophosphate, polyphosphate and thymidine polyphosphates.

Approximately 60% of the TMP in the reaction products was TMP-5' and the remainder was accounted for as TMP-3'. TMP-3' is not phosphorylated in the chemical preparation of TTP as described below and may therefore be regarded as an undesirable contaminant. Similarly, in experiments where (32p) TMP itself was used as substrate for the thymidylate kinases steps were taken to remove contaminants and the TMP-5' was separated from TMP-3' as follows: The bulk of the effluent from the Dower-1-C1 column (see above) was diluted to 500 ml. and adsorbed on a second column of Dowey-1-G1 (20 x 1 cm., dis.) and the column washed with water to remove traces of ultreviolet light absorbing unadsorbed material. The THP-5' and THP-3' were sluted from the column by gradient elution with 1000 ml. of distilled water in the mixing vessel and 0.06 N hydrochloric acid in the reservoir. collecting fractions of 10 ml. with an elution rate of 0.5 ml. per minute. The gradient so obtained was non-linear and similar to that described by Hurlbert, Schmitz, Brumm and Potter (1954). Two radioactive peaks possessing extinction at 267 mu were recovered, the first of which contained the THP -5' and the second contained the THP-3'. Complete separation of these two peaks was achieved by this procedure as shown in Fig. 13. Fractions from both peaks were separately combined and evaporated to dryness in vacuo at 25°. The residues were each extracted with 5 ml. of distilled water and the

- 66 -




Fig. 13. Separation of the remotion products in the chemical synthesis of (³²P) -labelled thymidine 5'-monophosphetes from (³²P) 2-cynnoethylphosphate and thymidine.

The nucleotide mixture was adsorbed on a column of Dowex-1-G1⁻ (30 x 1 cm., dis.) and elution was effected by applying to the column a non-linear gradient with 750 ml. distilled water in the mixing vessel and 0.06 <u>N</u> hydrochloric acid in the reservoir. Fractions of 8 ml. were collected at a column flow rate of 1 ml. per minute. The eluted fractions were scanned spectrophotometrically at 267 mu. Three peaks were obtained representing TdR, TMP-5' and TMP-3', in that order. Fractions 85 - 50 were combined and found to contain 12.8 µmoles of TMP-5' while fractions 57 - 107 contained 7.6 µmoles of TMP-3'; the recovery of totel nucleotide material was 95 per cent. resulting solutions passed through solumns of Dower-50-Na⁺ (5 x 1.5 cm., dia.). The effluents, containing the solium selt of TMP-5' or TMP-3', were collected and the total amounts of each nucleotide was determined after appropriate dilution by measurement of the extinction at 267 mµ. The yield of TMP-5' was 281 µmoles and that of TMP-3' lllµmoles, representing an overall yield of thymidine monophosphates of 39%. The specific activity was lo3 x 10⁶ counts per minute per µmole as determined in a Nuclear-Chicago windowless gas-flow counter.

The purity of the preparation was shocked by paper chromatography in ammonium isobutyrate solvent (Section 2.2.3 (b)) with subsequent detection of orthophosphate and bound phosphate (Section 2.2.3 (c)) and of the principal regions of radioactivity on appropriate lanes out from the chromatograms (Section 2.2.3 (d)).

Other methods for the elimination of contamination with the 3'-isomer were investigated and, in some phosphorylations, thymidine was replaced by 3'-O-abetyl-thymidine propared by the method of Michelson and Todd (1953). The potential advantage of employing a nucleoside with blosked 3'-hydroxyl group was that mucleophilic attack by the phosphorylating agent could be expected to result in phosphorylation of the 5'-hydroxyl group only. It was found, however, that this modification led to a proliferation of byproducts, many of which were of uncertain composition and led simultaneously to a depression in the total yield of TMP-5'. The method was therefore not used for routine proparations.

- 67 -

The method employed was essentially that described by Smith and Khorana (1958) with some minor modifications. Thymidine 5'-triphosphate proximally labelled with (³²P) was prepared by condensation of (³²P) THP with unlabelled orthophosphoric acid in presence of DGG (see Fig. 12, Stage III). The following reaction mixture was prepared:

per 100 µmoles of (32p) TMP (free soid form),

1.8 ml. of tri-n-butylamine,

6.0 ml. of pyridine,

0.2 ml. of 85% orthophosphoric acid,

3.0 gm. of dicyclohexyloarbodiimide (DCG).

The reactants were dissolved in the order shown and the contents of the flask thoroughly mixed. The flask was then securely stoppered and allowed to stand at room temperature for 48 hours.

After a few hours, a thick precipitate of DGU formed and the reaction mixture gradually solidified. The reaction was finally terminated by the addition of one volume of distilled water, the contents of the flask were agitated, breaking up the solidified reaction products, and then allowed to stand for one hour at 0°. The DGU sediment was removed by filtration through a Buchner funnel and the residue was Grafully washed on the filter with distilled water which was subsequently added to the bulk of the filtrate. The solution was extracted with several small volumes of ether to remove pyridine. Traces of nucleotide material were, in turn, recovered from the combined ether phase by washing twice with small volumes of distilled water and adding the washings so obtained to the main aqueous phase. The ether extracts were then discarded and the equeous solution was concentrated to dryness in vacuo at 25°.

The residue was discolved in distilled water and the solution persed through a column of Dower-50 -Na* (10 x 2 cm., dia.) to remove tri-n-butylemine and traces of pyridine and, at the same time, to convert the mixture of deoxyribonucleotides from the pyridinium to the sodium salts. The resin was repeatedly washed with distilled water until the ultraviolet light absorption of the effluent had reached a low level and most of the radioactivity had been recovered. On the basis of the extinction at 267 mu the recovery of thymidine nucleotides was of the order of 95 - 98 . The water washings and the original effluent were combined and diluted to 1000 ml. with distilled water to give a solution of pH 6 - 7. This solution was run slowly on to a column of Dowex-1-G1 (15 x 2 cm., die.) thus adsorbing the nucleotide meterial present. When the adsorption was complete, the nucleotides were divested of any decryribonucleoside contamination by washing the column bed repeatedly with distilled water until the extinction at 267 mu in the effluent approached zero.

Resolution of the nucleotide mixture was effected by gradient elution with hydrochloric acid and lithium chloride. A non-linear gradient (Hurlbert, Schmitz, Brunn and Potter, 1954) was employed with a single mixing vessel containing 0.01 N hydrochloric acid where the volume of hydrochloric acid was determined by the total emount of material to be separated. Thus, for amounts up to 100 µmoles of nucleotides, the volume of HGl was 1000 ml. for amounts up to 500 µmoles 1500 ml. and for amounts up to 1 mmole a volume of 2000 ml. was used. The reservoir contained 0.3 M lithius chloride in 0.01 N hydrochloric acid. The elution rate was adjusted to 1.5 - 8.0 ml. per minute and fractions of 20 ml. were collected. The fractions so obtained were scanned spectrophotometrically at 267 mm and the extinction plotted against the fraction number; alternatively, redicactivity was plotted against fraction number. The final elution pettern revealed three definite and well separated peaks corresponding to thymidine 5' -monophosphate thymidine 5'-diphosphete and thymidine 5'-triphosphete in thet order as illustrated in Fig. 14. All the three peaks were shown to be essociated with redicactivity. The final peak containing 60 - 80% of the total radioactivity represented the (32P-a) TTP and all fractions from this peak were combined and the amount of triphosphate estimated, after appropriate dilution, by the extinction at 267 mu.

Contamination of the triphosphete with inorganic polyphosphetes necessitated further purification. The solution containing the TTP was therefore treated with 30 ml. of a suspension of acid-washed animal charcoal (see Section 2.2.6 (a)) and the mixture shaken occasionally and stored for 18 hours in the cold to complete the adsorption. The charcoal was subsequently packed into a column (10 x 1 cm., dia.) and the column bed washed liberally with distilled water until the effluent was free from chloride ions. To remove contamination due to inorganic ortho- and pyrophosphete, washing

- 70 -



Fig. 14. Separation of reaction products in the chemical synthesis of (³²P) -labelled thymidine 5'-triphosphate from (³²P) thymidime 5'-monophosphate and un-labelled orthophosphate.

> The nucleotide mixture was adsorbed on a column of Dower-1-01" (15 x 2 cm., dia.) and elution was effected by applying to the column a non-linear gradient with 2000 ml. 0.01 <u>M</u> H31 in the mixing vessel and 0.3 <u>M</u> lithium chloride in 0.01 <u>M</u> HGl in the reservoir. Fractions of 20 ml. were collected at a flow rate of 2ml. per minute. Three peaks representing TMP, TDP and TFP, in that order, were obtained. Fractions 145 - 210, representing the third peak, were combined and found to contain 240 µmoles (³²P) TTP, the over-all yield being 60 per cent.

was commanded with 0.01 <u>M</u> sodium bicerbonste and this treatment was continued for about 10 column volumes or until ultraviolet light absorbing material began to appear in the effluent. At this stage the elution with bicerbonate was terminated and the column allowed to drain. The charcoal bed was then treated with two separate column volumes of distilled water and allowed to drain each time thus removing traces of bicerbonate. TTP was finally recovered from the column by elution with 0.14 <u>M</u> ammonium hydroxide in 70% ethanol and the elution continued until the extinction in the effluent at 267 mm fell to a low value. A recovery of 90 - 95% of the adsorbed TTP was achieved by the application of 15 - 20 column volumes of eluent.

The combined ethenol - ammonia eluste thus obtained was concentrated to dryness in vacuo at 25° to remove the ethanol and excess ammonia and the residue of TTP, now in the form of the ammonium selt, was converted to the sodium selt by extraction of the residue with a small volume of water and passing this extract through a column of Dowex-50-Me⁺ (5 x 2 cm., dia.). The (32 P-c) TTP (sodium selt) was then either lyophylised directly and stored at -50° or made up to a suitable concentration and stored in equeous solution at -10°. Starting with 200 µmoles of (32 P) TMP, the final yield was 120 µmoles and the specific activity of the product was 60 x 10⁶ counts per minute per µmole when measured in a windowless gas-flow detector (Section 2.2.3 (f)).

A sample taken for paper chrowntography (Section 2.2.3 (b)) revealed a single ultraviolet light absorbing spot corresponding to TTP and subsequent scanning of the chromatogram showed that all detectable redicectivity was

- 71 -

associated with the triphosphete region (Section 2.2.3 (d)).

 $({}^{32}P-\beta, \chi)$ TTP was prepared by a similar procedure using unlabelled TMP and $({}^{32}P)$ - labelled orthophosphate as starting materials. In this once the product was obtained in a final yield of 70 µmoles with a specific activity of 18 x 10⁶ counts per minute per µmole when the reaction mixture contained 100 µmoles of TMP and 5 mc of carrier-free $({}^{32}P)$ - orthophosphate.

(c) Preparation of unlabelled thymidine 5'-diphosphate (TDP)

Thymidine 5' -diphosphate was prepared from thymidine 5' -triphosphate by partial aoid hydrolysis as follows: - 200 µmoles of TTP (trisodium salt) and 0.8 ml. of 5 M hydrochloric acid in a total volume of 20 ml. were incubated at 50° for 45 minutes with continuous shaking. The reaction mixture was then cooled rapidly and the pH adjusted to 7.5 with 10 H sodium hydroxide. Samples of this solution were submitted to chrometography on paper in the ammonium isobutyrate system (Section 8.2.3 (b)) to obtain a measure of the extent of hydrolysis and the bulk of the material was diluted with distilled water to 1000 ml., i.e., less than 0.01 H with respect to chloride ions. The neutralised solution was then slowly applied to a column of Dowex-1-G1" (20 x 2 cm., dia.) to adsorb the nucleotide meterial present. The resin bed was washed repeatedly with distilled water to remove any unadsorbed nucleoside and the weshing was continued until the extinction at 267 mu in the effluent fell to a low level (2267 less than 0.01) The thymidine nucleotides were then eluted by non-linear gradient elution with 2000 ml. of 0.01 N hydrochloric acid in the reservoir. Elution was carried out at the rate of 1 ml. per minute and fractions of 20 ml. were

collected. The progress of the elution was followed spectrophotometrically at 267 ms and the contents of the tubes from the second peak (tubes 75 to 100) were combined for the isolation of thymidine 5'-diphosphete (Fig. 15).

The TDP so produced was contaminated with inorganic ortho- and pyrophosphate and the combined TDP fraction was treated with 15 ml. of a suspension of acid washed charcool (see Section 8.2.6 (a)). Lithium chloride and inorganic ortho- and pyrophosphate were removed in the manner described for the preparation of $(3^{30}P+a)$ TTP above whereupon the TDP was recovered from the charcool by eluting the column with 0.14 <u>H</u> emmonium hydroxide in 70% ethanol. This eluste was concentrated to a small volume in vecuo at 25° and the concentrate passed through a column of Dower-50-Mm⁺ to convert the TDP to the acdium selt.

Paper chrometography of the product (Sections 2.2.3 (b) and (c)) revealed a single ultraviolet light absorbing region corresponding to TDP and the overall yield was 40 µmoles or 20%.

(d) <u>Preparation of thymidine 5'-diphosphate labelled with (³²P) in the</u> <u>proximal position ((³²P-a) TDP) and labelled with (³²P) in the</u> <u>terminal position ((³²P-b) TDP)</u>

 $({}^{32}P-a)$ TDP was propared from $({}^{32}P-a)$ TTP by the procedure described ebove, the product in this case being slightly contaminated with inorganic pyrophosphete. No other nucleotides could be detected in the preparation and with a starting meterial of specific activity of 20 x 10⁶ counts per minute per µmole, that of the product was 18 x 10⁶ counts per minute per





Fig. 15. The separation of reaction products in the preparation of thymidine 5'-diphosphate from thymidine 5'-triphosphate by partial sold hydrolysis.

> The nucleotide mixture was adsorbed on a column of Dower-1-21" (80 x 1 cm., dis.) and elution was effected by applying to the column a non-linear gradient with 2000 ml. of 0.01 <u>H</u> hydrochloric acid in the mixing vessel and 0.2 <u>H</u> lithium chloride in 0.01 <u>N</u> hydrochloric acid. Fractions of 20 ml. were collected at a flow rate of 1.5 ml. per minute. The elution was discontinued after the two peaks containing TMP and TDP, in that order, had been collected. Fractions 72 - 97, representing the second peak, were combined and found to contain 30.3 µmoles of TDP, the over-all yield being 16 per cent.

umole. The overall yield of (32P-a) TDP was 181.

 $(^{33}P-\beta)$ TDP was prepared from $(^{33}P-\beta, \chi)$ TTP by partial hydrolysis with dilute hydrochloric acid as already described. A sample of the product taken for chromatography in the amonium isobutyrate system (Section 2.8.3 (b)) revealed one major ultraviolet light absorbing region correspondin to TDP: however, subsequent analysis (Sections 2.8.3 (c) and (d)) showed that about 25% of the total radioactivity was present in the form of (^{32}P) orthophosphete presumably due to partial degradation of the TDP in the course of isolation and during storage. With a $(^{32}P-\beta, \chi)$ TTP preparation of specific activity 18 x 10⁶ counts per minute per µmole as starting material, the $(^{32}P-\beta)$ TDP produced had a specific activity of 10 x 10⁶ counts per minute per µmole when determined in a windowless gas-flow counter. The overall yield was 20%.

On some occessions, when only small amounts of starting material ware evaluable, a simplified isolation procedure was adopted in the preparation of $\binom{32}{2}$ -)-labelled thymidine diphosphates. This procedure entailed separation of 1 - 8 µmoles of total mucleotide material obtained by the usual partial hydrolysis of the appropriate labelled TTP. The neutralised hydrolysis products were applied as streaks on Wintman SHI paper chrometograms which were subsequently developed in the ammonium isobutyrate system (Section 8.8.3 (b)). It was found that while TDP could be obtained free from other nucleotide contaminants in this way, the products invariably contained a relatively high proportion of redicective inorganic pyrophophete and the TDP so prepared could only be used as substrate for the thymidylate kinness in preliminary or qualitative experiments.

2.2.3 General applytical methods

(a) Retimetion of protein

The estimation of protein was in all cases carried out by the method of Lowry, Rosebrough, Farr and Randall (1951) using bovine serum elbumin (Armour Pharmaceutical Company Ltd., Eastbourne, England) as standard. The colour intensity was determined using a Unicam SP-600 spectrophotometer at 750 mi.

(b) Chrometography on paper

For separation and identification of decxyribonucleotides and nucleotides obtained as reaction products in enzyme conversions the procedure described by Keir and Smellie (1959) was used with certain modifications. A similar method of paper chromatography was employed at various stages in the chemical preparations of thymidine nucleotide substrates.

Deproteinised neutral solutions containing nucleotide material were applied to Whatman No. 1 chromatographic paper as small spots using as markers the appropriate nucleosides or nucleotides. The solvent system used was a single phase containing: 100 ml. isobutyric acid, 4.8 ml. ammonium hydroxide (specific gravity 0.880), 1.6 ml. of 0.1 <u>M</u> ethylenediaminetetraacetate (EDTA) and 55.8 ml. distilled water. This solvent was first described by Krebs and Hems (1953) but was modified by Keir and Smellie (1959) to give a pH of 4.6. The paper was developed as a descending chromatogram at 23° for times varying from 18 to 30 hours depending on the composition of the mixture to be separated. It was found that while 18 hours sufficed

- 75 -

to bring about separation of decayribonucleosides from decayribonucleotides, irrigation times of up to 30 hours were necessary to achieve reproducible resolution of decayribonucleoside di- and triphosphates. After drying the paper in air at room temperature, the spots were located by virtue of the quenching of fluorescence by purine and pyrimidine derivatives unler a lowpressume mercury lamp emitting at 2537 Å. This procedure provided satisfactory resolution of mixtures where the total concentration of nucleosides and nucleotides did not fall below 0.05 µmoles or exceed 2 µmoles. Table 1 gives R_g values of all relevant nucleosides and nucleotides as determined using the above procedure.

(c) Location of free or bound phosphate on paper chromatograms

It was frequently desirable to be able to demonstrate the production or presence of inorganic ortho- or pyrophosphete or ester-linked phosphete in mixtures submitted to paper chromatography. In such cases the following procedure was adopted: the dry, developed chromatogram was dipped in a solution prepared by dissolving 1 gs. of emmonium molybdate in 8 ml. of distilled water to which was added 3 ml. 70% perchloric acid, 3 ml. concentrated hydrochloric acid and 86 ml. of acetons. The paper was then allowed to dry in a stream of air after which the areas representing inorganic phosphates were visible as light green coloured spote. Exposure of paper so treated to ultreviolet light from a low-pressure mercury lamp emitting at 2557 Å for 30 minutes revealed all phosphorus containing sections as blue spots on a white background. It was thus possible to distinguish between spots containing inorganic phosphates and those containing seter-

- 76 -

Table 1.

Substance	Rf		
AMP	0. 52		
ADP	0.40		
ATP	0. 32		
CdR	0.67		
dCMP	0. 53 .		
dCDP	0.40		
dCTP	0.30		
UdR	0.45		
dUMP	0.33,		
dUTP	0.18		
TdR	0. 63		
TMP -5'	0. 44		
TMP-3'	0. 49		
TDP	0. 30		
TTP	0.26		
inorganic phosphate	0. 40		

Table 4. R_f values of nucleosides and nucleotides as determined by descending chromatography with isobutyric acid: concentrated annonium hydroxide (specific gravity 0.880) : 0.1 <u>M</u> IDTA: distilled water (100:4.2:1.6:55.6) as solvent. Irrigation was for 18 hours at 23⁰. linked phosphate. The colours produced by this reaction were somewhat transient but could be made to reappear by exposing the paper to emmonie.

(d) Location of redicectivity of paper chrometograms

Mixtures of radioactive nucleosides and nucleotides were separated on paper chrometograms as described above (Section 2.2.3 (b)). Nonradionative marker spots of AMP, ADP, ATP and the appropriate deoxyribonucleosides and deoxyribonucleotides were also applied. After irrigation and drying, the channel bearing the radiosctive mixture was out from the paper as a strip, 1.5 inches wide, and attached to a strip of unused chrometographic paper of the same width. The paper strip was then scanned for redicactivity using a Nuclear-Chicago 2 pi Actigraph system with a windowless counting chamber. In order to relate the chrometogrem strip to the chart from the recorder, marker spots of material containing the appropriate radioactive isotope were applied to the chrometogram strip at a point 1 cm. beyond the origin. Areas bearing radioactivity were identified by comparison with non-radionative marker spots on the original chrometogram. At maximum sensitivity, the Actigraph system could detect approximately 1500 counts per minute (or 0.002 µc) of ("H) in a spot 1 inch in dismeter. the corresponding value for (32P) being 200 counts per minute (or 0.0005 µc).

(e) <u>Spectrophotometric</u> messurements

For general spectrophotometric measurements in the ultraviolet region, it was found convenient to use either the Unicam Model SP-500 or the Beckman Model DU spectrophotometers. In general, measurements at pH 1 - 2 gave

- 77 -

the most consistently reproducible results and, unless otherwise stated, this condition applies throughout the present investigation. In some enzyme assays where relatively large amounts of unlabelled deoxyribonucleotides were included in the reaction mixtures, the identity of the products was checked by determination of the absorption spectra with the use of the Cary Recording Cuartz Spectrophotometer Model 11. The adsorption spectre of TMP, dOMP, dUMP and AMP as determined with the use of the above instrument are shown in Fig. 16. The spectral constants; molar extinction coefficient (E max). A max A min and the ratio \$280/\$260 were obtained directly or calculated from the ultraviolet absorption spectra of the appropriate nucleotide. Table 2 shows the values for E _____ obtained experimentally in comparison with those reported in the literature for THP (For and Shugar, 1952; MacNutt, 1952; Beaven, Holiday and Johnson, 1955), dGMP (MacNutt, 1952; Boch, Ling, Morell and Lipton, 1956, and dUMP (Fox and Shugar, 1952). Since the moler extinction coefficient is a fundamental property of the pyrimidine deckyribose moisty this value was assumed to be a constant for all members of the same homologous series, e.g., thymidine, TMP, TDP and TTP.

(f) Techniques used for determination of radioactivity

The radioactive isotopes employed in the present investigation were (^{32}P) and (^{3}H) and the emitted pulses were recorded with the use of a Nuclear-Chicago Nodel G-lloB windowless gas-flow counter. When substrates bearing (^{32}P) were involved it was sometimes convenient to modify the counting chamber by the introduction of a thin end window. The efficiency of the detector system with (^{32}P) was near to 50% and this efficiency was only

- 78 -



Fig. 16. Ultraviolet spectra of nucleoside 5'-monophosphates at pH 2.

> The spectra were determined using a Cary Medel 11 recording spectrophotometer and refer to solutions of TMP (TMP-Ga · 3 H₂O), dGMP, dUMP (dUMP-Na₂ · 3 H₂O) and AMP (AMP · 3gO) of the concentrations 0.991 x 10⁻⁴ M, G.925 x 10⁻⁴ M, 1.032 x 10⁻⁴ M and 0.662 x 10⁻⁴ M, respectively.

THP		_	-
AGIP	13 <u>1</u> 3		
	1.6.3		
aute			• •

AMP

AND	ADIND	4000 as	THE	Sube tance
258	262	278	267 mu	×
229	229	239	234 mu	A nin
14.95	10,00	12.85	9.63	€ <u>m</u> x 10 ³
0.224	0.36	2.05	0.715	E280 E260
15.1	10.0	13.0	9.6	€ x 10 ⁻³ from the literature

Table N

Table 2. Spectral constants of nucleotides applicable in the ultraviolet region and st pH 2.

The constants were calculated or derived from the spectra recorded in Fig. 16 and the molar extinction coefficients (\in_{max}) so obtained were compared with those reported in the literature (Fox and Shugar, 1951; Beaven, Holiday and Johnson, 1955). fractionally reduced with the introduction of the end window.

The samples (0.1 ml.) to be counted were plated on circular stainless steel planchettes and the moisture allowed to evaporate under an infra-red Such samples frequently contained relatively high concentrations Lamp. of hydrochloric acid and aluminium planchettes could therefore not be used; steinless steel planchettes appeared to be resistant to concentrations of hydrochloric acid up to 0. 5 N. It was found that the efficiency of the counter in detecting the powerful B-radiation from (32P) samples remained unimpaired by variations in the plating technique and it was therefore not necessary to ensure an even spread of the sample. Tritiun-containing meterials, being weak \$-emitters, are especially susceptible to selfabsorption and corrections must normally be introduced in order to obtein the true counting rate. However, if the sample can be regarded as being of infinite thickness or of infinites thinness such corrections may be disregarded. In the present experiments care was taken to spread the sample evenly over the surface of the planchette with the aid of 5 - 10 drops of absolute alcohol and evaporating the fluid slowly under an infra-red lamp. All samples to be counted were protein-free and invariably contained only minute amounts of solid material; films formed by the above procedure were invisible approaching infinite thinness. The validity of the technique was checked by counting progressive dilutions from the same sample and showing that the number of counts per minute was inversely proportional to the dilution. The efficiency of the windowless gas-flow detector for (3H) was 30% .

The background redionctivity was 20 - 25 counts per minute as determined with a windowless counting chamber while in the end window system used for (³²P) only, the background redionctivity was reduced to 14 - 17 counts per minute. Such background redionctivity was estimated by plating 0.1 ml. of distilled water and 5 drops of ethanol on steinless steel planchettes and introducing the resulting dried planchette into the appropriate counting chamber.

2.2.4. Procedures used for the assay of kinase activities

(a) Assay of kinases ontelysing the formation of thymidine 5'-triphosphate from thymidine

The thymidine and thymidylets kineses were asseyed by a modification of the method described by Weissman, Smallie and Paul (1960). Unless otherwise indicated, the reaction mixtures contained the following components in a total volume of 3.0 ml.: 15 µmoles of MgGl₂, 15 µmoles adenosine 5'-triphosphate, 300 µmoles of tris buffer, pH 7.9, 0.3 µmoles of 2merceptoethenol and either 3 µc (³H) thymidine of specific activity 360 mc per mmole or (³²P) thymidine 5'-monophosphate, (³²P-a) thymidine 5'-diphosphate or (³²P- β) thymidine 5'-diphosphate in the quantities indicated in the legends to the figures and tables. Extracts from Landschutz assites tumour cells (protein concentration 3 - 6 mg. per ml.) or extracts from ret liver (protein concentration 5 - 7 mg. per ml.) were occasionally used directly as the source of enzymes in which once a volume representing 4.5 mg. of protein was included in the reaction mixture described above. Mormally, the enzyme solutions were prepared from a stock of lyophylised cell-free extract preparations obtained as described cerlier (Section 2.2.1.). Solutions containing 6 - 8 mg. of protein per ml. were prepared by extracting 40 mg. of the lyophylised enzyme preparation with 3 ml. of distilled water and removing the undiscolved material by centrifugation at 2000 x g for 80 minutes at 0° . Portions of the enzyme solution so obtained, containing 4.5 mg. of protein, were incubated with the above remotion mixture for varying times at 37° in a shaking water-bath.

The reaction was terminated by adding 1.5 ml. of ice-cold 2.1 \underline{I} perchloric acid and all subsequent manipulations were carried out in the cold to prevent non-enzymic hydrolysis of the reaction products. The protein precipitate was removed by centrifugation, washed three times with 0.5 ml. of 0.5 \underline{M} perchloric acid and the supermatant fluid and the washings were combined. The solution was adjusted to pH 7.8 with 7 \underline{H} potassium hydroxide, allowed to stand for 6 hours in the cold and then centrifuged to remove the precipitate of potassium perchlorate. The supermatant fluid was decented and diluted to 7.0 ml. with distilled water and after samples had been reserved for measurement of redicactivity and for paper chrometography (Section 2.8.3 (b)), pert of the remainder was applied to columns of ECTEOLA cellulose, prepared as described elsewhere (Section 2.2.6 (d)), for the separation of thymidine derivatives.

The procedure adopted was as follows: Portions of the supernatant fluids from the reaction mixtures (3.5 ml.) were run on to columns of NOTHOLA cellulose (10 x 1.5 cm., dis.) and the columns were eluted

- 61 -

successively with 100 ml. of distilled water, 50 ml. of 0.01 <u>M</u> hydrochloric acid, 50 ml. of 0.08 <u>M</u> hydrochloric acid, 100 ml. of 0.04 <u>M</u> hydrochloric acid and 70 ml. of 0.1 <u>M</u> hydrochloric acid. Thymidine was removed in the water fraction, TMP in the 0.02 <u>M</u> HGL, TDP in the 0.04 <u>M</u> HGL and TTP in the 0.1 <u>M</u> HGL fraction. The effluent was collected in fractions of 10 ml. with a maximum flow rate of 0.5 ml. per minute and 0.1 ml. portions from each fraction were plated on stainless steel planchettes. The time for 400 disintegrations was determined in a Nuclear-Chicago windowless gas-flow counter (see Section 8.2.3 (f)) and the total counts per minute in each portion of the eluste were calculated as a percentage of the total counts per minute, recovered from the column. The pattern of elution of thymidine, TMP, TDP and TTP is shown in Fig. 17.

The assay procedure invariably resulted in losses of radioectivity perticularly when (³H) TdR was used as substrate. In such cases it was found that 25 - 30% of the total counts per minute present initially were lost in course of the manipulations leading to the neutralised supermatant fluid and that a further 10 - 15% of the total redicectivity present prior to adsorption on ECTEDIA cellulose could not be recovered from the columns. Direct losses to the environment by randomisation and by radiochemical decomposition have been implicated as a major source of removal of tritium label in biological systems (Evens and Stanford, 1963a, b; Goutier, 1964). In addition to effects of this mature, it became evident that serious losses of radioectivity occurred in conjunction with the precipitation and removal of Kolo, from the reaction medis. Naming of the precipitate with distilled

- 82 -



Fig. 17. Chrometographic separation of (³H) -labelled thymidine, TMP, TDP and TTP on columns of ECTEDIA cellulose (lo x 1.5 cm., dia.) by stepwise elution with increasing concentrations of hydrochloric acid. water was impracticable because of the relatively high solubility of $EGlO_4$ in water (0.75 gm. 1 at 0°) and subsequent interference from high solt concentrations with chromatography on EGTEOLA collulose. When substrates bearing a (³²P) label were employed, the recovery of radioactivity was consistently high, 90 - 951 of the total counts per minute present initially being recovered prior to adsorption on EGTEOLA cellulose while the loss of radioactivity subsequently incurred by adsorption and elution from such columns never exceeded 5 - 10%. Galculations and results as presented in Section 8.3. were based on the assumption that losses of radioactivity derived in equal proportion from each of the radioactive reaction products.

In spite of the reservations listed above and the relatively long time required for the completion of a single essay (5 - 7 days), the present procedure has the advantage of giving a high degree of resolution, particularly in the clear separation of TDP and TTP, and it also provides a means for the precise measurement of all the four reaction products.

(b) Determination of specific activities

In some experiments with (³²P)-lebelled substrates, the specific activities of the TMP, TDP and TTP present in the reaction mixtures were determined. Because of the minute absolute amounts of these compounds normally present in reaction mixtures, a large excess of TMP, TDP and TTP was added in certain experiments to provide sufficient nucleotide material for accurate spectrophotometric measurement of the amount present. Direct spectrophotometric measurement of the amount present. Direct in the fractions recovered from the SCTEOLA cellulose columns were imprediced

- 83 -

on account of the large quantities of adenosine nucleotides present in the reaction products.

Portions of the elustes from the columns of MCTEDIA cellulose corresponding to THP, TDP and TTP were separately combined in 150 ml. flasks and 5.0 N hydrochloric acid was added in the quantity required to make the solution 0.2 M. Flacks containing TMP were heated at 100° for 20 minutes, those containing TDP or TTP were heated at 100° for 60 minutes after which time the mixtures were cooled under the tap and the pH was adjusted to 7.5 with 10 H sodium hydroxide. A series of preliminary experiments had shown that under these conditions of controlled acid hydrolysis, THP remained intest while TDP and TTP were quantitatively degraded to thymidine monophosphate and adenosine phosphates were hydrolysed to give a mixture of AMP and edenine. Each of the fractions were then diluted to a finite volume and a portion, usually one-half of the total volume, was diluted with distilled water to give a sodium chloride concentration of less than 0.01 M. The solutions so obtained were applied to columns of Dowex-1-G1" (12 x 2 cm., die.) for subsequent separation of TMP from AMP and ademine (Gray et al., 1960). The purine derivatives were removed from the columns by repetted washing with distilled water followed by 50 ml. portions of 0.01 N hydrochloric acid until the extinction of the affluent at 257 mu approached zero. TMP was finally recovered from the columns by eluting with 10 ml. portions of 0.05 N hydrochloric acid, the elustes were each diluted to 50 ml. and the nucleotide concentrations were determined spectrophotometrically at 867 mu. Complete resolution of the thymidine nucleotides from adenosine

nucleotides was achieved by this procedure and the recovery of THP was quantitative.

2.2.5. Buffer solutions

(a) Tris buffer

Tris buffer was propared by adjusting the pH of a 1.0 M solution of 2-amino-2(hydroxymethyl)-propane-1,3-diol to the desired value with concentrated hydrochloric acid. Tris buffers of lower ionic strengths were prepared by dilution of the appropriate concentrated buffer with distilled water.

(b) Phosphate buffer

Phosphate buffers were prepared by adding 0.1 \underline{M} KH₂PO₄ to 0.1 \underline{M} Na₂HPO₄ . 2 H₂O or by adding 0.01 \underline{M} KH₂PO₄ to 0.01 \underline{M} Ma₂HPO₄ . 2 H₂O until the desired pH was attained.

2.2.6. Preparation of adsorbents for column chromatography

(a) Preparation of charcoal adsorbant

Animal charcoal was obtained commercially (British Charcoals and McDonalds Ltd., Grangemouth) and purified as follows: The crude 20 - 60 grist charcoal was ground and sieved to 40 - 60 mesh, suspended in two volumes of 5 M hydrochloric acid and allowed to stand overnight. The fluid was decanted and the charcoal was resuspended in distilled water and washed several times until the supermetent had attained a pH of 7.

Geloured and ultraviolet light absorbing substances were then removed by washing the charcoal liberally in absolute alcohol. The most convenient procedure entailed washing the material embedded in a large Buchner funnel fitted with a Westman No. 1 filter paper in one litre portions with resuspension of the charcoal in ethenol between each filtration. This treatment, which required large volumes of ethenol, was continued until all colouration was absent from the filtrate and the ultraviolet light absorption approached zero.

The charconl was then resuspended in distilled water and washed several times to remove the ethanol whereupon the adsorbant was packed into a column (5 cm., dis.). Further treatment was by washing with 10 - 18 column volumes of <u>H</u> hydrochloric acid followed by distilled water until the effluent attained pH 7. Several column volumes of 0.01 <u>H</u> sodium bicerboante were then applied to the column bed and washing was continued with this solvent until no ultraviolet light absorbing material could be detected in the effluent. After washing with distilled water to pH 7, the nucleotide solvent (0.14 <u>H</u> ammonium hydroxide in 70% ethanol.) was applied to the aharcoal bed until the extinction at 320, 260 and 290 mu of the effluent expressible sero. The solvent was removed by a final washing with distilled water, the column was unpacked and the charcoal suspended and stored in 0.1 <u>H</u> hydrochloric acid.

The charcoal was regenerated after use by further washing with the ethanol-ammonia solvent until all the residual adsorbed meterial had been

- 86 -

removed. Further treatment was by washing with distilled water to pH 7, treatment with 10 column volumes of <u>N</u> hydrochloric sold and the charcoal was finally washed with distilled water until the pH of the effluent returned to neutrality.

(b) Pretreetment of Dower-1-G1 for column chrometography

Dowex AG1 - XS (chloride form) of 100 - 200 mesh was suspended in M hydrochloric sold for 18 hours and the resin so sativated packed into a column fitted with a sinter glass filter. The resin bed was washed liberally with distilled water until the effluent was free from chloride ions. A small plug of glass wool was finally inserted 2 cm. above the surface of the resin bed to prevent any disturbance of the surface in the course of elution. The resin could be regenerated after use by further washing with 0.5 M lithium chloride in 0.1 M hydrochloric acid until all residual adsorbed meterial had been removed followed by washing with water until the pH of the effluent returned to 7.

(s) Preparation of columns of Dower-50-Na

Dower AG50 W - X8 (hydrogen form) of 100 - 200 mesh was suspended in <u>N</u> sodium hydroxide for 18 hours and the resin so transformed was packed into a column. Continuous washing of the resin bed with distilled water was performed until the effluent had attained pH 7. After use, the resin could be regenerated by treatment with three column volumes of <u>N</u> hydrochloric acid followed by washing with distilled water to pH 7, and reconversion to the He form by the application of 10 column volumes of N sodium hydroxide followed by washing with distilled water until the effluent returned to neutrality.

(d) Preparation of ECTEOLA cellulose for column chrometography

EGTEOLA cellulose (Kodek Ltd., London or Serve Entwiklungslebor, Heidelberg, West Germany) was prepared for use by preliminary steeping in twice its own volume of distilled water followed by vigorous agitation with a Vibro-mix Model Bl (AG for Chemie-Apparatebau, Manadorf, Lurich. Switzerland) for periods of 30 minutes. Between each treatment the adsorbent was allowed to settle and the fluid was decented to remove the smaller particles. This procedure was repeated until the supermatant fluid was water-clear (6 - 8 treatments). The suspension of SCTBOLA cellulose was then trensferred to columns and the adsorbant bed washed twice with 50 ml. portions of distilled water followed by 50 mll portions of 0.5 H hydrophloric acid until the effluent was free from material absorbing ultraviolet light in the region 257 - 267 mu. The columns were finally washed with distilled water until the pH of the effluent returned to 7. To ensure that the surface of the adsorbent bed remained undisturbed a plug of glass wool was inserted 2 cm. above the surface and the solvent was never allowed to drain below this mark. Regeneration of the EGTHOLA cellulose was carried out by washing the adsorbent bed with four 50 ml. portions of 0.5 N hydrochloric acid or until the eluste was free from redicactive material followed by washing with distilled water until the effluent reached pH 7. It was observed that the adsorbent was slowly oxidised when used over long periods and the column contents were discarded after three adsorption - regeneration cycles.

- 88 -
2.2.7. Materials

TdR, UdR, GdR and ATP were purchased from The Sigme Chemical Gorporation, TMP, dGMP and dUMP were purchased from Johwars Bioresserch Inc. and from Salifornia Corporation for Biochemical Resserch and TTP and dGTP were obtained from Pabet Inboratories. Additional samples of ATP were purchased from British Drug Houses Ltd. (³H) TdR of specific activity 360 no per mmole and (³H) GdR of specific activity 1100 no per mmole were purchased from Schwarz Bioresserch Inc. (³²P)-ayanoethylphosphete of specific activity not less than 50 nd per mmole was either purchased from The Radiochemical Centre, Ameraham or prepared by the method of Tener (1961) using carrier-free (³⁸P) from the same source. DOC was purchased from The Aldrich Chemical Gompany Inc., Milwakee, and 2-ayanoethenol was obtained from British Drug Houses Ltd.

2.3.1. Optimal conditions for the formation in vitro of TTP from thymidine by cell-free extracts of Landschutz ascites tumour cells.

Several preliminary experiments were performed in order to determine which were the most favourable conditions for the phosphorylation of thymidine to thymidine 5' -triphosphate under the influence of enzymes from cell-free extracts of Landschutz ascites tumour cells. Such experiments have inveriably shown that the activities of the thymidine and thymidylate kinases were particularly low and that the over-all conversion of TdR to TTP by the crude extracts was an inefficient process. One important consequence of this observation was that at no stage of the present investigation did it become feasible to study the formation of TTP in vitro with the use of unlabelled substrates. Thus refuge had to be taken in the use of redicactive substrates, variously labelled with (3H) and (32P), and thereby restricting the permissible interpretation of experimental results. The loss of an lytical accuracy was particularly serious in the onse of substrates labelled with (3H) and the procedures based on the use of this isotope have been regarded throughout the present investigation as little more than a semi-quantitative measure of the reactions under study. The powerful β -rediction emitted by (³⁸P), on the other hand, ellows the easy detection and precise measurement of meterial containing this isotope, and interpretations based on data from experiments with (32p)-labelled substrates have been regarded with confidence as the closest approach to

quantitative evaluation obtainable.

In the assembly of a reaction mixture for the study of the formation of TTP in vitro from its nucleoside precursor, due regard was peid to the known requirements for phosphokinase activity such as a phosphoryl group donor, a divalent cation and suitable stabilising and activating factors. Moreover, it was realised that little reliable information could be derived from the ensuing enzyme experiments in the absence of a suitable method for determination and detection of reaction products. In the present investigation, the manipulation of the reaction products was greatly facilitated by having available a method for the complete separation of all the presumed products of the reaction (TdR, TMP, TDP and TTP) by chromatography on columns of ECTROLA cellulose (see Fig. 17) in what was essentially a single operation. The failure of complete resolution of TDP and TTP has been a major objection to many previous essay procedures based on paper chromatography (Bollum and Potter, 1959; Bianchi, Butler, Grathorn and Shooter, 1961). The recent development of rapid assays as reported by Ives, Morse and Potter (1963) and by Furlong (1963) employing DEAE cellulose strips for the separation of thymine nucleotides likewise suffers from insbility to resolve the higher phospha tes.

Experiments performed to ascertain the most favourable pH for the transformation of $({}^{3}H)$ TdR to TTP showed that the reaction exhibited a relatively marrow pH -optimum giving maximal activity over the range pH 7.8 to 8.1 when the reaction was studied in a medium 5 x 10⁻³ M with respect to MgGlo and ATP. The proportion of the total redicactivity appearing in TTP

- 91 -

declined sharply below pH 7.7 and above pH 8.2 so that the percentage of the total radiosotivity recovered in TTP when the resction was performed at pH 7.5 was approximately one-third of the value obtained at pH 7.9. In such experiments (as shown in Fig. 18), only the proportion of the total rediosotivity present in fractions containing TdR or TTP was recorded, no account being taken at this stage of the radioscivity present in possible intermediates. The assay could, therefore, be expected to reflect the efficiency of the thymidine and thymidylate kinase system of crude extracts of Landschutz escites cells in catelysing the conversion of (³H) TdH to (³H) TTP. Tris was found to be a suitable buffer in which to conduct such experiments and at a concentration of 0.1 M provided a rigid and accourate control of the pH in the range 7 - 9.

Experiments in which the conversion of $({}^{3}\text{H})$ TdR to $({}^{3}\text{H})$ TTP was studied as a function of the Mg⁺⁺ concentration at a constant ATP concentration of 5 x 10⁻³ M, showed that the proportion of the total redicactivity recovered in TTP after a 60 minute incubation period was maximal in the range 5 x 10⁻³ to 15 x 10⁻³ M of MgGl₂ while no redicactive TTP could be detected in the absence of Mg⁺⁺ (see Fig. 19). Subsequent experiments designed to study the effect of variable concentrations of ATP as phosphoryl group donor conducted in the presence of 5 x 10⁻³ M MgGl₂, similarly revealed that the proportion of (${}^{3}\text{H}$) TdR phosphorylated to TTP over a 60 minute period was at its maximum when the reaction medium was from 5 x 10⁻³ to 15 x 10⁻³ M with respect to ATP (see Fig. 26). Such experiments also showed that radioactive TTP could not be detected among the products when ATP was omitted from the

- 92 -

- Fig. 18. The effect of pH on the formation of TTP from (³H) TdR by enzymes from cell-free extracts of Landschutz escites tumour cells.
 - The following reaction mixtures were prepared, each in a total volume of 3.0 ml.: 0.3 µmoles 2-merceptoethanol, 15 µmoles ATP, 15 µmoles MgOlg, 3 µc (³H) TdR (specific setivity 360 µe per unole; 100 µe per ml.), enzyme solution containing 4.1 mg. protein and 300 µmoles of tris buffer of the appropriate pH. Incubation was for 60 minutes at 37°. Of a total volume of deproteinised, neutralised ageny solution of 7.0 ml., 5.0 ml. were adsorbed on a column of ECTEDIA cellulose (10 x 1.5 cm., dia.) and eluted as described in the text (Section 2.2.4 (e)). The propertion of the total radioactivity present in each of the products, TAR, THP, TDP and TTP was determined. However, on the diagram is shown only the proportion of the total radioactivity being recovered in the TTP fraction. The source of enzymes was the crude cell-free extract from Landschuts asgites cardinoma.



umoles MgCl2 per ml.

Fig. 19. Effect of Mg⁺⁺ concentration on the formation of TTP from (³H) TdR by enzymes from cell-free extracts of Landschutz maxitum tumour cells.

> The reaction mixtures contained the components given in the legend to Fig. 13 in a total volume of 3.0 ml. except that they contained MgGl₂ concentrations verying from zero to 40 µmoles per ml., 300 µmoles trie buffer, pH 7.9, and 3.7 mg. protein from a cell-free extract of Landschutz macites carcinoms. Incubation was for 60 minutes at 37° and further treatment was as described in Fig. 18.



µmoles ATP per ml.

Fig. 20. Effect of ATP concentration on the formation of TTP from (³H) TdR by enzymes from cell-free extracts of Landschutz ascites tumour cells.

> Remotion mixtures contained ATP in concentrations varying from zero to 30 µmoles per ml. with other components, including 300 µmoles tris buffer, pH 7.9, in a total volume of 3.0 ml. as described for Fig. 18. The remotion mixtures also contained a volume of enzyme solution gorresponding to 4.5 mg. of protein. Inoutation was for 60 minutes at 37° and further treatment was as described for Fig. 18. The source of enzymes was the hyophylised cell-free extracts of Landschuts asultes carbinom.

reaction medium. While the proportion of the total radioactivity recovered in TTP rose sharply in response to an increase in the ATP concentration from sero to 5 x 10-3 M , at which point more than 60 per cent of the total redicectivity was present as TTP, the proportion fell only slightly with ATP concentrations increasing up to, and beyond SO x 10-3 M. The procedures involving (3H) -labelled substrates did not allow precise kinetic interpretation of results so obtained. Nevertheless, the data suggested that the phosphorylation of TdR to TTP with the crude cell-free extract operated at a maximal rate in the presence of ATP and Mg** in the ratio 1 : 1. Such a ratio would be expected to apply if the true co-substrate for the thymidine and thymidylate kineses was an ATP - Mg - helate. The presence in cell-free extracts of the Landschutz ascites cells of phosphononcesterases capeble of degreding thymidine polyphosphetes has been indicated previously (Gray at al., 1960) and since it is now well established that such enzymes show specific requirements for divalent entions, it was to be expected that an increase in the Mg++ concentration of the renotion medium would cause e stimulation of phosphotese activity. An Me" concentration in excess of that required to maintain an ATP - Mg++ chelate would, therefore, lead to a net stimulation of the est-bolic activity relative to the combined kinase activity and a situation such as that shown in Fig. 19 at the higher Mg ** concentrations could thus partly be accounted for by the availability of un-chelated Mg ++ ions to the nucleotide phosphateses. Another consequence of the presence of phosphatases was that all measurements of anabolic reaction rates reflected net (or apparent) kinase activities and not the absolute

- 73 -

potentialities of the thymidine and thymidylate kinese system.

In the majority of investigations of this type it has been customary to sugment the reaction mixture with a system which promotes the regeneration of ATP and thereby ensures maintenance of a constant phosphoryl group donor concentration. On dynamic grounds, it seems clear that the primary effect of introducing a regenerating system of this type would be to give the system a bias towards phosphorylation of a phosphoryl group acceptor. 3-Phosphoglycerate has been favoured as a "regenerating agent" in experiments with nucleoside and nucleotide kinases and Potter (1963) has shown that the presence of this egent enhanced three-fold the thymidize and thymidyle te kinase activity of cell-free extracts from Novikoff hepatoms. In the present investigation, the effect of the ATPase in the crude ascites extracts on the phosphorylation of thymidine nucleotides was minimized by the use of an ATP - Mg chelate in wast excess of the stoichiometric requirements of the thymidine nucleotide substrates. The ratio of ATP - Mg ++ chelate concentration to phosphoryl group acceptor concentration was 2000 : 1 is the onse of (3H) -labelled substrates and 200 : 1 in experiments with (3H) -labelled substrates. Moreover, several preliminary experiments had shown that over 50 per cent of the ATP so added remained intact at the end of incubation periods up to one hour.

A number of other preliminary experiments had indicated that the ability of the crude ascites extracts to produce TTP from TdR was quickly $O-2^{\circ}$ lost over a period of a few hours even when a rigid temperature control was applied and it became necessary, therefore, to supplement the reaction

- 94 -

mixtures with a stabilizing factor in order to minimise possible progressive enzyme inactivation in the course of enzyme experiments at 37° . 2-Herospto thanol, as a well established protector of enzyme sulphydryl groups, was considered a suitable candidate to fill this function and a series of experiments showed that the presence of 2-merosptoethenol (0.1 x 10^{-3} M) caused a slight stimulation of the ability of the extract to convert (³H) TdP to (³H) TTP. Higher concentrations of this compound, however, depressed the proportion of total redicactivity recovered in TTP and serious inhibition of the thymidine and thymidylete kinese system was observed in the presence of 2-merceptoethenol concentrations exceeding 5 x 10^{-3} M, this effect being presumebly due to chelstion of Mg⁺⁺ ions.

However, it was early recognised that the phosphorylation of TdR to TTP was a composite reaction involving the participation of certain intermediate thymidine nucleotides. It became clear also that interpretations regarding the efficiency of the phosphorylation of TdR to TTP based only on the messurement of the formation of TTP from its nucleoside precursor were relatively superficial unless account was also taken of the intervening reaction stops.

The factors which regulate the formation of TTP is vitro may have wider significance in relation to the synthesis of DNA and a series of studies were initiated in order to elucidate the mechanism of the formation of TTP from TdR with particular regard to the sequence in which each intermediate intervened in the reaction.

- 95 -

2.3.8. The machanism of formation of TTP from thymidine by enzymes from cell-free extracts of Landschutz ascites tumour cells.

As a preliminary to any study on the mechanism of the formation of TTP from TaR by grude extracts of the tumour cells, it was necessary to ascertain that the assay procedure used for the determination of the thymidine and thymidylate kineses (Section 2.2.4 (a)) was valid under the conditions indicated in the preceding section. The results shown in Fig. 21 demonstrate that the formation of THP, TDP and TTP from (3H) TdR was a linear function of the protein concentration of the grude extracts of Landschutz ascites cardinos A set of data obtained with enzymes derived from a redissolved lyophylised preparation from such extracts, likewise, showed that the production of reaction products was directly proportional to the ensyme concentration. In the experiment shown (in which incubation was for 60 minutes at 37°), it expensed that the proportion of the total redicectivity recovered in TTP rose at a slightly faster rate than did redicactive TMP and TDP so that at a protei concentration of 1.65 mg. per ml., the TTP contained 60 per cent of the total radioscivity while TDP contained 11 per cent and TMP 19 per cent. The redioactivity remaining associated with TdR. similarly was seen to decrease as an inverse function of the ensyme concentration. While these experiments clearly indicated that at no point within the protein concentration range of sero to 1.65 mg. per ml. did the enzyme concentrations exceed the substrate concentrations, no information could be glenned from such data as to the sequence in which the various products appeared in the reaction.

Studies on the time course of the formation of TTP from ("H) EdR such



Fig. 21. The effect of enzyme concentration on the formation of TMP, TDP and TTP from (³H) TdR with enzymes from cell-free extracts of Landschutz ascites tumour cells.

> The reaction mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 7.9, 0.3 µmoles 8-merceptosthemol, 15 µmoles MgGlg, 15 µmoles ATP, 3 µc (³H) TdR and enzyme solution corresponding to the quantities of protein given on the diagram. Incubation was for 60 minutes at 37° and further treatment was as described in the taxt (Section 2.2.4 (a)) and in the legend to Fig. 18 except that the percentage of total recovered radioactivity present in each of the products, TdR, TMP, TDP and TTP was plotted against the protein concentration of the reaction mixtures. The source of enzyme was a solution prepared from the lyophylised cell-free extracts of Lendschutz endites cercinoms.

TAR • ; THP A ; TDP D ; TTP o.

as those illustrated in Fig. 25 have invariably shown that labelled TMP and TDP appear in the reaction products very shortly after the start of the reaction. The proportion of the total radioectivity in TMP rose very sharply from the start of the incubation and reached a plateau at about 15 minutes which was maintained until about 60 minutes. Thereafter, there was a further alight increase in the total radioectivity in TMP. The radioactivity in TDP also began to rise from time zero and repidly reached a steady low level that never exceeded 50 per cent of the total radioectivity. Labelled TTP also appeared early in the course of the reaction, the proportion rose to a maximum value of the order of 60 per cent of the total radioectivity in 15 to 20 minutes, remained reasonably constant up to 60 minutes, and then declined slightly, presumbly as a result of dephesphorylation of the TTP.

It became eleme from experiments of this type that the crucial events related to the sequential formation of the reaction intermediates occurred early in the reaction and experiments in which the early time intervals were examined in more detail (see Fig. 23) showed that while small amounts of radioactive TMP and TDP were produced within the first minute, no radioactivity could be detected in the TTP until after 3 minutes.

These results were consistent with the view that THP and TDP are intermediates in the formation of TTP from TdR, but, in the absence of more discriminating kinetic information about the relative rates of the different reactions, could not be taken as a definite indication of such a reaction sequence.

- 97 -





Fig. 22.

Fig. 22. The time course of phosphorylation of (³H) TdR by cell-free extracts of Landschutz ascites tumour cells.

> The remotion mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0, 0.3 µmoles 2-mercepteethol, 15 µmoles MgGl₂, 15 µmoles ATP, 3 µc (³H) TdR and enzyme solution containing 4.1 mg. of protein. The remotion mixtures were incutated for the intervals indicated on the diagram in a shaking water-both at 37°. Further trantmost was as doscribed in the text (Section 2.2.4 (a) The source of the enzymes was the crude cell-free extracts of Landschutz assites parcinoma.





Fig. 23. The initial time course of phosphorylation of (³H) TdR by cell-free extracts of Landschutz escites tumour cells.

> The composition of the reaction mixtures was the same as that given in the legend to Fig. 22 except that each reaction mixture contained enzyme solution corresponding to 4.5 mg. protein. Further treatment was as given for Fig. 28 and in the text (Section 2.2.4 (a)). The source of enzymes was the grude cell-free extracts of Landachutz ascites carcinoma. Different enzyme preparations were used for the experiments recorded in Figures 22 and 23.

THP \triangle ; TDP ∇ ; TTP o.

It was apparent that while the initial reaction in the formation in <u>vitro</u> of TTP from TdR was the relatively well established phosphorylation of TdR to TMP (Ives, Morse and Potter, 1963; Belts, 1962; Sköld, 1960b, little information was available on the precise sequence of the subsequent anabolic events and the orucial point at issue would seem to be the part played by TDP in the process. A series of experiments was, therefore, carried out in order to determine whether the oruce enzyme system could catelyse the formation of TTP directly from TDP.

Then (³²P - a) TDP was used as substrate, it was found that TDP was extensively converted to TTP. Thus, more than 60 per cent of the total radioactivity was recovered as TTP at the end of a one hour insubstion (see Fig. 24). Moreover, the addition of a 20-fold excess of unlabelled TMP did not significantly alter the small proportion of radioactivity found in the TMP; subsequent investigation indicated that the radioactivity of the TMP fraction was due to non-ensymic degradation of the TDP during storage and incubction.

The basic assumption in all experiments with (³²P) -labelled substrates was that any (³²P) orthophosphate present or formed did not exchange with co-valently linked phosphate by other than metabolic means. With this assumption in mind, the object of the present experiment was to discover whether the addition of unlabelled TMP led to any dilution in the radioactivity appearing in TTP. Two possible pathways for the transfer of radioactivity from TDP to TTP could be visualised under the present aircumstances, one of which was the direct transformation of TDP to TTP (a). The elternative was a preliminary degradation of TDP to THP followed by pyrophosphorylation of

- 98 -



Fig. 24. The phosphorylation of (³²P - a) TDP by cell-free extracts of Landschutz ascites tumour cells in the presence and absence of added unlabelled TMP.

> The reaction mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0, 0.3 µmoles 2-merceptoethenel, 15 µmoles MgGl₂, 15 µmoles ATP, 24 mµmoles (32 P - a) TDP of specific activity 3.5 x 10⁶ counts per minute per µmole and, where indicated, 500 µµmoles TMP in addition to a volume of enzyme solution corresponding to 4.5 mg. of protein. Incubation was for 60 minutes at 37° and further treatment was as indicated in the text (Section 2.2.4 (a)). The experiment designated "control" represents an incubation in which enzyme solution was omitted from the reaction mixture. The gourge of enzymes was a solution prepared from the lyophylized cell-free extracts of Lanischutz ascites carcinome.

TMP to give TTP (b) thus:



An experiment in which (32p - a) TDP was incubated with enzyme could, therefore, not be expected to differentiate between the two alternative pathways for producing TTP. However, in the presence of an excess of unlabelled TMP, the operation of alternative (b) would inevitably result in a dilution of the redicectively labelled meterial entering the monophosphate pool so that, in subsequent pyrophosphorylation, a relatively smaller proportion of the TTP would be labelled. No dilution in the radiosctivity recovered in TTP could be expected if the reaction proceeded according to alternative (a). Thus, while the experiment shown in Fig. 24 seems to favour the direct formation of TTP from TDP, the application of steady-state tracer kinetics was inadmissible since the criterion of measurement was the proportion of total radioactivity present in TMP, TDP or TTP. Other experiments were, therefore, carried out in which the specific activities of the TTP formed in the absence and the presence of a large excess of unlabelled The were determined. As recorded in Table 3, such data was generally consistent with the earlier conclusion in that only a relatively spall reduction in the specific activity of the TTP resulted from the addition of

THUP	2.5 unoles	Complete	Complete + 1.25 µmoles THP - enzyme	Complete +2.5 µmoles TMP - enzyme	Complete - enzyme		Resction mixture	
68	18	90	94	56	86		Percent- age of total radio- activity recovered	
92	94	•	T 6	3	•		Fraction o total nucl tide conter recovered	
10	9	9	e.6	01	10	THE	#8 **	
27	τĸ	32	91	90	90	TOP	PERCENT total r setivit in:-	
63	8	59	•	•		TTP	AGE of adio- y recovere	
4.21 x 10 ³	2.27 x 10 ³	* x 10	4-3 3	2.24	1	THE	Specific setivity of thymidine nuc tides counts/min./u	
1.4 106	1.6 x 106	2.4 10 x 10	2.2	2.14	-	TOP		
1.7	1.4 × 10	3.6 x 1	1	1	1	TTP	leo- mole	

Deuterenen an ann

Table 3.

Table 3. The formation of thymidine 5'-triphosphete from $\begin{pmatrix} 32 \\ P - a \end{pmatrix}$ TDP in presence of added, unlabelled TMP.

> The complete reaction mixtures contained the following conponents in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0, 0.3 µmoles 2-merosptoethenol, 15 µmoles MgGlo, 15 µmoles ATP, 85 µµmoles $(^{32}P - a)$ TDP of specific activity 2.41 x 10⁶ counte per minute per µmole and enzyme solution containing 4.5 mg. of protein. Some reaction mixtures included 2.5 µmoles TMP or 1.25 µmoles TMP representing an excess of 100 or 50, respectively, vis a vis the absolute quantity of TDP. Incubation was for 60 minutes in a shaking water-bath at 37°. Deproteinised, neutralised reaction mixtures were diluted to 7.0 ml., 5.0 ml. of which were fractionated on columns of EGTROLA cellulose (10 x 1.5 cm., dia.), 0.1 ml. portions of each 10 ml. eluste fraction being taken for counting Eluste fractions containing either TMP, TDP or TTP were hydrolysed and the adenosine nucleotides removed as described in the text (Section 2.8.4 (b)) and the total extinction at 267 mu determined in the recovered reaction products. The ensyme source was the lyophylised cell-free extracts of Landschutz escites cercinome.

100-fold or 50-fold excess of unlabelled TMP to the reaction mixtures. The minor fluctuations in specific activities that were observed could be accounted for by the considerable error involved in measuring the extremely small emounts of TTP present and no differences were observed such as would have been necessary had the TTP been formed from the TDP vis a large pool of TMP. Moreover, some of the observed reduction in the specific activity of the TTP in the presence of unlabelled TMP could be due to phosphorylation of such unlabelled material.

These experiments clearly demonstrated the ability of the enzyme preparations to form TTP from TDP and provided no evidence for the intermediate formation of TMP. Furthermore, they provided no evidence for the formation of any higher phosphorylated dorivatives of thymidine than TTP in contrast to the experience of Bianchi, Butler, Grathorn and Shooter (1961). This was shown by washing the columns of EGTEOLA cellulose repeatedly with 0.5 M hydrochloric acid after all the TTP had been eluted (see Section 8.2.4 (a)).

Further confirmation of these conclusions was furnished by experiments on the time course of phosphorylation of $({}^{38}P - \beta)$ TDP as shown in Fig. 25. It became clear that the TDP was very rapidly and extensively converted to TTP more than 60 per cent of the total radiosctivity being recovered in the TTP within 5 minutes. About 20 per cent of the total radiosctivity at all time intervals was accounted for as inorganic orthophosphate, but this proved to be due to contamination of the TDP preparation (see Section 2.2.2 (d)) and did not very significently during the course of the incubation.



Fig. 25. The time course of phosphorylation of $(32 - \beta)$ TDP by cell-free extracts of Landschutz excites tumour cells.

The reaction mixtures contained the following components in ext total volume of 3.0 ml.: 300 µmoles tris buffer, pH 7.9, 0.3 µmoles 2-merceptoethenol, 15 µmoles MgGl₂, 15 µmoles ATP, 0.172 µmoles ($^{32}P - \beta$) TDP of specific activity 1 x 10⁶ counts per minute per µmole and a volume of enzyme solution corresponding to 4.5 mg. of protein. Incubation was for the time intervals indicate on the diagram in a shaking water-bath at 37°. Further treatment was as described in the text (Section 2.2.4 (a) The source of enzymes was a solution prepared from the lyophylised cell-free extracts of Landschutz macites carcinoms.



TTP _____ -

Inorganic orthophosphate (P) -----O----

Since the TDP used in these experiments was labelled with $\binom{34}{2}$) in the β -phosphate group, any degradation of the TDP, by for example phosphatese action, would have given rise to unlabelled TMP and $\binom{32}{2}$) orthophosphate so that labelled TTP could only have been formed from such TMP if the $\binom{32}{2}$) orthophosphate had been utilised in the phosphorylation of the TMP or had taken part in an enzyme catelysed exchange with the phosphate group of TMP. The two main elternatives for the formation of TTP from TDP in such a system my be indicated as follows:



In order to exclude the possibility that $\binom{32}{p}$ orthophosphate might be involved in exchange reactions with covalently linked phosphate groups under the prevailing experimental conditions, experiments were carried out in which unlabelled TDP together with an emount of carrier-free $\binom{32}{p}$ orthophosphate corresponding to the amount of $\binom{32}{p}$ in the $\binom{32}{p} - \beta$ TDP was substituted for the $\binom{32}{p} - \beta$ TDP. No labelled TTP was formed under these conditions, nor was there any evidence in any of these experiments for the formation, however transient, of higher phosphorylated derivatives of thymidize than TTP.

From these two types of experiment, it became clear that the crude preparations of enzyme contained a kinase for the phosphorylation of TEP to TTP and that this reaction occurred very rapidly without there being any evidence for the intermediate formation of TMP or of higher phosphates than TTP.

Further studies were focussed on the remaining step in the phosphorylation of thymidine to TTP. It seemed likely that if the initial step consisted of phosphorylation of TdR to TMP and the ultimate step was the phosphorylation of TDP to TTP, there should also exist an enzyme in the grude enzyme preparations gapable of gatelysing the formation of TDP, with TMP as its substrate.

Studies on the time course of phosphorylation of (³²P) THP were, consequently, carried out and the results of such an experiment is shown in Fig. 26. It is clear from this that at very early time intervals labelled TDP appears among the reaction products before there is any evidence of labelled TTP, thus the proportion of total redicactivity in TDP increased from time zero while no labelled TTP could be detected until after 5 minutes. thereafter labelled TDP reached a plateau at about 15 per cent of the total recovered radioactivity. That the radioactivity in the TDP never rose to a high level may be due to the rapid rate at which TDP is converted to TTP as shown in Fig. 25. The general pattern of phosphorylation of ("P) THP was consistent with the picture revealed by the experiments recorded in Figures 22 and 23 where the initial substrate was (3H) TdR. While the interpretation of these experiments were subject to the reservations stated above (p.99), the data seems to suggest the possibility that the rate of formation of TDP may be rate limiting in TTP synthesis.

Fig. 26. The time course of phosphorylation of (³²P) TMP by cellfree extracts of Landachuts socites tumour cells.

> The reaction mixtures contained the following components in an total volume of 1.0 ml.: 100 µmoles tris buffer, pH 7.9, 0.1 µmoles 2-merceptoethenol, 5 µmoles MgCl₂, 5 µmoles ATP, 5.5 mumoles (³²) TMP of specific activity 86 x 10⁶ counts per minute per µmole and enzyme solution containing 1.5 mg. of protein. Incubation was for the times indicated on the diagram in a shaking water-bath at 37° . Further treatment was as described in the text (Section 2.8.4 (a)). The source of enzymes was a solution prepared from Lyophylised cell-free extracts of Landschutz escites carcinoms.

THP A ; TOP 7 ; TTP 0.

In subsequent experiments, attempts were made to measure the specific activities of the TMP. TDP and TTP at various time intervals during the conversion of (32p) TMP to TTP. In these experiments, large excesses of unlabelled TDP and TTP were added to the reaction mixtures in order to provide sufficient quantities of TDP and TTP in the reaction products for accurate spectrophotometric measurements. The results of such an experiment, in which the reaction mixtures included a 10-fold excess of TDP and TTP relative to the absolute TAP concentration, greshown in Fig. 27. The high total thymidine nucleotide concentration thus contained in the reaction mixture, however, resulted in serious inhibition of the reaction so that only about 2 per cent of the (32p) THP was phosphorylated within time intervals up to one hour. Nevertheless, it was shown that the specific activity in the TDP rose to an early maximum value before any significant increase in the specific activity of the TTP was observed, this observation being therefore consistent with the reaction sequence THP = TDP = TTP. The significance of these data was, however, minimized by two important factors. Firstly, the fact that the extent of the total reaction was so anall made precise measurements of reaction products difficult and, secondly, it seemed clear that the ideal conditions as visualised by Allversmit. Entenman and Fishler (1943) for steady-state preductsor - product relationships did not obtain in the isolated reaction system studied here. In the present case, it seems reasonable to suggest that the ensyme system was particularly sensitive to accumulation of reaction products and that the efficiency of the process of TTP synthesis thus depends, in part, on the efficient



Fig. 27. The time course of phosphorylation of (³²P) THP in presence of added, unlebelled TDP and TTP by cell-free extracts of Landschutz ascites tumour cells.

> The reaction mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0, 0.3 µmoles 2-merceptoethenol, 15 µmoles MgGl., 15 µmoles ATP, 50 munoles (32 P) TMP of specific activity 1.5 x 10⁶ counts per minute per µmole, 0.75 µmoles TDP, 0.75 uncles TTP and enzyme solution containing 4.5 mg. of protein. Incubation was for the time intervals indicated on the disgram in a shaking water-bath at 37 . Neutralised, deproteinised assay solutions were diluted to 7.0 ml., 3.5 ml. of which were fractionated on columns of EGTEDLA cellulose, 0.1 ml. portions of each 10 ml. eluste fraction being plated on stainless steel planchettes and counted. Eluste fractions containing either THF, TDP or TTP were hydrolysed and the edenosine nucleotides removed as described in the text (Section 2.2.4 (b)) and the total extinction at 267 mu measured in the recovered reaction products. The source of enzymes was a solution prepared from Lyophylised cell-free extracts of Landschutz escites cercinome.

utilisation of TTP. Such a situation may be visualised in vivo when the cell is actively synthesising DNA.

Thus, diroumstantially, all the evidence presented in the present Chapter supports the hypothesis of Weissman, Smellie and Paul (1960) that the formation of TTP from thymidine proceeds in a stepwise manner according to the remotion sequence:

TOR THE THE TOP TTP

However, these experiments failed to establish beyond all doubt the participation of an enzyme capable of catalysing specifically the phosphorylation of TMP in the synthesis of TTP and attempts to resolve this uncertainty is the subject of the investigation reported in Chapter III.
GHAPTER III

"PRELIMINARY BNZYMOLOCICAL STUDIES ON THE THYMIDINE AND THYMIDYLATE KINASES OF LANDSCHUTZ ASCITES TUMOUR CELLS."

3.1. Introduction

General properties of nucleoside and nucleotide phosphokinases

The phosphorylation of nucleosides and nucleotides to the corresponding nucleoside 5'-triphosphates (see Figs. 10 and 11) has been presumed to occur under the influence of a series of phosphate transferring ensymes, termed phosphoryl group of a sucleoside triphosphate to the transfer of the terminal phosphoryl group of a sucleoside triphosphate to the acceptor molecule (see Section 1.8.(d)). Such kinases belong to a class of ensymes which cat-lyze coupled remetions involving the transfer of energy and, as such, they are normally considered markedly exergonic; the catalyzed remetions, therefore, have equilibria strongly in favour of the production of phosphorylated products. A large number of such ensyme activities has been described, but in the absence of discriminating kinetic date, it is frequently difficult to decide whether the reported ensymes represent unique molecular species or different forms of the same enzyme.

Griticel scrutiny of the aveilable data shows that only very few of the reported nucleoside and nucleotide kinases satisfy the rigid set of criteria demanded of specific enzymes while information on a number of other kinases is no more than the evidence of their existence. A few well known kinases, notably myckinase (see p. 25), have been intensively studied and it now seems probable that the reaction mechanism involves a nucleophilic attack by the acceptor hydroxyl group on the terminal phosphorus atom of the nucleoside triphosphate in the following manner:-



That the reaction catalysed by myckinese involves the transfer of a terminal phosphoryl group, and not a phosphate group, from ATP to AMP was demonstrated by Cohn (1956) with the use of 180-labelled substrates and there is currently reason to think that phosphoryl group transfer from the donor is a general feature of kinase ontalysed reactions. Most kinase ontalysed reactions appear to possess of optime of 7 - 9. a region where nucleotides can be considered to exist in their completely dissociated forms. One further important feature of enzymic phosphoryl group transfer reactions seems to be their dependence on the presence of di-valent cations, usually Mg ++ or Mn ++. The generally accepted explanation for the Mg ++ dependence of many kinese reactions is that the ion forms a chelate with the terminel phosphete groups of the aucleoside triphosphate and, in so doing, screens a number of negative charges on phosphate oxygens of both donor molecules and acceptor molecules (Lowenstein, 1958). However, there is little information as to the precise details of the mechanism or the sequence of

events leading through chelstion, binding of substrates at the active sites and the transfer of the phosphoryl group; but of the kinases so far studied sufficient evidence is available to justify the conclusion of Noda (1968) that: "Undoubtedly the mechanism of phosphoryl transfer is all the so-called high-energy phosphate-transferring enzymes is similar, and as the picture is pieced together for one enzyme the principles will become generally applicable

Since the kinness ontalyse coupled reactions it seems reasonable to suppose the existence of two binding sites, one for each nucleotide substrate, and, in the case of myokinese, it has in fact been found that AMP and ATP bind specifically at adjacent sites on the enzyme surface (Kuby, Mahowald and Noltmann, 1962). One consequence of this finding is that the kinases exhibit double substrate specificity, <u>i.g.</u>, specificity towards donor and specificity towards acceptor molecules. The majority of purified kinases prefer ATP as the phosphoryl group donor, but several systems have been described where other nucleoside 5'-triphosphates can substitute for ATP.

Nucleoside kineses are thought to estalyse the phosphorylation of a nucleoside to the corresponding nucleoside 5'-monophosphete. Such reactions are normally regarded as part of the preformed pathways in the formation of nucleosides and precursors. Kinese catalysed renations involving all the ribonucleosides and decayribonucleosides which occur as major components of the nucleosides are known to take place. However, only in the case of adenosime kinese (Caputto, 1951; Kornberg and Pricer, 1951) and uridine kinese (Skbld, 1960) have these activities been shown to be due to specific ensymes. With both types of preparation strict stoichiometry was observed and ATP was the preferred phosphoryl group donor.

Enzymes which catalyse the transfer of a phosphoryl group from nucleoside 5' -triphosphetes to nucleoside 5' -monophosphetes are widely distributed and include enzymes exhibiting a wide range of substrate specificities. At the extreme and of the scale, monimum requires the presence of MTP and MEP while, at the other and, the ATP-nucleoside menophosphate kinase from onlf liver (Strominger, Heppel and Maxwell, 1959) requires ATP as phosphate donor but can utilise AMP, OMP, SMP and UMP as acceptors. Similarly, the nucleoside triphosphate-AMP kinese from the same source (Heppel, Strominger and Maxwell, 1959) is specific for AMP but the nucleoside triphosphate could be ITP, ATP, GTP, CTP or UTP. Severel enzymes which phosphorylate decayribonucleoside 5' -monophosphates have also been reported, but only very few exhibit definable specificity. In particula it seems uncertain whether the substrate specificity of such kinases extends to distinction between decxyriboayl and riboayl moieties of the same base. This uncertainty steas from the properties of an ATP-dCMP kinase discovered in extracts of Azotobacter vinelandii (Maley and Ochos, 1958) which, after purification, was shown to gatalyse the phosphorylation of GUP at 60 per cent the rate of dGMP while being inactive on other nucleoside monophosphates. Similarly, a dGMP kinese purified from 3. coli (Bessman, 1963) appears to phosphorylate GMP at the same rate as dGMP while ATP is required as the domer of the phosphoryl group.

Further phosphorylation of the nucleoside 5' -diphosphates produced by the above reactions has been attributed to the action of nucleoside di-

- 108 -

phosphokinsses. Such enzymes are known to have a wide distribution, but the reactions which they catalyse are not well characterised and a specific nucleoside diphosphokinsse has yet to be discovered. Thus, the ensyme from rebbit muscle (Berg and Joklik, 1954) appeared to require ATP while those derived from yeast (Berg and Joklik, 1954) and from rat liver mitochondris (Thigs and Plaut, 1959, 1968) could utilize ITP, ATP, OTP, OTP and UTP; all preparations could use any ribonucleoside 5'-diphosphate as the phosphoryl group acceptor. Little evidence is available on the existence of ensymes capable of phosphorylating decayribonucleoside 5'-diphosphates but it seems probable that such reactions do occur as part of the <u>de novo</u> pethway for the formation of dOTP, dATP, dOTP and TTP. Evidence is presented elsewhere (see Fig. 25) that an ATP-TDP kinsse occurs in Landschutz escites tumour cells.

In the preceding Ghapter, a number of experiments have been described which strongly support the hypothesis (Weissman, Smellie and Paul, 1960) that the formation of TTP from TdR proceeds by stepwise phosphorylation through TMP and TDP. It seems plausible, therefore, to suggest that the reaction is out lyped by a series of at least three phosphokinases, each responsible for on step in the sequence:

TAR quanta TMP quant TMP - TTP

For the sake of brevity, it is proposed to give the names TdR kinase, THP kinase and TDP kinase to the enzymes catalysing reactions a, b and c, respectively. In the present Ghapter, an attempt is made to characterise and resolve the relevant kinase activities from extracts of Landschutz ascites tumour cells.

3.2. Experimental

3.2.1. General methods

The investigations recorded in Chapters 2 and 3 overlap sufficiently for a number of the experimental procedures given in Section 2.2. to be applicable to problems concerned with the enzymology of the thymidine and thymidylate kinases. Such methods include the proparation of substrates (Section 2.2.2.), the measurement of thymidine and thymidylate kinase activities (Section 2.2.4) and various analytical methods to which the appropriate references are made. Only those experimental methods which have been modified in any fundamental way to satisfy new requirements or new analytical procedures are described in detail below. Some additional techniques are also given in the legends to the figures and the tables.

3.2.2. Biological material

(a) Extracts from Landschutz ascites tumour cells

The source of enzymes for the majority of the experiments described hereafter has been an accumulated pool of lyophylised cell-free extracts from Landschutz ascites carcinoms prepared as described earlier (Section 2.2.1.). This pool of lyophylised material was derived from about 400 tumour bearing mice of the departmental colony, ascites fluid being withdrawn from 50 mice per preparation, and the resulting lyophylised powder carefully mixed with earlier preparations so as to give a relatively homogeneous stock from which material could be withdrawn for the ensuing enzyme experiments. Evidence for the homogeneity of the enzyme source was repeatedly obtained by determining the initial activity of the relevant enzymes. In such experiments, the initial activity showed insignificant variation over a period of months. Further information on the stability of the thymidize and thymidylate kineses is given elsewhere (Section 3.3.3.).

(b) Extracts from normal rat liver

While albino rate, weighing from 180 to 220 g., from the departmental colony were employed. The rate were annesthetised with other and the liver excised after severing the hepstic entery, and allowing the blood to drain. Immediately after removal, the liver was chilled on crushed ice, out into small pieces with solssors, weighed and homogenised in 8 volumes of ice-cold 0.01 <u>M</u> tris buffer, pH 7.5, in a cooled Potter-type homogeniser. After the initial suspension of the tissue, about 5 passes of the homogeniser peatle was found to give an extract relatively free from whole cells without disrupting many nuclei, as judged by microscopic examination of wet smears, steined with 1 per cent crystel violet is 0.1 <u>M</u> sitric acid. The resulting homogenete was centrifuged in a Servall Type SS-4 Superspeed centrifuge at 27 000 x g for 1 hour at 0⁰, and the practicelly clear supernatent fluid decented. This extract had a protein concentration of 5 - 7 mg. per ml. and was either used directly in enzyme experiments or lyophylised and stored at -50⁰.

3.2.3. Procedures used for the assay of enzymes

(a) Thymidine kinese

While the assay procedure described earlier (Section 2.2.4 (a))

- 111 -

provided a precise means for the determination of each of the products (TdR, TMP, TDP and TTP) involved in the formation of TTP from TdR, this procedure was found to be unsuitable for the purpose of assessing TdR kinnse activity in the course of enzyme fractionation experiments. The repid assay which was developed for TdR kinnse followed in principle the earlier procedure but involved less rigorous separation of the remotion products. This assay was based on the determination of the total radioactivity found in all phosphorylated remotion products when the substrate was (³H) TdR.

The following reaction mixture was prepared: - 100 µmoles of tris buffer, pH 8.0, 5 µmoles ATP, 5 µmoles of MgGl2, 0.1 µmoles of 2-mercaptoethenol, 1 µ0 (3E) Tak of specific activity 360 mc per mole and enzyme solution containing various quantities of protein as given in the legends to the figures and the tables. The total volume of the reaction mixture was 1.0 ml. Enzyme solutions, when prepared from the stock of lyophylised cellfree orude extracts, were made up in distilled water to give a protein concentration of 6 - 8 mg. per ml. Incubation was for various times in a shaking water bath at 37°. The reactions were terminated by heating for 3 minutes at 100° and all subsequent manipulations were carried out in the cold. Congulated protein was removed by centrifugation and washed twice with 0.5 ml. distilled water each time, adding the washings to the supernatent fluid. The solution was then diluted to 3.0 ml. with distilled water. A portion of this solution (2.0 ml.) was adsorbed on a column of ECTEOLA cellulose (10 x 1.5 cm., die.) and the TdR and thymidine nucleotides were eluted by applying successively 150 ml. distilled water and 100 ml. 0.2 H

- 118 -

hydrochloris acid. The two elustes were diluted separately to finite volumes, 0.1 ml. portions of which were plated on stainless steel planchettes and counted in a Nuclear-Chicage windowless gas-flow counter. The results were expressed in terms of the percentage of the total radioactivity recovered in all phosphorylated thymidine derivatives (THP + TDP + TTP) or as the number of mumoles of TdR phosphorylated.

Preliminary experiments had shown that the water fraction contained TdR only while the HGl fraction contained the TMP, TDP and TTP. Thus the method provided a direct means of measuring the total phosphorylation of TdR.

In order to assess the recovery of total radioactivity, blanks in which enzyme solution was omitted were included in each series of experiments. Such blank determinations showed a consistent recovery of total radioactivity of 80 - 90 per cent. In experiments where the specific activity of TdR kinese was sought, appropriate corrections were made for the blank values. A further check on the validity of the assay procedure was by paper chromatography. The method adopted involved taking a portion (0.5 ml.) from the assay solution prior to the separation on ECTEDIA cellulose, concentrating the reaction products by lyophylisation and separating the thymidine nucleotides by chromatography on paper as described earlier (Section 3.2.3 (b)). The fully developed chromatogram was then scenned for radioactivity by means of the Actigraph scenning device (Section 2.2.3 (d)).

Several preliminary experiments had shown that TMP was relatively resistant to non-enzymic hydrolysis (see also Section 2.2.4 (b)) and placing reaction mixtures in which (^SH) TdR was replaced by 5 µmoles of TMP in a boiling water-bath for periods up to 10 minutes did not result in the production of sufficient TdR to be detectable by paper chrometography. It would, therefore, appear that the pattern of total phosphorylation of (^{3}H) TdR remained undisturbed whether the TdR kinnse was inactivated by heat or by the use of perchloric acid (Section 2.2.4 (a)).

(b) Radioactivity assay for ThP kinase

On a few occessions when it was necessary to obtain exact data on the relative amounts of each of the reaction products, the original kinnse assay (Section 2.2.4 (a)) was employed. On most other occasions, however, this procedure proved too time consuming and a repid assay was developed based on a principle similar to that underlying the repid assay for TdR kinnse (see above). Thus the repid assay measured the total phosphorylation of (32p) TMP by determination of the total redicectivity found in the phosphorylated derivatives (TDP + TTP).

The following reaction mixture was prepared in a total volume of 1.0 ml.:- 100 µmoles tris buffer, pH 7.6, 5 µmoles ATP, 5 µmoles MgGl₂ and (³²p) TMP and enzyme solution in the quantities given in the legends to the figures and the tables. In experiments where the TMP kinnse activity of the hypophylised orude escites extract stock was measured, the dry material was dissolved in distilled water or in 0.01 <u>M</u> phosphate buffer, pH 7.6, to give solutions containing 6 - 8 mg. protein per ml. Incubation was performed for various times in a sheking water-bath at 37°. The remotions were terminated by heating at 100° for 8 minutes and subsequent manipulations were carried out in the cold. The precipitated protein was removed by centri-

fugation and washed twice with 0.5 ml. portions of 0.01 M phosphete buffer, pH 7.6, the washings and supernstant fluid were combined and the solution diluted to 3.0 ml. A portion of this solution (1.0 ml.) was adsorbed on a column of ECTEOIA cellulose (10 x 1.5 cm., die.) and the thymidine nucleotides eluted by applying successively 150 ml. of 0.015 N hydrochloric acid and 100 ml. of 0.2 % hydrochloric acid. TMP was recovered in the 0.015 M HCl fraction and TDP and TTP were eluted in the 0.2 N HGl fraction. The two fractions were separately diluted to finite volumes, 0.1 ml. portions of which were plated on stainless steel planchettes and counted in a Muclear-Chiongo gasflow counter fitted with a thin and window. The results were expressed in terms of the percentage of the total redicactivity recovered in thymidine polyphosphetes (TDP + TTP) or as the number of mumoles of TMP phosphorylated. Control incubations in which enzyme solution had been omitted showed that 90 - 95 per cent of the total redicectivity present initially was consistently recovered.

Preliminary elution experiments had shown that complete resolution of TMP and thymidine polyphosphates could be effected by this procedure which, therefore, provided a measure of the total phosphorylation of TMP. Other experiments had indicated that the two reaction products, TDP and TTP, were considerably more thermolabile than TMP and that inactivation of the ensyme system by heating at 100° was likely to result in some hydrolysis of these products. However, it was found that heating of the reaction mixture at 100° for 2 minutes led to only marginel degradation of TDP and TTP while the treatment sufficed to compulate the protein and inactivate the ensyme system.

Additional data on the proportion of the reaction products due to the action of TMP kinase and the TDP kinase was obtained by mper chrometography. In such experiments, a portion (1.5 ml.) of the deproteinised essay solution at the stage prior to adsorption on ECTEOLA cellulose was lyophylised, redissolved in 0.1 ml. of distilled water and a 0.05 ml. portion of this solution was submitted to descending chrometography on paper as described earlier (Section 2.2.3 (b)). The positions of the radioactive areas on the chromatogram were ascertained by the use of the Nuclear-Chicago Actigraph system (Section 8.2.3 (d)). Moreover, since (32P) -labelled THP was employed as substrate in such experiments and as the efficiency of the windowless gasflow chamber for this isotope was relatively high, it proved possible to obtain semi-quantitative estimates of the three nucleotides TMP, TDP and TTP by scenning appropriate fully developed chrone togram strips. Such estimates were obtained by weighing relevant peaks out from the recorder chart. The velidity and reproducibility of this procedure was checked by cutting relevant redicective areas from the chromatogram strip, eluting the nucleotides by capillary flow in distilled water and counting portions of the elustes on stainless steel planchettes in a Nuclear Chicago windowless gas-flow counter. Although the kinese activity as measured by the above procedure primarily reflects THP kinase activity, it was possible to obtain a measure of the TDP kinase activity by suitable correlation of the data from the two types of product fractionation outlined above. TDP kinase activity was not measured in enzyme fractions in which TMP kinase activity was absent.

- 116 -

(c) Spectrophotometric assay for TMP kinnse

A spectrophotometric same for TMP kinese was developed on the basis of procedures described for the measurement of adenosine kinese (Kornberg and Pricer, 1951) and myckinese (Lieberman, Kornberg and Simpe, 1955) activities. This method mensured the rate of formation of nucleoside diphosphates by coupling their formation to the exidation of NADH. In the presence of excess lactic dehydrogenese, pyruvete kinese, phospho(emol)pyruvete and NADH, ADP was quantitatively converted to ATP with the simultaneo stoichiometric exidation of NADH and the following sequence of recetions can be visualised:-

THP + ATP \longrightarrow TDP + ADP Phospho(enol)pyruvate + ADP \rightarrow pyruvate + ATP Pyruvate + NADE \rightarrow lactate + NAD⁺

Thus, if TMP kinese can be considered the rate limiting factor in the reaction the disappearance of MADH as measured by a decrease in extinction at 340 mu should be a direct measure of the TMP kinese activity. However, on account of the nonspecific nucleoside diphosphokineses present in pyruvate kinese preparations (Klenow and Lichter, 1957), both ADP and TDP were converted to their respective triphosphates. Thus 2 moles of MADH were oxidised for every mole of TMP phosphorylated.

A reaction mixture containing the following components in a total volume of 2.29 ml. was prepared in a 10 m.m. quarts suvette: - 0.1 µmoles TMP, 10 µmoles ATP, 10 µmoles MgGl₂, 0.25 µmoles MADH, 0.5 µmoles phosphe(emol)pyruvate, 300 µmoles tris buffer, pH 7.6, 30 µg. lactic dehydrogenese, 20 µg. pyruvate kinase and enzyme solution. The reaction was initiated by the addition of TMP and the rate of NADH oxidation followed by the decrease of extinction at 340 mµ in a Beckman Model DU spectrophotometer fitted with a chart recorder. All experiments were conducted at 30°. The average decrease in extinction over a 10 minute period was determined in each experiment. A time curve of the spectrophotometric procedure for two different enzyme concentrations is shown in Fig. 23. It beceme apparent, however, that the TMP kinase preparations frequently contained phosphateses acting on ATP thus distorting the pattern of NADH oxidation due to TMP kinase end in such cases the observed reaction rate was corrected by subtracting the values obtained in reactions where TMP had been omitted (see also Section 3.2.3 (e)). The results were calculated as mumoles TMP phosphorylated per 25 minutes per ml. enzyme solution or per mg. protein.

The assay could not be used for measuring TMP kinnse in the crude escites extract because of endogenous phosphetases and NADH exidases and it was not used for other than highly purified TMP kinnse preparations. In a series of determinations with 50, 100, 800 and 400 µg. of protein from the SGgAS 4.7 fraction (see Section 3.3.5 (a)) the respective corrected ΔE_{340} values of 0.062, 0.060, 0.065 and 0.063 per minute per mg. protein were obtained thus reflecting a linear relationship between reaction rate and enzyme concentration.

The spectrophotometric assay was occasionally also used for the quantitative measurement of nucleoside diphosphates. When the assay was employed for this purpose, the TMP kinasë was omitted from the reaction



Fig. 28. Time course of TMP kinese action as followed spectrophotometrically at two concentrations of enzyme.

> The remotion mixture contained the components given in the text (Section 3.2.3 (a)) in a total colume of 2.29 ml. The arrows indicate the addition of TMP kinase and TMP, respectively, and the exidation of NADH was followed spectrophotometrically by the decrease in the extinction at 340 mm and at 30°. The drop in optical density observed upon addition of TMP kinase was due to the ATPes contaminant in the enzyme preparation employed, the source of enzyme being the SGY AS 4.7 Fraction (see Section 3.3.5, Method I). The corrected TMP kinase remotion rates ($\frac{\Delta \Xi_{340}}{\Delta_{3}}$) were 0.009 per minute for 0.1 ml. enzyme solution (curve B). In this situation, the TMP kinase solution (curve B). In this situation, the TMP kinase solution (curve B).

1.145
$$\begin{bmatrix} \Delta = 340 \\ \hline \Delta = 10^3 \\ \hline \epsilon_{340} \end{bmatrix}$$
 THP
$$\begin{bmatrix} \Delta = 340 \\ \hline \Delta = 10^3 \\ \hline \epsilon_{340} \end{bmatrix}$$
 ATP

a uncles TMP phosphorylated

where the suffixes ATP and TMP refer to the reaction rate observed after the addition of TMP kinase and TMP respectively. 1 Unit of TMP kinase activity was defined as 1 mumole of TMP phosphorylated per 85 minutes. mixture and the extinction of NADH was followed at 340 mm until the reaction had subsided and the extinction reached a low, constant value. The total amount of NADH exidised was corrected for the exidation observed in reaction mixtures where the nucleoside diphosphates were absent.

In all spectrophotometric assays, highly purified ATP was required in order to reduce the blank values to a minimum. The ATP employed in the present experiments (Signs Chemical Corporation) was shown spectrophotometrically to contain less than 1 per cent ADP. Pyruwate kinnse, lactic dehydrogenese, phospho(enol)pyruwate and NADH were obtained from Biochemica "Boeringer", Mannheim. The Molar extinction coefficient of the NADH supplied was shown to be 6.2 x 10³ cm.² per mole when measured at 340 mµ.

(d) Assay of decrycytidine and decrycytidylate kinases

The decay cytidine and decay cytidylate kinases were esserved by a procedure similar to that described elsewhere (Section 2.2.4 (a)) for the thymidi. and thymidylate kinases. In practice, it proved difficult to secure sufficient resolution of the decay cytidine nucleotides by elution from columns of ECTEDIA collulose and a procedure was developed which incorporated deamination of the decay cytidine nucleotides to the corresponding decayuridine nucleotides prior to the separation of reaction products on ECTEDIA collulose. The demmination method was based on the procedure described by Birmie (1959).

The reaction mixtures contained the following components in a total volume of 3.0 ml. :- 300 μ moles tris buffer, pH 7.9, 15 μ moles HgGl₂, 15 μ moles ATP, 0.5 μ moles 2-meromptoethenol, 3 μ s (³H) GdR of specific activity 1100 mc per muche and enzyme solution in the quantities given in the legends to the tables. The enzyme solutions were derived from hypphylised extracts of ascites carcinoms or from rat liver and contained 6 - 8 mg. protein per ml. The remotion mixtures were incubated for various times in a shaking waterbeth at 37°.

The reactions were termineted by the addition of 1.5 ml. of 2.1 M perchloric soid and the precipitated protein removed by centrifugation at 0". The residue was washed three times with 0.5 ml. portions of 0.5 N perchlorie acid and the washings and the supermetant fluid combined. The solution was brought to pH 7.5 with 7 N potessium hydroxide, allowed to stand overnight in the cold, centrifuged to remove the precipitate of potessium perchlorete and diluted to 7.0 ml. with distilled water. A portion (1.0 ml.) was reserve for estimation of total radioactivity and paper chromatography, and of the remainder, 5.0 ml. was employed for the conversion of decryoytidine nucleotides to the decuyuridine derivatives as follows: - After concentration by freeze-drying in a 15 ml. conical centrifuge tube, 350 µl. RN NaNDy and 70 µl. glacial agetic acid was added to the residue and the contents of the tube thoroughly mixed. The reaction mixture was incubated in a shaking water-bath for 5 hours at 40°. A considerable amount of the potessium perchlorate prese: in the lyophylised residue failed to dissolve in the reaction medium, but control experiments showed that this material was without effect on the deamination process. The precipitate was removed by centrifugation at 0° at the end of the treatment, washed twice with 0.5 ml. portions of distilled water and the washings and the supernatant fluid were combined. The solution was adjusted to pH 7.5 with 7 <u>H</u> potessium hydroxide and finally diluted to 3.0 ml. A portion (1.0 ml.) of this solution was run on to a column of EGTEOIA cellulose (10 x 1.5 cm., dia.) and the decayuridine nucleotides recovered by successively eluting the column with 100 ml. of distilled water, 100 ml. of 0.08 <u>M</u> hydrochloric sold, 50 ml. of 0.035 <u>M</u> hydrochloric sold, 100 ml. of 0.05 <u>M</u> hydrochloric sold and 70 ml. of 0.2 <u>M</u> hydrochloric sold. UdR appeared in the water fraction, dUMP in the 0.02 <u>M</u> Holl fraction, dUDP in the 0.05 <u>M</u> HGl fraction and dUTP in the 0.2 <u>M</u> HGl fraction. The column flow rate was 20 ml. per hour and portions of 10 ml. were collected. Samples (0.1 ml.) were removed from each portion, plated on stainless steel planchetter and counted in a Hudleer-Chicago windowless gas-flow counter. The pattern of elution of UdR, dUMP, dUDP and dUTP is shown in Fig. 29. The results were calculated as the percentage of the total redioactivity recovered is each of the resction products.

A series of control experiments with unlabelled GdR, dGMP and dGTP had shown that the domination was quantitative unlar the conditions described above. Such control experiments were conducted by the use of paper chromatography (Section 2.2.3 (b)) in the ammonium isobutyrate system in which the R_{f} values of the nucleotides of the decryoptidine series and decryuridine series differed sufficiently to indicate the progress of the reaction (see Table 1). That the domination was quantitative was also shown by running the domination experiments in a Gary Recording spectrophotometer with measurements at regular time intervals of the ratio E_{260}/E_{260} and recording the changes in the ultraviolet spectra of the reaction mixtures.

- 121 -



Fig. 29. Chrometographic separation of (³H) -labelled UdR, dMUP, dUDP and dUTP on columns of ROTEOLA cellulose (10 x 1.5 om., dia.) by stepwise elution with increasing concentrations of hydrochloric soid.

> The (³H) -labelled decayuridine derivatives were obtained from the corresponding (³H) -labelled decayoytidine derivatives by demination with nitrous acid as described in the text (Section 3.8.3 (d)).

In general, it was found that the decay cytidine nucleotides were relatively resistant to non-ensymic demnination and complete transformation to decay uridine derivatives required incubation at 40° for 4 - 5 hours in presence of a large excess of acidified NaNO₂, while the demnination of GdR was complete within 1 hour under similar reaction conditions.

(e) Phosphetase espers

Although phosphetase activities were not routinely determined, it was eccasionally necessary to have access to the absolute values of kinase activities in which case the apparent kinase activity was corrected for the competitive hydrolysis of reaction products due to endogenous phosphetases. Thus, no attempt was made to measure optimal phosphetase activities and the only values that were considered were those obtainable in an environment dictated by the parameters of optimal kinase activity.

An essey for phosphetases acting on TDP or TTP was developed on the basis of the redicectivity assay for TMP kinese (Section 3.2.3 (b)) in which (^{32}P) -labelled TDP or TTP replaced (^{32}P) TMP and ATP was omitted. Reaction mixtures were prepared containing the following components in a total volume of 1.0 ml.:- 100 µmoles tris buffer, pH 7.6, 5 µmoles MgGl₂, 20 mµmoles $(^{32}P - a)$ TDP or $(^{32}P - a)$ TTP of specific activity 10⁷ counts per minute per µmole and enzyme solution in the quantities given in the legends to the tables. Incubation was in shaking water-bath for 25 minutes at 27°. The reactions were stopped by heating at 100° for 2 minutes and the congulated protein removed by centrifugation at 0°. Each protein sediment was washed twice with 0.5 ml. portions of 0.01 M phosphete buffer, pH 7.6, the washings and supernetent fluid combined and the solution diluted to 3.0 ml. A portion (1.5 ml. from each solution was lyophylised and the resulting residue dissolved 0.1 al. of distilled water, 0.02 ml. of which being taken for chrometography on paper (Section 2.2.3 (b)). The percentage of total radioactivity present in each of the products was determined by scenning the appropriate chrometogram strips in the Actigraph windowless gas-flow detector (Section 2.2.3 (d)) and subsequently outting and weighing the relevant peaks from the recorder chart. The validity of the results so obtained was checked by cutting out radioactive sections from the chromatogram strip and eluting the nucleotide material by capillary flow in distilled water and diluting each eluate to 2.0 ml., 0.1 ml. portions of which were plated on steinless steel planchettes and counted in a Nuclear-Chicago windowless gas-flow counter. The results were calculated as mumoles TDP or TTP hydrolysed per 25 minutes per mg. protein. Controls were included in which enzyme was omitted in order to correct for non-enzymie hydrolysis of the substrates in the course of incubation and heat inactivation

Phosphetases acting on ATP (ATPases) were frequently found in the crude ascites extracts and at several stages of the purification of TAP kinase. ATPase was measured by means of the spectrophotometric procedure described above (Section 3.2.3 (c)) in which TMP was omitted. The reaction mixtures (2.29 ml.) contained the following components: - 10 µmoles ATP, 10 µmoles MgCl_g, 200 µmoles tris buffer, pH 7.6, 015 µmoles phospho(enol)pyruvate, 0.25 µmoles NADH, 20 µg. pyruvate kinase, 20 µg. lactic dehydrogeness and enzyme in the quantities given in the legends to the figures. The reaction mixtures were prepared in 10 mm. quartz cuvettes and the rate of NADH oxidation followed spectrophotometrically at 240 mµ and 30°. In order to obtain reproducible results, pyruvate kinese was added only after temperature equilibration (3 minutes). The subsequent drop in $_{300}$ was adjudged to be due to small amounts of ADP present in the ATP employed. Enzyme solution was added after the E_{340} had attained a constant value (Fig. 30). The average decrease in the extinction at 340 mµ over a 10 minute period using a tungstan light source was determined and the results were calculated as µmoles ADP formed per 25 minutes per mg. protein.

3.2.4. Ensyme fractionation techniques

(a) Fractionation with dilute a cetic acid

Fractionation of grude assitues extracts or extracts from lyophylised material from the same source formed an integral part of the majority of the purification procedures outlined in Section 3.3. In experiments where the source of enzymes was the lyophylised grude extract pool (Section 3.8.2 (a)), the material was dissolved in 0.01 M phosphate buffer, pH 7.6 or pH 7.8, for purification of TMP kinese and TdR kinese, respectively, to give solutions containing 5 - 5 mg. protein per ml. The details of the fractionation procedure in which all manipulations were carried out at 0° were as follows:-A volume (10 - 200 ml.) of enzyme solution, from which a sample had been removed to serve as control for the enzyme under study, was transferred to a 50 ml. round-bottomed centrifuge tube or to a beaker of appropriate size. The solution was stirred mechanically is an ice-bath at 0° and the imitial pH measured with electrodes previously adjusted to operate over the range pH 4.5 - 7.5. To this solution was then edded 1.0 H mostic moid drop by



Fig. 30. The time course of ATPase action as followed spectrophotometrically.

> The reaction mixtures contained the components given in the text (see Section 5.8.3 (e)) in a total volume of 8.29 ml. and the arrows indicate the addition of pyruvate kinase and ATPase, respectively. The rate of disappearance of NADR was measured in a Beckman DU spectrophotometer at 340 mm and at 30°. The drop in total extinction observe upon addition of pyruvate kinase is presumably due to the small amount of ADP present as a contaminant of the ATP used. The source of enzyme was the ATPase containing fraction of the effluent from an enzyme fractionation experiment with a column of Sepheder G-100 (see Section 3.3.5, Method II). The ATPase remotion rate ($\frac{\Delta E_{340}}{\Delta t}$), was 0.014 per minute and the ATPase activity was given as

$$\frac{\Delta \frac{1}{340} \times 10^{3}}{(\frac{\Delta 5}{6} \times 10^{3})}$$

umoles ATP dephosphory lated

drop until the desired pH hed been attained, the resulting mixture quickly transferred to centrifuge tubes and centrifuged at 2500 rev. per minute for 7 minutes. The supernatent fluid was returned to the pH mater and the pH quickly readjusted to 7.6 with the addition of 1.0 M sodium hydroxide while stirring the solution mechanically in an ice-bath. The sediment, if required, was added a volume of distilled water equivalent to one-half the original volume of the solution and the predipitate redissolved by adjusting the pH to 7.5 as above. Both the neutralised fractions so obtained were diluted to the original volume, assayed for the appropriate enzyme activities and the protein gencentrations determined.

In order to evoid serious losses of enzyme activities due to contect with the dilute acetic acid, it was essential to reduce the duration of this treatment to a minimum and the time taken over the whole operation was, consequently, not allowed to exceed 15 minutes (see also Section 3.3.3). Strict control of pH was necessary to achieve reproducible recoveries of TdR kinnes and TMP kinnes and the validity of the pH mater scale readings was frequently checked in the course of the operation with the use of standard buffers. A nerrow combination electrode was used in conjunction with a Beckman Zeromatic pH mater for fractionations involving 10 - 20 ml. of enzyme solution in which case the whole operation was carried out in a 50 ml. centrifuge tube. For fractionations involving 20 - 200 mll of enzyme solution a Direct Reading pH mater (Model 23A, Electronic Instrument Limited, Surrey) connected to a conventional electrode assembly was found more suitable Largor volumes than 200 ml. could not conveniently be headled, the upper limit being dictated by the time required for the completion of the acid fractionation in each case. Thus, in experiments where the original volume of enzyme solution exceeded 200 ml., fractionations were conducted as a series of experiments using small volumes of enzyme solution each time and combining the resulting fractions where appropriate.

(b) Fractionstions using 0 % -elumine mel

The $G \not=$ -sluming gel employed in the present investigation was prepared according to the method of Willstätter and Kraut (1983) and stored in equeous suspension containing 11 mg. AlgO₃ ' 3 HgO per ml. for a period of 4 years.

The procedure developed for fractionations using G X -elumina gel involved measurement of the kinese activity remaining in the supermatent fluid under various conditions after the addition of increasing quantities of GX -alumine suspension. The ensyme solution was prepared by dissolving the appropriate enzyme in 0.01 M phosphate buffer, pH 7.6, to give a protein concentration of 4 - 8 mg. per ml. and all subsequent manipulations were entried out at 0°. In small scale experiments, portions of the equeous $0 \times$ -elumine suspension ranging from 0.5 ml. to 10 ml. wre transferred to 15 ml. conical centrifuge tubes and the gel packed tightly in the tube by centrifugation at 1000 x g for 20 minutes at 0°, the aqueous supermatant fluid being discarded. To the gel was then added 5 ml. of 0.01 M phosphate buffer, pH 7.6, whereupon the resulting suspensions were vigorously stirred, centrifuged at 1000 x g for 10 minutes and the supernatant fluids discarded. Washing in this manner with 0.01 M phosphate buffer was repeated twice more; finally, 10 ml. of buffer was added to the gel and the suspensions left

overnight at 0° in order to ensure that the gel - buffer system had become fully equilibrated. After centrifugation and decontation, a 1.0 ml. portion of ensure solution was added to each tube and the gel dispersed by stirring the suspensions manually from time to time over a 5 minute period. Thereafter, the mixtures were left at 0° for 30 minutes to complete the adsorption. The gel was then sedimented by centrifugation at 1000 x g for 15 minutes and the supermatant fluid reserved for mensurement of protein concentration and for the assay for the appropriate kinase. Equivalent values representing the initial enzyme activity were obtained by incubating portions of untreated enzyme solution. From these two sets of date, the fraction of the appropriate kinase activity adsorbed under the prevailing conditions could be calculated.

(c) The use of gel filtration techniques in ensyme fractions tions

The uncharged dextran gels Sephedex G-85 and Sephedex G-100, obtained from AB Pharmacia, Uppeals, were employed in the present investigation either as a substitute for conventional dialysis or for the purpose of enzyme fractionation in which case the dextran operated as a molecular sieve.

Pretreatment of Sephadex G-100 (200 - 400 mesh) was performed by adding the dry dextran powder to excess 0.01 M phosphate buffer, pH 7.6, with stirring thus evoiding the formation of elumps, and the gel - buffer system was allowed to equilibrate over a 24 hour period. While the process of swelling was in progress it was found convenient to remove fine particles by stirring the suspension and then allowing the gel to sediment. When a sharp boundary layer became visible, the supermatent fluid was removed by suction and this treatment was repeated until the supermatent fluid was clear. After equilibration, the gel suspension was further diluted to a viscosity low enough to allow rapid escape of air bubbles during the packing of columns.

Careful packing of the column was essential in order to obtain reproducible enzyme fractionations. The following procedure was adopted to ensure the preparation of a completely honogeneous bed of destran particles: -A column (1 cm., dis.), fitted with a small plug of cellulose wadding above the outlet tube, was carefully mounted vertically and filled to one-third of the column height with 0.01 M phosphete buffer, pH 7.6, the outlet being closed. The ensuing pecking process was conducted at room temperature to prevent the formation of air bubbles on the inside sufface of the glass column The column was then filled with gel suspension which immediately started to sediment. After a layer of 4 - 5 cm. had formed, the outlet tap was opened so as to allow the escape of a slow stream of buffer and the progress of the packing was followed visually from the appearance of the horisontal bed surface. More gel suspension was added before all previously added gel particles had settled and the process repeated until the column surface had reached a level 2 - 3 on. above the desired bed height. A small volume of buffer was then added with the escape tap closed and the top layer of the bed agitated with a thin glass rod whereupon the particles were allowed to resettle. This operation was repeated until a perfectly horizontal bed surface had formed. The bed surface so prepared was protected mechanically by placing a well-fitting disc of glass filter paper in contact with the surface. The gel was then washed with 2 - 3 column volumes of phosphate buffer with a flow fate of 15 - 30 ml. per hour until the column was stable. Because of the

- 128 -

danger of gradual packing of the gel particles with consequent decrease in the flow rate, a constant hydrostatic pressure head was not used, the eluent buffer being added instead in small volumes giving a head of not more than 5 cm. The column parameters, column void volume (V_{c}) and column imbibed volume (V_i) were determined by measuring the volume of elumnt required to move peak concentrations of, respectively, X -globulin (Mol. wt. 150 000. Armour Pharma coutical Company, Ltd., Eastbourne) and KGl from the top to the bottom of the column bed. Samples were layered on the bed surface with a pipette after the removal of supermatant buffer, whereupon the escape tap was opened and the sample was allowed to enter the bed. Immediately, after the disappearance of the sample through the bed surface, a small volume of buffer was added to wash the surface and when this, in turn, had entered the bed a larger volume of buffer was added and the elution started. In ensyme frectionation experiments, eluste fractions of 2.0 ml. were collected with a flow rate of 15 - 25 ml. per hour. All experiments involving active enzymes were carried out at a temperature of between 0° and 4° with columns previously equilibrated with ice-cold eluant buffer. The columns were regenerated after use by washing with 10 column volumes of eluant buffer.

In spite of the precentions taken, the column beds were frequently observed to contract over a period of weeks and to allow for this change in bed volume, V_0 was routinely determined prior to each fractionation experiment. The columns were discarded when the minimum flow rate fell below 10 ml. per hour in which case interference from diffusion effects could be expected to effect markedly the reproducibility of the fractionations. In some enzyme experiments, a relatively large amount of THP was added as a stabilising factor for THP kinase. Assays reflecting absolute values of the THP kinase activity could, therefore, not be performed directly due to the dilution of (³²P) THP resulting from the presence of the unlabelled nucleotide.

In experiments where TMP was added to enzyme preparations in order to protect TMP kinese, the TMP could subsequently be removed by treatment with Sephedex G-25. Sephedex G-25 (Genree grade) was pretreated with 0.01 <u>M</u> phosphete buffer, pH 7.6, in the manner described above and, after coullibration, was packed carefully into a column (10 x 1 cm., dia.) observing the usual precautions. The column constants, V_0 and V_1 , were determined by the use of sheep blood haemoglobin and KGL, respectively, at a flow rate of 50 mL per hour. The sample from enzyme solutions containing TMP was layered on the bed surface as described above and the protein fraction appearing at V_0 after the start of the elution was collected and used for the assay of TMP kinese. The dilution of the enzyme solution resulting from this treatment was observed to be of the order of 1 : 3. The column was regenerated after use by washing with 5 column volumes of eluent buffer.

Several preliminary experiments had shown that TMP had a relatively high distribution coefficient (Kg) under the conditions described above. This was contrary to the observations of Flodin (1962) who found that acidic nucleotides were almost completely excluded from the gel phase when eluted with distilled water or very dilute buffers, an effect which had been attributed to the mutual repulsion of fixed negative charges. No such

- 130 -

effect was found in the present experiments even when distilled water was used as eluant. In experiments with columns of Sephader G-25 using 0.01 \underline{M} phosphate buffer as eluant, the K_d for TMP was shown to be approximately equal to that of KGl (<u>i.g.</u>, about 1.0).

(d) <u>Easyme fractions tions using ammonium sulphate</u>

Fractionation of enzyme solutions with (NH4)250, was occasionally performed in which case the experimental procedure was as follows: - ensyme solution (10 ml.), prepared by dissolving lyophylised crude ascites extract in 0.05 M phosphate buffer of the appropriate pH to give a protein concentration of 8 - 10 mg. per ml., was transferred to a precooled 50 ml. centrifuge tube and all subsequent operations were carried out at 0°. To this solution was added, with continuous stirring, the volume of saturated ammonium sulphate required to give 30 per cent ammonium sulphate saturation. The precipitate so formed was allowed to settle for 30 minutes at 0°. After the precipitation was complete, the mixture was contrifuged at 1000 x g for 15 minutes and the supermatant fluid transferred to a clean, precooled 50 ml. centrifuge tube. The sediment was redissolved in 10 ml. of 0.01 M tris buffer, pH 7.7. To the supermatant fluid were added further volumes of saturated ammonium sulphate and the precipitates obtained at 40, 50 and 60 per cent ammonium sulphate saturation recovered and dissolved in 10 ml. of 0.01 M tris buffer. The annonium sulphate fractions in this diluted form were assayed for the appropriate kinase activity without further pretreatment.

(.) Ensyme fractionations using ecotone

Acetone fractionation of enzyme solutions was sometimes employed as an initial operation in the purification of the thymidine and thymidylate kinases. In such experiments, the lyophylised orule asoites extract was dissolved in 0.01 M phosphate buffer of the appropriate pH to give a solution containing 8 - 10 mg. protein per ml. Strict temperature control was required in all experiments using acetone; thus, the fractions described below were performed in an ettylene glycol bath maintained at -15°. The ensyme solution (10 ml.) was transferred to a precooled 50 ml. centrifuge tube to which was added 1 ml. of acetone, previously chilled to -15°, to prevent freezing of the enzyme solution when introduced into the glycol bath. The tube was placed in the bath at -15° and 4 ml. of a cetone was added with stirring, thereby giving a solution containing 30 per cent acetone. The resulting mixture was centrifuged at 1000 x g for 15 minutes at -15° and the supernatant fluid was transferred to a clean, precooled 15 ml. centrifuge tube for further fraction-The precipitate, if any, was immediately redissolved in ice-cold stion. 0.01 M phosphate buffer, pH 7.8, is order to reduce the acetone concentration to a relatively harmless proportion. The precipitates obtained similarly at 40, 50 and 60 per cont a cotone concentration were preserved in the same way. Assays for the appropriate kinases were performed without further pretreatment of the diluted acetone fractions.

- 132 -

3.3.1. Biological factors affecting the activity of the thymidine and thymidylate kinese system from cell-free extracts of Landschutz assites tumour cells

An original observation by Gray <u>et al.</u> (1960) that the formation of TTP from TdR by cell-free extracts of ascites cells was seriously inhibited by addition of extracts of non-proliferating manualian tissues merited reinvestigation. The experiments of Gray <u>et al.</u> (1960) had indicated, in particular, that the activity of the thymidine and thymidylate kinase system from extracts of Shrlich ascites tumour cells was cancelled in the presence of extracts of rat liver. It seemed of interest to re-examine this phenomenon with the use of the relatively powerful tool provided by the kinase essay described earlier (Section 2.2.4 (a)). Such experiments sought to reveal the nature of the inhibitory effect caused by the eddition of liver extract to the ascites extract and, specifically, whether the inhibition was manifest at any single stage of the reaction sequence leading from TdR to TTP.

Experiments in which the phosphorylation of (³H) TdR was investigated under the influence of cell-free extracts of normal rat liver (Section 3.2.2 (b)) have invariably shown that the activity of the thymidine and thymidylate kinases in such extracts is very low when compared with cell-free extracts of Landschutz ascites carcinoma. Thus only 5 per cent of the total radioactivity was recovered as phosphorylated thymidine derivatives after a one hour incubation period while, in comparable experiments, ascites extracts catalysed the phosphorylation of over 80 per cent of the (³H) TdR. Moreover,
such experiments indicated that the phosphorylation of $({}^{3}H)$ TdR was restricted to the formation of TMP from TdR when rat liver extracts were employed as the source of enzymes, and no evidence was obtained for the production of TDP or TTP. Subsequent experiments in which $({}^{3}H)$ TdR was replaced as substrate by $({}^{32}P)$ TMP have revealed that the liver extracts were incompable of phosphorylating TMP under experimental conditions which allowed phosphorylation of 60 per cent of the TMP in the presence of escites extracts.

In another type of experiment, rat liver extract was added to ascites extract and the effect of such mixtures on the phosphorylation of (3H) TdR was studied as a function of the ratio of the concentration of liver protein to ascites protein. Such experiments (Table 4) have shown that the presence of liver extract causes pronounced diminution of the net phosphorylating activity of the ascites extracts, and that the degree of inhibition so produced was dependent on the concentration of liver protein in the reaction mixtures. Thus, over an incubation period of one hour, the proportion of the total radioactivity recovered in (TMP + TDP + TTP) was reduced from 65 per cent to 30 per cent in response to the addition of increasing concentrations of liver protein to reaction mixtures containing a constant concentration of ascites protein. The ratio of liver protein to ascites protein thus added varied from 1 : 4 to 2 : 1. The percentage of the total radiomotivity found in all phosphorylated derivatives of thymidine was 65 per cent when the incubations were performed in presence of ascites tumour extracts only. By inspection of the results recorded in Table 4, it can be seen that the

11ver (3.7mg. + 7.4mg.)	Complete + macitos + liver (5.7mg. + 3.7mg.)	Complete + sacites + liver (3.7mg. + 1.8mg.)	Complete + apoites + liver (3.7mg. + 0.9mg	Complete + liver extract (5.7 mg.)	Complete + ascites extract (3.7 mg.)	Complete - enzyme		Reaction mixture
99	5	79	.) 86	56	85	95		recovered
72	40	32	26	95	15	100	TdR	Per cent thymidine
25	16	27	32	J	10	1	THUP,	total red nucleoti
ų	16	12	17	1	13	1	TDP	loactivit .des
1		16	25		62	•	TTP	y in

Table 4. The effect of adding cell-free extracts of normal rat liver on the phosphorylation of (³H) TdR by cell-free extracts of Landschutz accites tumour cells.

> The reaction mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0, 0.3 µmoles 3-merceptosthemol, 15 µmoles Mg31₂, 15 µmoles ATP, 3 µc (³H) TdR and a volume of enzyme solution corresponding to 3.7 mg. of protein, the enzyme solutions being obtained either from cell-free extracts of normal rat liver (Section 3.2.2 (b)) or by redissolving lyophylised cellfree extracts of Landachutz escites carcinoms. In inhibitio experiments, the remotion mixtures contained 3.7 mg. protein from the assites extracts to which was added volumes of rat liver extracts corresponding to the quantities of protein recorded in column 2 of the table. Incudetion was for 60 minutes in a chaking water-bath at 37° and further treetment was as given in the text (Section 2.2.4 (a)).

percentage of inhibition of the thymidine and thymidylate kinases, as measured by the proportion of the total radioactivity recovered as (THP + TDP + TTP), appears to vary as a linear function of the liver protein concentration.

It was of some interest also to determine whether a similar situation obtained in the phosphorylation of other deoxyribonucleosides. As the additional substrate in such comparative experiments, deoxy cytidine (OdR) was shown to be phosphorylated at only one-third the rate of TdR when the source of enzymes was the cell-free extracts of Landachutz ascites tumour This was in contrast to the observations of Gray et al. (1960) and celle. Manstevinos and Genellakis (1959) using Ehrlich ascites cells and regnerating rat liver. Those studies were, however, performed with (32P) dGMP as substra and the observed discrepancy may thus be due to the rate limiting activity of the OdR kinese under the present experimental conditions. Similarly, the rate of phosphorylation of (3H) GdR, as measured by the proportion of the total radioactivity recovered as (dGMP + dODP + dOTP), by extracts from rat liver was lower than that observed by Gray et al. (1960, using (32p) dOMP as substrate thus indicating that the reaction catalysed by GdR kinese may be rate limiting in this tissue also. However, such extracts of normal rat liver were shown to phosphorylate 4 - 5 times as much (3H) OdR as (3H) TdR over a one hour incubation period.

Given these differences in the patterns of phosphorylation of TdR and OdR, it seemed of interest to determine to what extent the phosphorylation of OdR by ascites extracts would be affected by the addition of liver extract to the incubation mixtures. In experiments of this type, the proportion of the total radioactivity recovered as phosphorylated derivatives of $({}^{3}H)$ 3dR (see Section 3.2.3 (d)) and $({}^{3}H)$ 7dR were compared after incubation for one hour as recorded in Table 5. Again, addition of extracts of normal rat liver to extracts of Landschutz assites tumour cells reduced the thymidine and thymidylate kinase solivity by about 40 per cent when the ratio of liver protein to escites protein wes 1 : 1. In contract, the addition of liver extract to escites tumour extracts had no effect on the phosphorylation of $({}^{3}H)$ 3dR; indeed, no inhibitory effect could be observed on the rate of formation of decaycytidine nucleotides even when the liver protein to escites protein ratio was increased to 3 : 1 while the same treatment all but abolished thymidine kinase activity. It access possible, therefore, that the liver extracts contain a factor specifically concerned with the inhibition of formation of thymidine polyphosphates.

The observed behaviour of the system thus coincides with the pattern expected if the inhibitory factor operates as a competitive inhibitor for the phosphorylating system and, if taken together with the observations of Gray et al. (1960) that the inhibitory factor of rat liver was non-dialyzable and was rendered ineffective in boiled liver extracts, seems to suggest the existence of dephosphorylating enzymes in the liver extract. If the inhibition of the phosphorylation of thymidine stems from this source, the observed effect could be accounted for by the presence in the liver extracts of a series of relatively specific phosphateses sufficiently active to abolish net synthesis of thymidine nucleotides. The relatively high level

- 136 -

				(³ H) TdR.				(⁵ H) CdR.		Reaction
	Complete + asoites + 11 (3.7mg. + 3.	Complete + liver (3.7mg	Complete + ascites (3.7	Complete - enzyme	Complete + aseites + 11 (3.7mg.+ 3.7	Complete + liver (3.7mg	Complete + ascites (3.7	Complete - engyme		mixture
-	7mg.) 92	93	Mg.) 98	95	lver mg.) 96	() 91	mg.) 98	66	vity re- cevered	Percentage of total radioacti-
					45	08	35	100	UdR	Р
					32	16	44	1	TUNP	or cent
-					10	N	9	1	aupp	total
-					IJ	N	12	1	aurp	radioac icleotid
	52	56	17	100				12	TdR	tivity
	32	7	12	•					THOP	Tecover
	10	•	15	•					TDP	ed in
-	6	•	56	1					TTP	

Table 5

- Table 5. Kinnse activities in extracts of Landschutz ascites tumour cells and of normal ret liver tissues: phosphorylation of (³H) TdR and (³H) 5dR.
 - The reaction mixtures contained the following components in a total volume of 3.0 ml.: 300 µmoles tris buffer, pH 8.0. 0.3 µmoles 2-merceptoethanol, 15 µmoles MgGlg, 15 µmoles ATP, 3 µ0 (³H) TdR of specific activity 360 mg per mole or 3 µc (3H) GdR of specific activity 1100 mc per mmole and enzyme solution taken from one of the following: redissolved, lyophylised cell-free extracts of normal rat liver (3.7 mg. protein); redissolved, Lyophylised extracts of Landschutz ascites tumour cells (3.7 mg.protein); a mixture of both types of extract containing equal proportions of liver protein and escites protein (3.7 mg. : 3.7 mg. protein) or containing liver protein and ascites protein in the proportions 3 : 1 (11.1 mg. : 3.7 mg. protein). Incubation was for 60 minutes in a shaking water-bath at 37°. for reaction mixtures containing (³H) TdR, further treatment was as given in the text (Section 2.2.4 (a)). Reaction mixtures containing ("H) GdR were deproteinised and neutralised in the usual maner to give solutions of total volume 7.0 ml., 5.0 ml. of which were submitted to desminstion with nitrous acid as described in the text (Section 3.2.3 (d)). The resulting deoxyuridine derivatives were separated on columns of STROLA cellulose and the results were expressed

Table 5. as the percentage of the total redicectivity recovered in (contd.): UdR. dUNP, dUDP and dUTP and taken as equivalent to the

respective decayoytidine derivatives produced by the enzymic phosphorylation.

of thymidine nucleotidese activity in extracts of normal rat liver has previously been noted (Bollum, 1958; Manstavinos and Ganellakis, 1959) and implicated as a reason for the failure of such extracts to form TTP from TdR and TMP.

The present experiments, however, did not cerry quantitative suthority and the interpretations given above must, as a result, remain somewhat speculative. In particular, no firm evidence has been recorded with a direct bearing on the mechanism of formation of TTP from TdR, neither has it been established whether the inhibition caused by the liver extract is specifically directed towards any single reaction in this process. Present evidence tends to indicate that all the relevant reactions are so inhibited, the phosphorylation of TMP being possibly more seriously effected than the thymidine kinase sotivity.

3.3.8. Experiments with the thymidine kinese from cell-free extracts of Landschutz ascites tumour cells

The evidence so far collected (see above and Section 2.3) demonstrated clearly that the thymidine and thymidylate kinases of Landschutz secites tumour cells were espable of phosphorylating thymidine to TTP and while experiments with crude ensyme extracts sufficed to reveal the overall pattern of this process, it became apparent that precise elucidation was hampered by the lack of information on the individual enzyme reactions.

Thus several types of experiment were performed in order to characterise and, if possible, resolve the separate kinase activities co-operating in the phosphorylation of thymidine to TTP; such studies being initially concentrated on the reaction catalysed by thymidine kinase (TdR kinase). Earlier studies (see Section 2.3.2.) had indicated that the formation of TMP from TdR was catalysed relatively afficiently by the orude cell-free extracts of Landschutz ascites tumour cells; thus when (3 H) TdR was used as substrate, over 80 per cent of the total radionativity was recovered as phosphorylated thymidine derivatives (TMP + TDP + TTP) after incubation for 60 minutes with the orude ascites extracts as enzyme source (see Fig. 22). By using the total radionativity recovered in (TMP + TDP + TTP) as the ariterion for measurement of TdR kinase activity, relatively reliable values could be obtained as the system was thus internally compensated for lowers of (3 H) -label (Goutier, 1964) by thymidine catabolism. This criterion was also the bhasis of a means of expressing the specific activity of TdR kinase (for definition of a unit of TdR kinase activity, see legend to Fig. 31).

Preliminary experiments were performed in order to formulate the optime requirements of the TGR kinese; the parameters being determined by the use of material from the hypphylised pool of cell-free extracts of Landschutz escites cercinome (see Section 3.2.2. (a)).

Experiments exploring the effect of pH on the action of TdR kinese showed (Fig. 31 (3)) that the reaction exhibited a relatively marrow pH optimum giving maximal activity between pH 7.8 and pH 8.1 when in a reaction medium containing 5 x 10^{-3} <u>M</u> MgGlg and 5 x 10^{-3} <u>M</u> ATP. The ATP and Mg⁺⁺ requirements of the reaction were also determined and the results indicated that the reaction catalyzed by TdR kinese operated optimally in the presence



Fig. 31. The motion of TdR kinese from extracts of Landschutz escites tumour cells as measured by the total amount of (³H) TdR phosphorylated.

- (A) Time course of total phosphorylation of ("H) TdR.
- The reaction mixtures contained the following components in total volume of 1.0 ml.:- 100 umoles tris buffer. pH 8.0 0.1, unoles 2-merceptoethenol, 5 umoles MgClo, 5 umoles AT 1 µc (³H) TdR of specific activity 360 mc per mole and enzyme solution (derived from lyophylised cell-free extre of ascites tumour cells) containing 1.0 mgs protein. Th mixtures were incubated for the times indicated in a shaking water-bath at 37°. Further treatment was as described in the text (Section 3.8.3 (a)). Freetionetic on columns of MCTHOIA cellulose yielded two fractions, on containing (3H) TdR and the other containing (3H) -labell (TMP + TDP + TTP). Results were obtained as the percentage of the total radioactivity recovered in (THP + TDP + TTP) from which the number of mumoles of TdR phosphorylated were calculated as follows:

2.78 x X Y x K : mumoles TdR phosphorylated

where the numerical factor represents the emount of TdR present initially (mumoles) K is counts per minute recovered as (TMP + TDP + TDP), Y is the fraction of total initial redicactivity recovered and K is the total redicFig. 31. activity of the TdR present initially. One unit of TdR (contd.): kinese activity was defined as that which phosphorylates

1 mumole TdR in 15 minutes.

(B) The effect of protein concentrations on the total phosphorylation of (³H) TdR.

The conditions were as described above except that the reaction mixtures contained protein in the quantities given on the diagram. The incubation time was 15 minute

(C) The effect of pH on the total phosphorylation of (³H) TdR.

The conditions were as described above except that the reaction mixtures contained 100 µmoles tris buffer of the pH indicated on the diagram and 1.0 mg. protein. The incubation time was 15 minutes. of ATP : Mg^{**} concentrations in the ratio 1 : 1. The reaction thus exhibited a set of requirements closely resembling those previously shown to ensure optimal efficiency of production of TTP from TdR (Section 8.3.1), the resemblance of the two systems being further emphasised by the marginal stimulation of TdR kinase activity resulting from the presence of 0.1 x 10^{-3} M 2-merosphoethanol. No stimulation was observed in the presence of bovine serum albumin in contrast to the findings of Okazaki and Kornberg (1964a).

The time course of TdR kinnse sotion (Fig. 31 (A)) was shown to be a linear function of the incubation time from time zero up to about 30 minutes after which time reaction product inhibition became manifest. The specific activity of the TdR kinnse under these conditions was about 0,006 mumoles TdR phosphorylated per minute per mg. protein over the initial 15 minutes of the remotion, more than 60 per cent of the total amount of TdR being recovered in the form of phosphorylated derivatives at the end of a 60 minutes' incubation period. The initial remotion rate was routinely measured prior to any new series of experiments in order to allow for fluctuations in the TdR kinese activity due to stornge of the lyophylised cell-free ascites extract pool. The specific activity of TdR kinese did not vary significantly in material drawn from this pool over a period of 12 months, values of 0.05 - 0.07 mumoles TdR phosphorylated per minute per mg. protein being representative.

Given the set of optimal requirements for TdR kinnse action described above, it was important to ascertain that the assay used for the determination of TdR kinnse activity (Section 3.2.3 (a)) was valid under the conditions

- 139 -

chosen. The results shown in Fig. 31 (B) reveal that the phosphorylation of TdR was a linear function of the protein concentration over the selected range and thus indicate that the enzyme remained fully saturated with substrat

A report in the work of Weissman, Smellie and Paul (1960) that treatment of crude Shrlich agoites extracts with dilute acid led to an enrichment of the TdR kinese of that tissue was further investigated. Experiments in which the crude ascites extracts were fractionated with dilute agetic acid have shown that a major proportion of the TdR kinase activity remained in the supernatant fluid when the pH was lowered to 4.5. The proportion of the tota initial TdR kinase activity so recovered was found to be dependent on a number of factors, chief among which were a strict control of pH and the presence of stabilising factors. Thus, while 80 per cent of the total TdR kinese activity was frequently lost when the Lyophylised crude extract preparation was dissolved in distilled water, experiments performed in the presence of 0.01 M phosphate buffers, pH 7.5 - 7.8, resulted in recoveries of 60 - 80 per cent of the total TdR kinese activity. One such experiment (see Section 3.2.4 (a)), performed in 0.01 M phosphate buffer, pH 7.8, is recorded in Table 6 from which it is seen that over-lapping fractionation of the enzyme solution results in an enrichment of the TdR kinase setivity, 95 per cent of the total activity being recovered in the fraction remaining soluble at pH 4.5 (AS 4.5 fraction). The material so obtained contained TdR kinese purifier 3-fold over the original extract. Moreover, examination of the recovery values makes it clear that the total measurable enzyme activity in the fraction recovered at pH 4.5 exceeds that present initially, thus indicating perhaps a

- 140 -

AR 4.5 AS 5.2 AR 5.5 AS 5.5 AR 5.2 SV AR AS Control Fraction 5.0 5.0 4-5 2.87 7.75 3.40 1.47 4.95 1.90 1.43 8.20 3.63 Units per ml. total no. unite 123-1 111.6 74.2 22.00 43-13 54.5 21.41 51.700 tion concentra-Protein mg. per ml. 4.16 3.2 3.12 3.52 2.95 2.95 2.52 4-48 7.3 Incovery of protein Per cent 100 43 61 63 35 40 48 Per cent recovery activity 100 45 45 42 61 35 3 18 units per Botivity Specific 0.81 0.69 0.45 1.13 1.09 0.47 1.69 0.47 5.05 tion Pur1--1.0 1 5 0.7

Table 6.

- Table 6. The effect of fractionation with dilute acetic acid on TdR kinase of cell-free extracts of Landschutz ascites tumour cells.
 - The fractionation experiments were performed as described in Section 3.2.4 (a). Enzyme solution was prepared by dissolving lyophylised cell-free extracts of Landschutz ascites tumour cells in 0.01 M phosphete buffer, pH 7.8. 15 ml. portions being transferred to centrifuge tubes. Rach portion was then adjusted to either pH 5.5, 5.2, 5.0 or 4.5 with 1.0 blecetic acid, centrifuged to give the appropriate acid-soluble (AS) fraction or acid-insoluble (AR fraction and the pH of each individual fraction readjusted to 7.8 with 1.0 N sodium hydroxide. Samples were taken for measurement of TdR kinese activity using the radioectivity easey (Section 3.2.3 (a)) with (H) TdR as substrate. One unit of TdR kinese activity was defined as that which is capable of phosphorylating 1 mumole TdR in 15 minutes (colculated from the formula given in the legend to Fig. 31). Specific TdR kinase activity was given as units per mg. protein.

differential precipitation of a TdR kinese inhibitor whenever the pH is lowered beyond 5.0. The phenomenon may be due to the removal of dephosphorylating enzymes or to the unblocking of represend enzyme molecules, both of which events would be expected to result in the gradual unmasking of TdR kinese activity. However, it seemed alear also that some destruction of enzyme activity occurred in the acid medium, only 60 per cent of the TdR kinese activity being recovered when the precipitation was performed at pH 5.5 - 5.0. A slight improvement in the recovery values was observed when the fractionation was performed in the presence of 0.8 x 10^{-3} M 2-merceptoethenol or 0.1 M KG1 while the presence of NaG1 appeared to occuse some destruction of TdR kinese activity.

While these results largely confirmed the observations of Weissman, Smellie and Paul (1960), other experiments indicated that the TGR kinese of Landschutz ascites tumour cells behaved differently from the Ehrlich ascites enzyme. Thus it was shown that the TGR kinese did not respond favourably to fractionation with annonium sulphate (Table 7) under the conditions employed by Weissman, Smellie and Paul (1960). Best recoveries were obtained under conditions where the pH of the medium and of the saturated annonium sulphate solution was strictly controlled, the most favourable pH being 7.8. In such fractionations with the crude ascites extracts, a 8-fold purification of the TdR kinase activity was achieved in the fraction precipitated between zero and 30 per cent annonium sulphate saturation. Unfortunately, this represented no more that 20 per cent of the total kinase activity, the

- 141 -

Table 7 (B).

11 11	0	0	I	0.05	0	0	Supernat- ant fluid
1	0.415		0	1.26	2.62	0.523	60 - 75
	0. 42	¥	66	5.45	24.7	2.47	52.5 - 60
	0.369	7	17	1.39	5.12	0.512	45 - 52.5
1.2	1.042	4		0.72	5.0	0.75	0 - 45
I	0.907	100	100	8-1	735	7.35	Control
Puri- fiom- tion	Specific activity unite per mg.	Per cent recovery of notivity	Per cent recovery of protein	Protein concentra- tion mg. per ml.	Total no. of units	Units per ml.	Per cent noctone (V/V)

6	
0	
Jul I	
9	
100	
-1	
-	
-	
-	
•	
-	

Supernat- ant fluid	50 - 60	40 - 50	30 - 40	0 - 30	Control	per cent nulphate saturation
0.523	0.535	1.405	0.888	2.01	5.44	nite per ml.
26.15	5.35	14.05	8.88	20.1	108.8	Total no. of units
0.84	0.96	3.3	1.55	1.34	6.00	Protein concentra- tion mg. per ml.
35	8	27.5	12	ш	100	Per cent recovery of protein
24	5	13	00	19	100	Per cent recovery of activity
0.623	0.777	0.426	0.574	1.496	0.901	Specific activity units per
				1.6	ч	Puri- fion- tion

- Table 7. Effects of ammonium sulphate and acetone fractionations on TdR kinase of cell-free extracts of Landschutz escites tumour cells.
 - (A) Ammonium sulphate fractionation at pH 7.6.

Fractionation was performed as described in Section 3.8.4 (d) with an ensyme solution prepared by dissolving the lysphylised cell-free extracts of Landschutz ascites tumour cells in 0.05 M phosphete buffer, pH 7.6. Fractions precipitated following the addition of saturated annonium sulphate (preadjusted to pH 7.6) were redissolved in 0.01 M tris buffer, pH 7.7. Samples were taken for measurement of TdR kinase activity using the redioactivity essay (Section 3.2.3 (a)) with (³M) TdR as substrate. One unit of TdR kinase activity was defined as that capable of phosphorylati: 1 mimole of TdR in 15 minutes.

(B) Agetone fractionation at pH 7.8.

Frectionstion was performed as described in Section 3.2.4 (e). Enzyme solution was prepared by dissolving lyophylised cellfree extracts of Landschutz escites tumour cells in 0.01 <u>N</u> phosphate buffer, pH 7.8, and fractions precipitated by soutcome treatment were dissolved in the same buffer. Measurement of TdR kinase activity was performed as above. fractionation of the AS 4.5 fraction, similarly, did not result in significant enrichment in the TdR kinnse activity in contrast to the observations of Weisemen, Smellie and Paul (1960). Fractionation with acetone under condition strictly controlled with respect to temperature and pH (see Section 3.2.4 (e), as shown in Table 7 resulted in marginal enrichment of TdR kinnse activity only at the expense of the recovery of total enzyme activity.

While the present experiments did no more than demonstrate the presence of TAR kinnse activity in the crude ascites extracts, some additional studies (described in the following Sections) have clearly shown the separate existence of this kinnse which, by virtue of its relatively greater stability, my readily be separated from the thymidine nucleotide kinnses.

3.3.3. Stability of the thymidize and thymidylate kinases of cell-free extracts of Landschutz ascites tumour cells

Several types of preliminary experiment had demonstrated that the thymidine and thymidylate kinases were highly unstable in equeous solution and precautions invariably had to be taken in order to reduce treatments in equeous media to a minimum. Thus, the crude assites extracts retained their kinase activities for a few hours only and extracts aged for 13 - 24 hours at 0[°] frequently lost all detectable TMP kinase and TDP kinase activities while retaining 40 per cent only of their TdR kinase activity. Similarly, frozen extracts stored for 24 hours at -10° lost 40 per cent of their TdR kinase solivity. However, it was found that the thymidine and thymidylate kinase activities from such extracts could be preserved by lyophylisetion and

subsequent storage of the dried material at -50°. In this form, the ThP and TDP kinases retained their activities for a few months while the TdR kinese activity remained unimpaired after storage for 12 months. In general. therefore, the crude cell-free escites extracts were lyophylised immediately after preparation unless specifically required for enzyme experiments in which case incubation was carried out within 1 - 8 hours of the preparation. In the course of enzyme purification also, kinase activities were preserved by lyophylisation whenever appropriate. The instability of the thymidine and thymidylate kinases from such extracts was further illustrated by their sensitivity to dislysis. The fractionstion experiments involving the use of saturated ammonium sulphate and acetone, it was particularly desirable to have access to procedures for the removal of excess selt or organic solvent (see Tables 7 and 10). However, several experiments performed in both the presence of and the absence of high anmonium sulphate concentrations revealed that the kinase activities were quickly destroyed when dialysis was performed against distilled water at 0°. In the most favourable circumstances, when dialysis was performed with the use of thoroughly washed and boiled dialysis tubing sgainst three changes of 0.01 M tris buffer, pH 7.8, in 0.1 x 10-3 M 2-merosptoethenol at 0°, 30 per cent only of the total TdR kinese activity was recovered after 18 hours in contrast to the experience of Bojarski and Hiatt (1950) and Weissman, Smellie and Paul (1960) while TMP and TDP kinese activities were all but completely destroyed by this treatment. The instability of the thymidylate kinases to dialysis has also been observed by Ives, Morse and Potter (1963). However, both TdR and TMP kinases appeared

- 143 -

to be unaffected by the inclusion of low concentrations of ammonium sulphate in enzyme experiments (Tables 7 and 8).

It would appear, therefore, that the thymidine and thymidylate kinase of the crude ascites extracts exhibited variable relative stability under different conditions, the TdR kinase being relatively more stable to elmost all types of treatment then were the THP and TDP kineses. This conclusion was further substantiated in experiments designed to study the response of the different kinases to treatment with dilute acid. In the fractionation experiments described earlier (Section 3.3.2.), it was observed that a proportion of the TdR kinese activity was lost in the course of treatment with dilute acetic acid, the recovery of kinase activity being dependent on the duration of the period in which the enzyme was in direct contact with dilute sold. Satisfactory recovery of kinese activities by this type of treatment was therefore only schieved when the period in acid solution was reduced to a minimum and the acid soluble fractions so obtained were immediate! neutralised with dilute alkali (see Section 3.2.4 (a)). Another interesting observation made in the course of such preliminary experiments was that the THP and TDP kinases appeared to lose their activities more rapidly in the presence of dilute acetic acid than did TdR kinese. Thus it seemed conceivabl that this apparent differential stability could be utilized as a means of separating the relevant kinage activities and experiments were initiated to study the fate of kinase activities as a function of the duration of the acid treatment. In such experiments (Fig. 32), the percentage of the total TdR, TWP and TDP kinage activities recovered in the fraction remaining soluble at

Table 8.

Source of enzymes	Reaction mixture	Concentration of additions	THP kinase specific actvity units per mg. protein
Crude soluble		x 10 ⁵ H	
ascites tumour	Complete	and a start	1.93
Gella.	+ EtSH	0.1	1.87
	+ EtSH	0.5	1.72
	+ EDTA	0005	1.75
	+ EDTA	0.40	1.85
	+ (MH4)25	04 2.0	1.86
	+ (HH ₄)2 ⁵⁰ 4	10.0	1.93
AS 4.7 Prestic	n Complete		6.22
Care Care and	+ EtSH	0.05	5.92
	+ Etsh	0215	4.53
	+ 5489	0.5	3.21
	+ EDTA	0.05	6.52
	+ EDZA	0.4	6.293
	+ (Mi4)2804	2.0	6.36

- Table 8. Effect of various ions on the phosphorylation of (³²P) TMP by cell-free extracts of Landschutz escites tumour cells and by the A3 4.7 Fraction.
 - TMP kinase activity was measured by the radioactivity essay (Section 3.2.3 (b)) with (SP) THP as substrate. The reaction mixtures contained the following components in a total volume of 1.0 ml.: - 100 µmoles tris buffer, pH 7.6. 5 µmoles ATP, 5 µmoles MgCl2, 20 mumoles (P) TMP of specific activity 5 x 10° counts per minute per umole. 1.0 mg. protein from enzyme solutions prepared from the sources given in the Table in 0.01 M phosphete buffer, pH 7.6, and 2-merceptoethanol (EtSH), EDTA or (NH4)2304 in the concentrations indicated. In cubation was for 25 minute at 37°. One unit of THP kinese ectivity was defined as that which catalysed the phosphorylation of 1 mumole THP in 85 minutes (celculated from the formula given in the legend to Fig. 33) and the specific ensyme activity was defined as units per mg. protein.



Fig. 32.

Fig. 32. The effect of treatment with dilute scotic acid on the activities of TdR kinese, TMP kinese and TDP kinese of cell-free extracts of Landschutz ascites tunour cells.

> Two enzyme solutions were prepared by dissolving lyophylised cell-free extracts of Landschuts ascites tumour cells in 0.01 M phosphate buffer, pH 7.6. Both solutions contained 815 mg. protein per ml. To one solution was added 80 mumoles of TWP per 1.5 mg. protein whereupon both solution ware added 1.0 H mostic moid until pH 4.5 was attained. Samples of the solutions were either removed immediately, centrifuged and the pH of the supermetant fluids readjuste to 7.6, or allowed to stand at 0° for the times indicated on the diagram before being similarly neutralised. The details of the acid treatment were those given in Section 3.2.4 (a). Enzyme activities were determined as indionted; TdR kinnee (Section 3.2.3 (a)), THP kinnee (Section 3.2.3 (b)) and TDP kinase (Section 3.2.3 (b)) and the results expressed as the percentage of total initi enzyme activity recovered in the neutralised supernatant fractions. For the solution containing added TMP, only the curve for TMP kinase is recorded, no change being observed in the recovery of the TdR or TDP kineses.

pH 4.5 was determined at different time intervals after precipitation with 1.0 N acetic acid, the time intervals being measured as the time elepsing between the initial attainment of pH 4.5 and the readjustment of the fractions to pH 7.8. The results showed that the ability of such solutions to catalyse the formation of TTP from TdR was progressively lost over a 30 minute period of soid trestment, the TMP and TDP kinase activities being lost at a more rapid rate than was the TdR kinase activity. Indeed, enzyme solutions exposed to eaid treatments extending beyond 30 minutes still reteined their ability to phosphorylate TdR while all measurable activity on TMP and TDP had been destroyed, the TdR kinese preparation so obtained rateining some 60 per cent of the total initial TdR kinase activity. A preparation with TdR kinase alone could, therefore, be conveniently obtained by judicious timing of the duration of soid treatment and provided strong evidence for the separate existence of the TdR kinase in the cell-free extracts of Landschutz ascites orginome. Although experimentel details are not available, it seems probably that a similar procedure was applied to extracts of Novikoff hepatoms (Ives, Morse and Potter, 1963) in the preparation of a TdR kinase uncontaminated with thymidylate kinese activities.

Subsequent experiments in the present investigation have focussed ettention on the TMP kinese, the instability of which proved to be a major obstacle in its characterisation. It was shown, however, that addition of excess TMP to the ensyme solutions appeared to lend increased stability to the TMP kinese component of such solutions (see Section 3.3.4.). Thus, the addition of excess TMP to the crude ascites extracts prior to fractionation

- 145 -

with dilute acid resulted in the recovery of SO - 120 per cent of the TMP kinese activity in the supermatent fraction following acidification to pH 4.7 while much of the TDP kinese activity was destroyed. The degree of stabilisation afforded to TMP kinese by its substrate was nicely illustrated by experiments such as that shown in Fig. 32 from which it may be deduced that significant loss of TMP kinese activity did not occur within the initial 80 minutes of dilute acetic acid trantment at pH 4.5. The relative stability of the TMR and TDP kineses remained unaffected by the addition of excess TMP.

3.3.4. Experiments with the TMP kinese from cell-free extracts of Landschutz ascites tumour cells

The results of <u>in vitro</u> experiments on the mechanism of the formation of TTP (see Section 8.3.8.) provided strong circumstantial evidence in favour of the hypothesis propounded by deisemen, Smellie and Paul (1960) that the process proceeds by stepwise phosphorylation from TdR through TMP and TDP. In the view of Bianchi, Butler, Grathorn and Shooter (1961), formation of TTP from TMP occurs by pyrophosphorylation and the argument thus hinges on whether or not the formation of TTP from TdR requires the participation of a kinese specifically responsible for the phosphorylation of TMP to TDP. The remainder of the present investigation was devoted to the search for unequivocal proof of the existence of a TMP kinese in the cell-free extracts of Landschuts ascites percincema.

Rarlier experiments (Fig. 26) had indicated that the cell-free extracts were capable of phosphorylating TMP to TDP and TTP in a reaction where the

- 146 -

production of TDP appeared to be rate limiting. When (32p) TMP was used as substrate, over 50 per cent of the total redicactivity was recovered as phosphorylated derivatives (TDP + TTP), only 10 per cent of which was associated with the TDP, after incubation for 60 minutes with the enzymes of orude ascites tumour extracts. As in the case of TdR kinase activity (see Sections 3.2.3 (a) and 3.3.2.), it was found convenient to use the total redicectivity recovered in (TDP + TTP) with (32P) TMP as substrate as a criter for the measurement of THP kinese activity (Section 3.2.3 (b)), this criterios being also employed to express specific TMP kinass activity (for definition of a unit of TMP kinase activity, see legend to Fig. 33). While this type of essay could not distinguish between phosphorylating activity due to TMP kinese alone and that due to the combination of TMP kinage with TDP kinage, corroborative evidence on the relative activities of the two kinases was obtained by suitable correlation with paper chromatography as described elsewhere (Section 3.2.3 (b)).

The results recorded in Section 3.3.2. indicated that the requirements for optimal activity of the TdR kinese coincided with those promoting maximal formation of TTP from TdR when the reactions were entalysed by the crude extracts of Landschutz escites parcinoms. It was thus of interest to discove whether the optimal requirements for TMP kinese activity were similarly governed.

A number of preliminary experiments had borne out the fact that the rat of phosphorylation of TMP by the crude extracts was relatively low compared wit the ability of such extracts to phosphorylate TdR (Fig. 31) and TDP (Fig. 25). Enrichment of the TMP kinnse activity was, therefore, a necessary prelude to the determination of optimal requirements and the parameters sucted hereafter refer to the partially purified AS 4.7 Fraction (see Fig. 38 and Table 9) rather than to the crude cell extract.

Experiments on the time course of the phosphorylation of TMP with partially purified enzyme preparations have shown (Fig. 33 (A)) that the reaction proceeds as a linear function of the incubation period over time intervels from zero to 30 minutes when studied in a reaction medium containing 5 x 10-3 M MgGl, end 5 x 10-3 M ATP. With (32P) TMP-5' es substrate in a concentration of 2 x 10-5 M, the initial rate of phosphorylation as measured by the proportion of the total redicectivity appearing in (TDP + TTP), was of the order of 0.05 mumoles TMP phosphorylated per minute, the specific enzyme activity being most conveniently given as muncles TMP phosphorylated per 25 minutes per mg. of protein. A falling off in the reaction rate was observed with incubation periods extending beyond 30 minutes, presumably due to inhibition of the thymidylate kinases by accumulating reaction products combined with the effect of endogenous phospheteses. It was observed with interest that no reaction occurred under these conditions when (32P) TMP-3' replaced (32p) TMP-5' as substrate, the enzyme therefore exhibits specificity for the decxyribonucleoside 5'-monophosphete. In subsequent experiments it was shown (Fig. 33 (B)) that the rate of phosphorylation of TMP by the partially purified TMP kinese of the AS 4.7 Fraction was directly proportional to the protein concentration over the range 0.1 - 2.0 mg. per ml., thus showing that the esent used for determination of THP kinese activity was valid within this range.

Table
9
•

	AS 4.5	AS 4-7	AS 5.0	AS 5.2	AS 5-5	Control	Fraction
	23.45	36-11	15.92	17.81	21.61	28. 96	Units per ml.
	1117	1818	797	891	1079	1445	Total no. of units
and the second second	4.05	4.25	4.36	4.75	7.47	11.5	Protein concentra- tion mg. per ml.
	32	77	86	41	65	100	Per cent recovery of protein
	18	125	55	62	75	100	Per cent recovery of activity
	5.75	8.51	3.66	3.75	2.89	2.52	Specific activity units per mg.

- Table 9. The effect of fractionation with dilute soutic acid on TMP kinnes of the cell-free extracts of Landschutz ascites tumour cells in the presence of added TMP.
 - The details of the fractionation procedure were those described in Section 3.2.4 (a). Enzyme solution was prepared by dissolving Lyophylised cell-free crude extracts of Landschut ascites cells in 0.01 M phosphate buffer, pH 7.6, to which was added 80 mimoles of TMP per 1.5 mg. protein and the solution was divided into six equal portions. Each portion was then adjusted to either pH 5.5, 5.8, 5.0, 4.7 or 4.5 with 1.0 M acetic acid, centrifuged immediately and the pH of the supernatant fractions readjusted to 7.6 with 1.0 N sodium hydroxide. Samplas ware taken from each neutralised fraction and the TMP kinase activity measured using the redicactivity essay (Section 3.2.3 (b)) with (38 P) THP as substrate (after pretreatment on a column of Sephedex G-85 to remove added TMP; see Fig. 36). One unit of ensyme activity was defined as that capable of phosphorylating 1 mumole TMP in 25 minutes.



Fig. 33. The action of TMP kinase from extracts of Landschutz ascites tumour cells as measured by the total amount of (³²P) TMP phosphorylated.

> (A) The time course of the total phosphorylation of (³²P) TMP.

The reaction mixtures contained the components given in the text (Section 3.2.3 (b)), the (³²P) TMP having the specifit fext (Section 3.2.3 (b)), the (³²P) TMP having the specifit activity 5 x 10⁶ counts per minute per µmole. Each re- hylised escites tumour cell-free extracts and incubation was carried out for the times indicated on the diagram in a shaking water-bath at 37⁶. After the separation of reaction products, the results were obtained as the percentage of the total radioactivity recovered in (TDP + TTP from which the number of mumoles of TMP phosphorylated could be calculated as follows:

T x 2 (counts per minute) = mumoles TMP phosphoryleted

where X is the total number of counts per minute recovered in (TDP + TTP), Y is the fraction of the total initial redicectivity recovered and Z is the number of counts per minute corresponding initially to 1 mumole TMP. One unit of TMP kinese activity was defined as that which phosphory lates 1 mumole of TMP in 25 minutes.
Fig. 33 (Contd.): (B) The effect of enzyme concentration on the total phosphorylation of $(\frac{32}{P})$ TMP.

The remotion mixtures contained the components described in the text (Section 3.2.3 (b)) with enzyme concentrations as given on the diagram. The incubation time was 25 minutes.

In experiments designed to study the effect of pH on the phosphoryletic of TMP, it was shown (Fig. 34) that the reaction exhibited a particularly sharp pH optimum at pH 7.6 when catalysed by the AS 4.7 Fraction. The rate of the reaction declined repidly above pH 7.7 and below pH 7.5 so that when the phosphorylation was conducted in the presence of 0.1 M tris buffer. pH 7.3 or pH 8.0, the amount of (TDP + TTP) produced within 85 minutes was only one-half of that formed at pH 7.6. Thus it would appear that the pH optimum for the TMP kinase differs from those of the TdR kinase and of the over-all phosphorylation of TdR to TTP which are in the range pH 7.8 - 8.1. The significance of this observation remains uncertain, but it seems conceivable that the differentiation of pH optime so indicated may provide a partial explanation for the rate limiting effect of the phosphorylation of TMP in the formation of TTP as estalysed by the crude ascites extracts. It seems clear, however, that these optime represent the most fevourable balance between phosphorylation and dephosphorylation in the complex escites extract enzyme mixture and the pH optimm of highly purified kineses may prove to be different from those presented here.

If, as has been supposed, the true cosubstrate for the TMP kinase were the ATP - Mg⁺⁺ chelate, it would be of some interest to examine the effects of variation in the magnesium concentration on the phosphorylation of TMP by the partially purified AS 4.7 Fraction. In the experiment shown in Fig. 35, the ATP concentration was kept constant at 5 x 10^{-3} M, a concentration which appeared to be optimal for the TMP kinese, and the Mg⁺⁺ concentration was varied. While the results indicated that the net rate of TMP phosphorylation



Fig. 34. The effect of pH on the action of TMP kinase.

The resction mixtures contained the following components in a total volume of 1.0 ml.:- 5 µmoles ATP, 5 µmoles MgOl₂, 20 mµmoles (³²P) TMP of specific activity 3.5 x 10^6 counts per minute per µmole, 1.0 mg. protein and 100 µmoles tris buffer of the pH indicated on the diagram. Incubation was for 25 minutes in a sheking water-bath at 37^0 and further truetment was as described in the text (Section 3.2.3 (b)). The source of enzyme was the AS 4.7 Fraction (see Section 3.3.5; Methods I and II).



Fig. 35. The effect of Mg concentration on the motion of TMP kinnse.

The reaction mixtures contained the following components in a total volume of 1.0 ml.:- 5 µmoles ATP, 100 µmoles tris buffer, pH 7.6, 30 mµmoles (32 P) TMP of specific activity 3.5 x 10⁶ counts per minute per µmole, 1.0 mg. protein and Mg3lg in the quantities indicated on the diagram. Incubation was for 35 minutes in a sheking water-bath at 37⁰ and further treatment was as described in the text (Section 3.2.3 (b)). The source of enzyme was the AS 4.7 Fraction (See Section 3.3.5; Methods I and II). was optimal at 5 x 10^{-3} M MgGl₂, there were also indications that there occurred a second optimum between 15 x 10^{-3} and 20 x 10^{-3} M MgGl₂ concentration No TMP was phosphorylated in the absence of added MgGl₂, thus showing that any endogenous Mg⁺⁺ was insufficient to sustain the reaction, while the relatively slow decline in the TMP kinese activity with higher lovels of Mg⁺⁺ could be indicative of the availability of unchelated Mg⁺⁺ to nucleotide phospheteses. The double Mg⁺⁺ optimum which seems to be exhibited may be due to different Mg⁺⁺ requirements of the TMP and TDP kineses. In subsequent routine kinese assays, the ATP and Mg⁺⁺ concentrations were maintained in a ratio of 1 : 1.

The instability of the TMP kinese in equeous solution (see Section 3.3.3.) necessitated a search for possible stabilising factors. Factors were sought which could fulfil the two separate functions of minimizing denaturation of the enzyme during incubations at 37° and of leading added stability to the TMP kinese component during enzyme fractionations. 8-Merceptosthanol was previously (Section 8.3.1) shown to stimulate the overall phosphorylation of TdR to TTP by the arude macites extracts and seemed a possible candidate to not as stabilizing factor for the TMP kinese also. Experiments which investigated the effect of this compound, as well as some other ions, on the phosphorylation of TMP have shown (Table 8) that the TMP kinese derived from the arude lyophylised assites tunour extract was unaffected by the messance of 8-merceptostanol, MDTA and $1 \ge 10^{-2} \ge (ME_4)_2 \odot_4$. It was observed, however, that the rate of phosphorylation of TMP by the partially purified A3 4.7 Fraction declined markedly in the presence of

- 150 -

2-merceptosthenol, the specific ectivity of the TMP kinese being reduced to one-helf that of untreated preparations in the presence of 0.5 x 10^{-3} <u>M 2-merceptosthenol.</u> EDTA and $(NH_4)_2SO_4$ had no significant effect on the phosphorylation of TMP by the AS 4.7 Fraction. The addition of bovine serum albumin to the reaction mixtures, similarly, had no effect on the rate of phosphorylation of TMP even when the activity of highly purified kinese preparations was measured.

An original observation made by Hiatt and Bojarski (1960) that the administration of thymidine resulted in a stimulation of the thymidylate kinese activity of ret tissues suggested thymidine as a possible stabilising factor for TMP kinese. However, in the experiments of Histt and Bojarski (1960) thymidine was presumably converted to TMP within the cell thus suggesting that TMP was the true stimulatory agent. The present series of experiments have confirmed the validity of this assumption and have shown that addition of excess TMP to the TMP kinese derived from extracts of Landschutz ascites tumour cells results in increased kinase stability. This device has proved particularly valuable in the characterisation and purification of THP kinase as described in the following pages. Thus, the redissolved lyophylised, crude extract, which previously had been observed to lose its ability to phosphorylate TMP over a period of 2 - 3 hours, showed retention of TMP kinese activity up to 24 hours when dissolved in 0.01 M phosphate buffer, pH 7.6, containing 80 mumoles TMP per 1.5 mg. protein The protection of TMP kinese by its substrate raised several important questions as to the precise nature of this effect. Some knowledge of this

- 151 -

was necessary because it was clear that the assay employed to measure THP kinase activity became quantitatively invalid once applied to enzyme solutions containing high concentrations of TMP. It seemed probable that the association of TMP with TMP kinese under such dircumstances would represent a situation akin to an enzyme - substrate complex in which case association should be a reversible process. The reversibility of the TMP - TMP kinese association was demonstrated by experiments in which enzyme golution containing TMP was fractionated on columns of Sepheder 0-25 (see Section 3.2.4 (c In such experiments, (32P) THP replaced THP and elution of the mixture with 0.01 M phosphate buffer, pH 7.6, revealed that unlabelled TMP kinese apperred early in the elution, followed, in subsequent fractions, by protein-free (32p) THP. A perfect separation of the two components was thus achieved as shown in Fig. 36. As this provided a convenient and rapid means of removing the added TMP, the procedure was routinely resorted to prior to the aspay for THP kinese.

The investigations of Weissman, Smellie and Paul (1960) had given some indication that the TMP kinese responded fevourably to fractionation with dilute scatic acid. Early experiments with the enzyme from Landschutz escites turnour cells, however, demonstrated variable but low retention of TMP kinese activity when exposed to treatment with dilute acid. However, when fractionation (described in Section 3.3.3) with dilute acid was conducted in the presence of excess TMP (80 mumoles TMP per 1.5 mg. protein), 80 - 180 per cent of the total initial TMP kinese activity was recovered in the fraction remaining soluble at pH 4.5 (AS 4.5 Fraction). The recovery of enzyme

- 152 -



Fig. 36. Resolution of a mixture of TMP kinnse and TMP on a column of Sephader G.25.

Enzyme solution was prepared by dissolving the lyophylised cell-free crude extracts of Landschutz ascites tunour cells in 0.01 M phosphete buffer, pH 7.6, to give a protein concentration of 17.25 mg. per al. To this solution was added 40 mumoles (32p) TMP being 0.6 x 10 counts per minute per uncle. A portion (1.0 ml.) of the enzyme solution, now containing 17 mg. protein per ml., was fractionated on a solumn of Sephedex G-25 (20 x 1 om., dia.) previously equilibrated with 0.01 H phosphate buffer pH 7.6, and fractions of 0.5 ml. were collected. Enzyme activity was assayed as described elsewhere (Section 3.2.3 (b)) and other samples were taken for the direct measurement of radioactivity. The column V was 8.0 ml. The recovery of THP kinese activity was 96 per cent while that of (32P) TMP, after collection of 35 ml. effluent, was 95 per cent.

activity, nevertheless, declined as a reciprocal function of the time in contact with dilute acid and this period was never allowed to exceed 15 minut (see Fig. 32). Overlapping fractionation of redissolved, lyophylised crude ascites extract with 1.0 N mostic acid showed that the proportion of the tota initial TMP kinase activity recovered in the acid soluble fraction declined stendily when the pH was lowered from 7.6 to 5.0, below which value retention of TMP kinase increased sharply so that 125 per cent of the total activity was recovered at pH 4.7 (see Table 9). A possible explanation for this observation is that lowering of the pH beyond 5.0 causes a preferential predipitation of contaminating dephosphorylating enzymes or a progressive removal of endogenous inhibitors. While it seems clear that the addition of TMP protects the TMP kinese activity over a wide range of pH values. this effect does not appear to extend beyond pH 4.7, presumably due to the increased rate of enzyme - substrate complex dissociation at lower pH values. The AS 4.7 Fraction retained TMP kinase activity purified 8 - 3-fold over the lyophylised orude extracts and contained TdR kinese activity together with low, but measurable TDP kinese activity.

Subsequent experiments were focussed on methods leading to the elimination of contaminating kinese activities from the TMP kinese preparation One such possible fractionation procedure involved the use of annonium sulphate and preliminary experiments were performed in order to discover the type of fractionation most favourable. Strict control of pH and temperature was shown to be a prerequisite of even moderate success, enzyme solutions being prepared in 0.01 M phosphate buffer, pH 7.6, with saturated amonium sulphate solutions at the same pH. Such fractionations, however, met with little quantitative success and the pattern of fractionation obtainable is well illustrated by the results recorded in Table 10 using the AS 4.7 Fraction. While a fraction precipitated between 50 and 55 per cent ammonium sulphate saturation showed a 2-fold enrichment of TMP kinese activity, this fraction represented no more than 10 per cent of that present initially, the remainder being distributed over the other fractions. Moreover, some 40 per cent of the total TMP kinese activity was unaccounted for and was not found in the supermetant liquor remaining after 70 per cent emmonium sulphate saturation, thus indicating destruction of enzyme activity. This pettern did not appear to be affected by the addition of excess TMP prior to fractionation and further attempts to purify TMP kinese by this method were abandoned. Fractionation with acetons at -15°, similarly, resulted in serious inhibition of TMP kinese activity.

Efforts were directed, next, to the possible use of adsorbent gels as a means of fractionation of the complex socies extract enzyme mixture. $G \not =$ numins is well established as a potentially powerful adsorbent of phosphateses (Gourtois, 1947) and the ense with which this gel can be manipulated makes it a convenient tool in enzyme fractionations. Several exploratory experiments carried out on material from the hypphylised cell-free extract pool (Section 3.2.2 (a)) revealed that the optimal adsorption of protein on $G \not =$ numine occurred at pH 6.0 - 6.3 and that other requirements included low temperature (0 - 8°), high protein concentration in the enzyme solutions and low ionic strength in the buffers used for the adsorption. However,

- 154 -

0	able
L C	L C

	55 - 70	50 - 55	45 - 50	40 - 45	35 - 40	30 - 35	0 - 30	Control	Enzyme fraction per cent ammonium sulphate saturation
	7.33	10.17	7.99	9.77	5.33	4.42	8.87	26.39	Unita per ml.
	36.64	20.34	15.99	1954	10.66	8.85	17.74	210.9	Total no. of units
A Star I was a	1.70	1.62	2.86	2.56	1.50	1.92	2.42	6.85	Protein concentra- tion mg. per ml.
a the second of the	17	6	10	9	5	7	9	100	Per cent recovery of protein
The Contraction of the Contracti	16	10	7	9	5	•	09	100	Per cent recovery of activity
のないのである	4.31	6.2	2.8	3.82	3.55	2.31	3.66	3.84	Specific activity units per mg.

Table 10. The effect of ammonium sulphate fractionation at pH 7.6 on TMP kinase derived from the AS 4.7 Fraction in presence of added TMP.

> Ammonium sulphate fractionation was carried out scoording to the procedure described in Section 3.2.4 (d). Lyophylised AS 4.7 Fraction was dissolved in 0.01 <u>M</u> phosphate buffer, pH 7.6, (to which was added 80 mumoles TMP per 1.5 mg. protei and fractions precipitated by the addition of saturated ammonium sulphate (preadjusted to pH 7.6) were redissolved in the same buffer. TMP kinase activity was measured using the radioactivity assay (Section 3.2.3 (b)) and one unit of TMP kinase activity was defined as that capable of phosphorylating 1 mumole of TMP in 25 minutes.

as the TMP kinase could not be expected to tolerate lengthy operations conducted at pH 6, a compromise set of conditions was employed with both the enzyme solutions and the Cy-alumina suspension equilibrated at pH 7.6. Under these conditions, treatment of the AS 4.5 Fraction with increasing concentrations of GX-alumina gel led to adsorption of TMP kinase along with much of the protein as shown in Fig. 37, the progress of the edsorption being followed by measuring the proportion of the TMP kinese activity remaining unadsorbed (see Section 3.8.4. (b)). Such measurements, suitably correlated with paper chrometography (Section 3.2.3 (b)), indicated that TDP kinase activity was similarly adsorbed. Thus, addition of 5 ml. of I X-elumine suspension to the enzyme solution (1.0 ml.) resulted in the adsorption of all but a few per cent of the TMP kinase activity and further additions of gel did not markedly alter the proportion of adsorbed and unadsorbed enzyme. Significent differences in the edsorption pattern of TMP kinese were observed if excess TMP were added to the AS 4.7 Fraction prior to adsorption on 0 % -alumine gel. Under these direumstances, the rate of adsorption of TMP kinase was markedly reduced so that after the addition of 3 - 5 ml. of gel suspension to the enzyme solution (1.0 ml. /. there remained in solution 40 - 80 per cent of the total THP kinese activity. The relative rates of adsorption of protein and TDP kinase were not so effected. This operation thus led to a supermatent ensyme fraction containing TMP kinese practically uncontaminated with TDP kinese activity. The fraction remaining in the supermetant fluid after the addition of 3 ml. of gel suspension per ml. of enzyme solution was purified 60 - 80-fold over

- 155 -

Fig. 37.



Fig. 37. The adsorption of TMP kinnse on 0 g -alumine gel in the presence and absence of added TMP.

The lyophylised AS 4.7 Frection was dissolved in 0.01 K phosphete buffer, pH 7.6, and the resulting solution divided into two parts, to one of which was added 80 mumoles TMP per 1.5 mg. protein. Portions (1.0 ml.) of these solutions were transferred to 15 ml. centrifuge tubess containing the volumes of C X -alumina gel indionted on the diagram. Adsorption was cerried out for 30 minutes at 0° after which the mixtures were centrifuged and the protein and TMP kinnee activity remaining in the supernetent fluid w Exemessured (Sections 2.2.3 (a and 3.2.3 (b)). The details of the fractionation using C X-alumina gel are described in Sections 3.2.4 (b) and 3.3.5; Nethod I. The results were expressed as mumoles TMP phosphorylated per 85 minutes per supernatant fraction Protein concentrations were likewise expressed as mg. per supernatant fraction. Open circles denote protein concentration and the closed circles refer to THP kinase activity.

the orude socites extracts and was designated the $\Im_X AS 4.7$ Fraction. It seems possible that the pattern of adsorption of TMP kinase observed in the presence of excess TMP is due to rearrangements on the ensyme surface resulting in increased repulsion of fixed charges. That some adsorption does take place is shown by the decline in supermatent TMP kinase activity at the higher levels of \Im_X -aluming gel concentration.

3.3.5. <u>Partial purification of thymidine 5'-monophosphete kinase from</u> cell-free extracts of Landschutz ascites tumour cells

The results described in the previous Section demonstrated that substantial gains in the specific activity of TMP kinase could be achieved by suitable fractionation with dilute acetic acid and by treatment of the enzyme solutions with G_{γ} -alumine gel provided excess TMP was present as a stabilisin factor. It seemed possible, therefore, that such operations might provide the basis of a method for the purification of the TMP kinase present in the arude ascites tumour extracts. Additional preliminary experiments had indicated that some enrichment of the TMP kinase could be achieved by lyophylisation of the arude extracts followed by redissolution of the fary powder in 0.01 <u>H</u> phosphate buffer, pH 7.6, the insoluble portion of the protein being discarded. This observation may be related to the increased thymidylate kinase activities observed in "aged" extracts of celf thymus (Bojarski, 1968) and of Novikoff hepetome (Ives, Norse and Potter, 1962) and it may also have some connection with the release of TMP kinase activity from insolve complexes with small cellular particles as observed by freezing and thewing of mouse liver homogenetes (Kielley, 1963a). Additional enzyme fractionation techniques were required for the removal of various contaminating activities and advantage was taken of the recently developed procedures depending on molecular sieving (Flodin, 1962) with uncharged dextran gels.

(a) Method I

All operations were carried out in the cold $(0 - 2^{\circ})$.

Fractionation with dilute a cetic acid: The lyophylised cell-free extracts of Landschutz escites tumour cells (prepared as described in Section 2.2.1) was dissolved in 0.01 M phosphete buffer, pH 7.6, to give a solution containing 12 - 14 mg. protein per ml. and the residual protein and other insoluble material was removed by contrifugation for 20 minutes at 2000 x g. To the supernetant fluid was added TMP solution to a concentration of 80 mumoles per 1.5 mg. protein and the bulk of the solution diluted to some specified volume from which semples were removed for the estimation of protein and the measurement of TMP kinese activity. The ensyme solution, now containing 10 - 18 mg. protein per ml. in a volume not exceeding 200 ml., was submitted to fractionation with dilute acetic acid. To the solution, cooled in an ice-bath and with continuous stirring, was added 1.0 M acetic acid drop by drop at the pH meter until the pH reached 4.7. The suspension was then quickly transferred to centrifuge bottles and centrifuged at 1000 I g for 6 minutes after which the supernatant was returned to the pH meter and the pH readjusted to 7.6 with the addition of 1.0 M sodium hydroxide. The duration of the whole operation involving contact with a cetic acid should not be allowed to exceed 15 minutes. After the removal of samples for

protein estimation and measurement of kinase activity, the main bulk of the solution (designated the AS 4.7 Fraction) was preserved by hyphylisation and stored at -50° . Since TMP had been added to the solution, a preliminary treatment (Section 3.2.4 (c)) on a column of Sephadex 0-25 (SO x 1 cm., dis.) was required prior to the measurement of TMP kinase activity. The fractionation with acetic acid yielded a TMP kinase preparation purified 3 - 5-fol over the crude extracts. These preparations also contained partially purified TAR kinase and relatively low TDP kinase activity. Lyophylisation did not result in loss of activity.

<u>Fraction-tion with $G \searrow -alumina</u>: The lyophylised AS 4.7 Fraction containing$ TMP was dissolved in 0.01 <u>M</u> phosphete buffer, pH 7.6, to give a solutioncontaining 6 - 10 mg. protein per ml. and a portion was reserved for themeasurement of TMP kinese activity. The main bulk of the solution was added $to 3 <math>\chi$ -alumine gel pretrected as follows: - a volume of 0 χ -alumine suspension (11 mg. per ml.) corresponding to three times the volume of enzyme solution to be fraction-ted was transferred to a 1500 ml. centrifuge bottle and centrifuged at 1000 x g for 10 minutes after which the supernetant fluid was discarded. The residual gel was washed three times with 300 ml. portions of ice-cold 0.01 <u>M</u> phosphete buffer, pH 7.6, the gel being resuspended and centrifuged between each washing. The gel was finally dispersed in a further 400 ml. of buffer and allowed to equilibrate overnight et 0⁹. Prior to use, the suspension was assimented by centrifugation at 1000 x g for 15 minutes end the supernetant liquor was discorded.</u>

To the packed gel contained in the centrifugei bottle was added the

- 158 -

As 4.7 Frection and the gel was dispersed manually. Menual stirring was continued intermittently over a 5 minute period after which the mixture was ellowed to stand in the cold for 20 minutes to complete the adsorption. The gel was then pecked tightly in the centrifuge bottle by centrifugation at 1500 x g for 15 minutes and the supermetant fluid was decented into a precooled measuring cylinder. The volume was noted and, after the removal of samples for estimation of protein and measurement of TMP kinese estivity. the main bulk of the solution was hyphylised and stored at -50°. Measurement of TMP kinese activity required pretrestment of the preparation on a column of Sephedex G-85 in order to separate the enzyme from TMP. The fraction obtained by the 0 χ -alumine treatment was designated the 0 χ^{15} 4.7 Fraction and contained 60 - 80 per cent of the total kinese activity could be detected. The TMP kinese so obtained was purified 80 - 100-fold over the crude escites extract.

<u>Prectionation on Sephedex G-100</u>:- The lyophylised $G_X \pm 5$ 4.7 Frection was onrefully extracted with 3 - 4 ml. of 0.01 <u>M</u> phosphate buffer, pH 7.6, and any insoluble material was removed by centrifugation at 2000 x g for 20 minutes. The enzyme solution so prepared contained 30 - 40 mg. protein per ml., 2 ml. of which was transferred to a column of Sepheder G-100 (95 x 1 cm., dis.) previously prepared and equilibrated with ice-cold 0.01 <u>M</u> phosphate buffer, pH 7.6, as described elsewhere (Section 3.2.4 (c)), the remainder of the enzyme solution being used for the estimation of protein and macsurement of the TMP kinnes activity by the radioactivity essey (Section 3.2.3 (b). The void volume of the column had previously been determined by the volume required to move a sample of χ -globulin from the top to the bottom of the column and was generally found to be 25 - 30 ml.

Gradient elution with 0.01 M phosphate buffer, pH 7.6, was started after all the enzyme solution had passed into the column bed and proceeded at a flow rate of 15 - 20 ml. per hour, fractions of 2.0 ml. being collected from the point where a volume corresponding to V, had appeared in the effluent Samples were withdrawn from each fraction and assayed for THP kinase activity. A representative chromatographic profile is given in Fig. 38 from which it is seen that TMP kinase activity appears in the effluent as an irregularly shaped peak some time after the appearance of the main protein peak. 30 Fractions were collected in this way and the six tubes of the TMP kinese peak with the highest specific activity were combined and designated the SO XAS 4.7 Fraction The kinese activity of such fractions was sufficiently high to merit the use of the spectrophotometric assay for TMP kinase (Section 3.2.3 (e)). However, the use of this assay technique revealed also that superimposed upon the THP kinese peak there was a highly active ATPase which, when measured with the spectrophotometric procedure given in Section 3.2.3 (e), was shown to hydrolys ATP at the rate of 2 µmoles per 85 minutes per mg. protein. Mevertheless, 40 - 60 per cent of the total TMP kinase activity in the Cy AS 4.7 Fraction was recovered in the SCX AS 4.7 Fraction purified 1000-fold over the crude extracts and having a specific activity of 0.6 - 0.9 µmoles TMP phosphorylated per 25 minutes per mg. protein. No TDP kinese could be detected in this preparation, neither was there any evidence of phosphorylation of TdR.





Fig 38

Fig. 38. Ghrometography of the Gg AS 4.7 Fraction on a column of Sephader G-100 dextran gel (95 x 1 om., dim.).

The eluant was 0.01 Ξ phosphete buffer, pH 7.6, and the datails of the fractionation were those described in the text (Section 3.3.5; Method I) and the preparation and equilibration of the dextrem column were given in Section 3.8.4 (c). The column V₀ was 27 ml. TMP kinnse activity was measured with the spectrophotometric assay (Section 3.2.3 (c)) and the fractions within the brackets were combined and constituted the SC_XAS 4.7 Fraction. However, because of the presence of ATPase as a contaminant in the SG_X AS 4.3 Fraction, it seemed possible that the ATPase or some other phosphatase present in the preparation was capable of hydrolysing TTP. If this were so, no TTP would be expected to appear among the reaction products in assays wher (^{32}P) TMP was used an substrate since the activity of a possible TDP kinase would be masked by dephosphorylating activity, the TTP being reconverted to TDP as quickly as it was being formed. Several experiments in which this possibility was investigated have shown that no such TTP dephosphorylating activity (Section 3.8.3 (e)) was present in the SGYAS 4.7 Fraction (see Table 13).

The purification of TMP kinese by this method thus demonstrated that the kinese responsible for the phosphorylation of TMP to TDP was a discrete enzyme and was separable from the TdR kinese and TDP kinese responsible, respectively, for cetalysing the initial and the ultimate steps in the formation of TTP from TdR. The serious contamination of the TMP kinese with ATPase, unfortunately, minimized the usefulness of the SG gAS 4.7 Fraction as a source of TMP kinese for future kinetic characterisation.

The over-all results of the purification scheme are summarised in Table 11.

(b) Method II

Another type of purification scheme was developed with the express purpose of eliminating phosphatase contaminants from the TMP kinase. In the scheme described below, the initial step was identical to that employed in Method I, more care being taken, however, to elucidate the fate of the TdR

- 161 -

1.00
P
5
H
0
H
H

Pinzyme fraction	Units per ml.	Total olumo ml.	Total no. of units	Protein concentra- ticn mg. per ml.	Total amount of protein mg.	Specific activity units per mg.	
Grude . soluble extracts	2.34	3335	7800	3.52	9510	0	82
Redissolved Lyophylised crude extract	17-5	410	7000	14.12	5650	H	.25
AS 4-7	19-4	400	7950	6.04	2475	U	.21
CYAS 4.7	17.7	423	7500	0.22	01	82	4
SCYAS 4.7	415	10	4150	0.51	5	950	

Table 11. Purification acheme for TMP kinase derived from cell-free extracts of Landschutz ascites tumour cells (Method I).

> TMP kinnse activity was measured using the redicectivity assay (Section 3.2.3 (b)) throughout, and the values obtained by this means for the SC × AS 4.7 Fraction were checked against values obtained using the spectrophotometric method for the measurement of TMP kinnse activity (Section 3.8.3 (c)). Good agreement between the two methods was obtained with purified TMP kinnse preparations while there was considerabl divergency with more impure preparations.

and TDP kineses in the course of the enzyme fractionations.

Fractionation with dilute acetic acid: - Lyophylised orude extract was dissolved in 0.01 M phosphate buffer, pH 7.6, and centrifuged to remove residual protein. To the supernatant fluid, containing 12 - 14 mg. protein per ml., was added TMP to a concentration of 80 munoles per 1.5 mg. protein and fractionation with 1.0 N agetic acid was carried out as described above. The two fractions obtained by lowering the pH to 4.7 were preserved and immediately neutralised with the addition of 1.0 M sodium hydroxide. Samples were withdrawn from each of the fractions for the estimation of protein and measurement of kinese activities, the TMP being removed by treatment with Sephader G-25 (Section 3.2.4 (c)) prior to measurement of TMP kinnse activitie The main bulks of the two Fractions (AS 4.7 and AR 4.7) were lyophylised and stored at -50°. The AS 4.7 Fraction contained 120 per cent of the total initial TMP kinese activity purified 6 - 8-fold over the grude extracts together with 80 per cent of the total TdR kinese activity purified 2 - 3-fold and 5 per cent of the total initial TDP kinase activity. The AR 4.7 Fraction when similarly assayed for kinese activities, was shown to contain 25 per cent of the total initial TMP kinese activity, 40 per cent of the TdR kinese activity and 10 per cent of the TDP kinase activity of the lyophylised cruie extracts. Thus some 80 per cent of the TDP kinase activity was destroyed by the sold treatment while the recoveries of TdR and TMP kinase activities exceeded the measured values of the lyophylised grude extracts by 20 and 50 per cent, respectively.

Prectionation on Sepheder G-100:- Lyophylised AS 4.7 Fraction (contrining

- 168 -

THP) was dissolved in 3 - 4 ml. of 0.01 <u>M</u> phosphete buffer, pH 7.6, and centrifuged, if necessary, to remove any residual protein. After portions had been reserved for the estimation of protein and measurement of kinase activities, 2.0 ml. of enzyme solution (containing 30 - 40 mg. protein per ml.; were run on to a column of Sepheder G-100 (86 x 1 cm., dis.; previously equilibrated with ice-cold 0.01 <u>M</u> phosphate buffer, pH 7.6 (see Section 3.2.4 (a)). The column void volume as measured with χ -globulin, was 30 ml.

Oradient elution was performed with 0.01 H phosphate buffer, pH 7.6, at a flow rate of 15 ml. per hour and fractions of 2.0 ml. were collected from the point where a volume corresponding to Vo had passed through the 30 Fractions were collected in this way, semples of each being column. withdrews for estimation of protein and measurement of kinase activities. The remainder of each fraction was lyophylised and stored at -50°. Ho treatment with Sephader 0-25 was necessary prior to the TMP kinese assay since TMP was grougly retained on the Sepheder G-100 dextrem gel and only appeared in the column effluent after some 200 ml. of eluent had been applied. A representative chrometographic profile incorporating effluent protein concentration and the results of easeys of TdR kinese (Section 3.2.3 (a) /. THP kinese (Section 3.2.3 (b)), TDP kinese (Section 3.2.3 (b)) and ATPase (Section 3.8.3 (e)) is shown in Fig. 39. Three peaks of kinase activity were obtained. The first, representing TdR kinese, appeared in the effluent superimposed on the main protein peak and the second contained TDP kinese contaminated with ATPase and a small amount of TMP kinase accounted for by elight overlapping of the third peak of THP kinase activity. The third





Fig. 39. Chrometography of the AS 4.7 Fraction on a column of Sephader G-100 dextran gel (98 x 1 cm., dia.).

> The eluent was 0.01 M phosphate buffer, pH 7.6 and the details of the fractionation were those described in the text (Section 3.3.5; Method II). Fractions of 2.0 ml. were collected at 2° and at a flow rate of 15 ml. per TdR kinese was measured as described earlier hour. (Section 3.2.3 (a)) with (3) TdR as substrate and the results recorded as the percentage of total radioectivity recovered in (TMP + TDP + TTP) per ml. of effluent after an ingubation period of 15 minutes. THP kinase activity was measured by the radioactivity assay (Section 3.2.3 (b with (329) TMP as substrate. TDP kinase was measured by the use of paper chrometographic techniques (Section 3.8.3 (b)). ATPese activity was detormined by the spectrophotometric essay given in Section 3.2.3 (e). The frections containing TMP kinese uncontaminated with TDP kinese activity (after 64 - 75 ml. effluent had been collected) were combined and designated the SAS 4.7 Fract

Prot	tein	=	0
ATP		2	a
Tar	kinese		•
THP	kinse		4
TDP	kinse		1

peak contained THP kinase and the six fractions of the highest specific activity were combined and designated the SAS 4.7 Fraction. The recovery of THP kinase activity as represented by the SAS 4.7 Fraction was 40 per cent of that applied to the column which contained TMP kinase purified 60-fold over the crude ascites extracts. Moreover, the TMP kinese so obtained was cepeble of phosphorylating TMP to TDP only, thus indicating the absence of contamination with TDP kinase; neither was there any evidence of TdR kinase activity in this preparation. However, as (32p) TMP was used as substrate for the measurement of TMP kinese (and TDP kinese), the absence of TTP from the reaction products could be due to contamination of the SAS 4.7 Fraction with nucleoside triphos phate dephosphory lating enzymes. No such dephosphorylating activity was found when ATP was omitted from the reaction mixtures and (32P) TTP replaced (32P) TMP as substrate as recorded in Table 11 A peak of ATPase activity, as mensured by the spectrophotometric assay (Section 3.2.3 (e)), appeared in the effluent preceding the TMP kinese peak. There was evidence also of apparent tailing off of ATPase into the SAS 4.7 Fraction as samples taken from the latter Fraction catalysed a slight drop in the extinction at 340 mu when incubated with the ATPase assay mixture. Subsequent experiments, however, showed that the exidation of NADH, by the SAS 4.7 Fraction was independent of ATP and due, possibly, to flavoprotein contemination. It must be emphasized also that the TDP kinese as measured in the effluent need not represent more than a small proportion of the total TDP kinase present as no account was taken of possible TDP kinase activity appearing in the chrometogram independently of TMP kinase activity. The

- 164 -

progressive removal of TDP kinase from TMP kinase in the course of the purification is demonstrated in Fig. 40 where paper chromatograms of reaction products run at different stages of the purification of TMP kinase show the gradual disappearance of biosynthetically formed TTP.

The over-ell results of the purification scheme are summarised in Table 12.


Fig. 40. Paper chrometographic separation of reaction products obtained in the phosphorylation of (³²P) THP by ensyme preparations at various stages of the purification of TMP kinase.

> Enzyme experiments with (³²P) as substrate were carried out as described in the text (Section 3.2.3 (b)) followed by submitting the reaction products to descending paper chromatography in isobutyric acid/distilled water/ammonis HDTA (Section 2.2.3 (b)). Appropriate (³²P) -containing lanes were out from the chromatograms and scenned for radioactivity using the Nuclear-Chicago Actigraph windowless gas-flow detector (Section 2.2.3 (d)). Marker compounds included in the chromatograms are shown. The sources of enzyme were as given on the diagram.

Dasyme fraction	Units per ml.	Total volume ml.	Total no. of units	Protein concentra- tion mg. per ml.	Total amount of protein	Specific activity units per mg.	Furifi- oation
Crude, soluble extracts Rediscolved, lyophyl18ed	1.91		143	3.01	227	0.63	L
extrects AS 4.7	11.68	10	117	12.3	582 582	0.95	5.5
SAS 4.7	8.47	12	101	0.23	2.7	37-2	60

Table 12.

Table 12. Purification scheme of TAP kinsse derived from cell-free extracts of Landschutz ascites tumour cells (Method II).

> TMP kinase activity was measured throughout using the radioactivity assay (Section 3.2.3 (b)).

Table 13.

Ensyme fraction	er cont otal redio- ctivity re- overed	Per cent total redic- activity in thymidine nucleotides			TDP form
		TTP	TDP	THP	25 min.
(- ensyme; not heated)	95	95.5	4.5	-	0.90
Control (-ensyme: hea- ted)	90	95.5	4.5	-	0.90
Control (+ ensyme)	92	96	4	-	0.82
SC.AS 4.7 Fre-		-			
fraction 9	95	96	4	-	0.82
10	90	96	4		0.82
11	89	96.5	3.5	-	0.73
SAS 4.7 Prec- tion	and the ser	a start			
fraction 18	91	97	3	-	0.6
19	93	95.5	4.5	-	0.9
20	90	96	4	-	0.82

* protein (0.5 mg.; fraction 10) of the SCyAS 4.7 Fraction boild for 2 minutes before addition.

** calculated on the basis of the total radioactivity present in: tially.

Table 13. Determination of phosphetase activities in fractions eluted from columns of Sephadex 0-100 in the course of TMP kinase purification.

> Reaction mixtures were propered containing the following components in a total volume of 1.0 ml.:- Sumples Mg3lg, 100 µmoles tris buffer, pH 7.6, 20 mµmoles (38 P) TTP of specific activity 10⁷ counts per minute per µmole and 0.5 mg. protein from fractions eluted from columns of Sephadex G-100 comprising the 33 $_{5}$ AS 4.7 Fraction (Method I) and the SAS 4.7 Fraction (Method II). Incubation was for 25 minutes at 37⁰ and further treatment was as described for the redioactivity essay of phosphateses (Section 3.2.3 (e The results were calculated as the number of mµmoles TTP being hydrolysed in 25 minutes.

CHAPTER IV.

1

12.1.1

化比

GREERAL DISCUSSION

4.1. The mechanism of formation of thymidine 5'-triphosphete from thymidine and thymidine 5'-monophosphate by ensymps from Landschutz ascites tumour cells

It is now widely accepted that the replicative synthesis of DNA shows an absolute requirement for the 5'-triphosphates of decxyndenosine, decxyguenosine, decxycytidine and thymidine and that the synthetic reactions leading to the formation of these precursors are so controlled that they are produced in the correct amount, at the correct time and at the appropriate site. It seems equally certain that in the ultimate stages these mechanisms are identifiable with a set of decayribonucleotide phosphokineses, each component of which responsible for the phosphorylation of one of the decayribonucleoside 5'-monophosphates dAP, dOMP, dGMP and TMP to their respective 5'-triphosphates. Of the decayribonucleotide phosphokinese systems, that concerned with the phosphorylation of TMP (and TAR) to TTP has been singled out for special attention, the uniqueness of this system being derived partly from the chemical dissimilarity of its substrates to ribonucleotide counterparts and partly from its powerful potential as a regulator of ENA synthesis and hance of cell division.

The present investigation has shown that cell-free extracts of Landschu sacites tumour cells enclose the full complement of phosphokineses required to phosphorylate thymidine and thymidine 5'-monophosphate to thymidine 5'-triphosphate. Pertinent questions were asked as to the precise sequence of events in this phosphorylation process and several types of experiment were carried out with the express purpose of guaging the maximum potentialities of the system. As a result, the conclusion can now be confidently drawn that the phosphorylation of TdR to TTP proceeds through the intermediates TMP and TDP. This conclusion was based on two independent lines of evidence; firstly the kinetic experiments performed with cell-free extracts of Landschut ascites tumour cells provided strong evidence for the role of TDP as an intermediate in the formation of TTP from TdR and TMP (Section 2.3.2) and secondly, the identification and resolution of three phosphokineses derived from such extracts, each responsible for cetelysing one of the phosphorylations TdR to TMP. TMP to TDP and TDP to TTP (Sections 3.3.2., 3.3.4 and 3.3.5).

It is important, at this point, to consider to what extent these general conclusions are justified by the specific evidence presented, immediate implications of which have been noted elegwhere (Sections 2.3 and 3.3). It seems necessary also to reiterate some of the limitations inherent in the experimental techniques employed, particularly those concerned with the kinetics of thymidine and thymidylate kinase action. Thus, the reaction rate measurements invariably reflect net phosphorylation and, when operated under optimal conditions, represent the optimal net rate obtainable with the most favourable balance of phosphorylating and dephosphorylating activities. It follows that the formation of TTP from TdR, in general, and the individual intermediate reactions, in particular, may not have realised their moximum potential and the purified component enzymes may well exhibit different optimal requirements from those given here (Sections 2.3.1., 3.3.2 and 3.3.4).

From the time course experiments with (³H) TdR and (³²P) TMP (Figures 22, 23 and 26), it was clear that inbelled TDP was invariably found in the

- 167 -

reaction products prior to labelled TTP and the low relative radioactivity in TDP meintained throughout the incubations was consistent with the behaviour expected of a transient intermediate. These results also suggested other points. Examination of, for example, Fig. 28 indicates that the rate of formation of TDP from TMP was maximal when the concentration of TMP in the reaction mixtures was maximal also and the subsequent fall in the TMP concentration for full seturation then do the TdR and TDP kinases. Furthermore, from the experiments with $({}^{32}p - a)$ TDP (Fig. 24; Table 3) and $({}^{32}p - b)$ TDP (Fig. 25), it is clear that TTP was formed very repidly without the intermediate formation of TMP or higher phosphates of TdR (or TMP) than TTP.

These conclusions are clearly consistent with the hypothesis of Weissman, Smellie and Paul (1960) that formation of TTP from TdR proceeds through TMP and TDP when catalysed by enzymes derived from cell-free extracts of Ehrlich escites tunour cells. The enzyme fractionation experiments of those authors had indicated the existence of three different phosphokinases, each responsible for one dep in the formation of TTP from TdR. Sommistent with those findings is also the partial separation of TMP phosphokinase and TDP phosphokinase by the centrifugation of extracts of calf thymus in sucrose density gradients (Bojerski, 1962).

That the initial step in the formation of TTP from TdR is the phosphorylation of TdR to TMP has for some time been considered beyond dispute. Thus Bollum and Potter (1959) demonstrated a clear precursor-product relationship between TdR and TMP in time course experiments with extracts of regenerating ret liver. The strongest evidence for the identity of the initial reaction, however, came from enzymological studies on TdR kinase which, by virtue of its relatively higher stability, was readily separable from the thymidylate kinases (see Sections 3.3.2 and 3.3.3; Figures 32 and 39). These conclusions were further supported by the findings of Ives, More and Potter (1963) using extracts of Novikoff hepatoms and, recently, by the purification of a TdR kinase from <u>E.coli</u> (Okazaki and Kornberg, 1964a). All these preparations have been shown to catalyse the phosphorylation of TdR to TMP without also forming thymidine polyphosphates.

While the initial reaction, estalyzed by TdR kinese, is thus well established and characterised, the sequence of events in the phosphorylation of TMP resulting in the formation of TDP and TTP remains a matter of considerable argument. This existing uncertainty has not been elleviated by the fact that several of the reported investigations have neglected to make a precise distinction between the types of product formed by the action of the thymidylate kineses on TMP, one reason for this emission being the failure of the methods used to separate adequately the reaction products. Thus it has become customary to express the products of thymidylate kinese action as (TDP + TTP) (Beseman, 1959; Bollum and Potter, 1959; Manstavinos and Ganellakis, 1959; Kielley, 1961, 1963s, b; Ives, Morse and Potter, 1963). Tittle valid information on the sequences of the individual reactions could be deduced from experiments using this approach. Home of the other studies with a direct bearing on the mechanism of the phosphorylation of TdR and TMP to TTP suffer from similar analytical deficiencies. It was therefore

- 169 -

necessary to exercise some caution in comparing the results of such studies with those presented here (ofr. resolution of thymidine nucleotides Section 8.2.4 (c); Fig. 17).

The general conclusion that TDP is an intermediate in the formation of TTP from TdR and TMP has been supported by experiments on HeLa cells and Hela cells infected with the virus of cowpox (McAuslan and Joklik, 1962). In experiments on the time course of phosphorylation of (143) THP with enzyme preparations from HeLa cells, radioactive TDP was detected prior to the formation of any TTP and the results thus resembled the pettern obtained with extracts of Landschutz ascites percinome (Figures 22, 23 and 26). Moreover. the percentage of total redicactivity in TDP always exceeded that in TTP as expected from stendy state tracer kinetics and it was also shown that the accumulation of radioscivity in TDP from (140) TMP was independent of the addition of a 5-fold excess of unlabelled TDP. While these results all point to the reaction sequence TMP -> TDP -> TTP, an unequivocal conclusion does not appear justifiable from the evidence presented by McAuslan and Joklik (1962). Thus, for the reasons stated in Section 2.3.2, it does not seem conceivable that a closed in vitro system such as that employed by McAuslan and Joklik (1962) could approach the steady state claimed by these workers and, furthermore, measurements of specific redicactivity were not performed in any of the experiments quoted. The most acceptable evidence comes from the experiment in which the time course of phosphorylation of (14c) TMP was studied in presence and absence of added, unlabelled TDP. The significance of this experiment lies in that if TDP were an intermediate in

the formation of TTP from TMP, then the addition of unlabelled TDP would be expected to result in significant reduction of TTP specific activity (as was in fact observed . However, the alternative mechanism of Bianchi. Butler, Grathorn and Shooter, (1961) postulates the sequence THP TTP -> TDP and it seems possible that the results observed by McAuslan and Joklik (1962) could be explained in terms of this sequence depending on the relative rates of the backward reactions. In the present investigation, crude cell-free extracts of ascites tumour cells were found to contain dephosphorylating ensymes acting on TDP and TTP and a experiment of the type quoted above could therefore not, by itself, be expected to provide valid evidence on the mechanism of formation of TTP. Somewhat surprisingly, McAuslan and Joklik (1962) state that nucleotidases acting on TDP and TTP were not present in the cell-free extracts of Hele cells. This statement is unsupported by their evidence since phosphetase activity was being sought in the presence of ATP and thereby providing a regenerating system for thymidine nucleotide substrates.

Giroumstantial evidence for the intermediate formation of TDP in the phosphorylation of TMP to TTP from a somewhat different precept has been provided by Delamore and Prusoff (1962) who studied the competitive antegonism of 5-10do-2'-decryuridine (IUdR) of TdR phosphorylatica under the influence of several human neoplastic tissues. This antegonism was presumed to operate by phosphorylation of IUdR to the corresponding 5'-mono-, 5'-diand 5'-triphosphates and subsequent incorporation of IdUMP into DMA specifically in place of TMP. When the uptake of (³H) TdR for DMA synthesis by

- 171 -

whole cells was used as a guide to the effect of IUdR, it was found that inhibition was directed towards specific steps in the phosphorylation of TdR depending on the type of cell employed. However, it was found with all the cell populations tested that the proportion of total redicectivity is TDP always exceeded that in TTP whether or not the phosphorylation was inhibited by IUdR and that IUdR had no affect on the ratio of redicectivity in TDP to that in TTP. This provided an indication that formation of TDP preceded formation of TTP and Delemore and Prugoff (1962) argued further that if TDP were not on the direct pathway for the biosynthesis of TTP, then the latter product must be particularly unstable in the system studied.

All the findings described so far are in direct contrast with those of Bianchi, Butler, Grathorn and Shooter (1961) who found, in experiments with cell-free extracts of regenerating rat liver, leukaemic nouse spleen and Ehrlich escites tumour cells, that the formation of TDP was always preceded by accumulation of TTP. These observations led these workers to conclude that TTP was formed from TMP by a one-step reaction involving the addition of pyrophosphete and that TDP only arose by degradation of TTP by, presumbly, phosphatese action. However, upon closer examination, it is apparent that this conclusion is kinetically untenable if based on the evidence presented.

Thus it is clear that the chromotographic procedure used by Bianchi, Butler, Grathern and Shooter (1961) for the separation of thymidine nucleotides was inferior to the separation of such compounds on columns of SOTEOLA cellulose as employed in the present investigation (Fig. 17), particularly with regard to the resolution of TDP and TTP. Since at the best only small

- 172 -

amounts of TDP were detected at the early stages of time course experiments (Figures 22, 23 and 26), it seems quite possible that TDP might not have been detected in all the experiments of Bienchi, Butler, Grathorn and Shooter (1961). Moreover, close inspection of the time course experiments on the phosphorylation of (14G) TdR as given by these authors revealed that inadequate data was presented for the redicectivity in TDP at time intervals of less than one hour in spite of the observation that little or no TdR remained in the reaction mixtures at the end of this time. Bianchi, Butler, Grathorn and Shooter (1961) admit that their results would be compatible with the reaction sequence: TdR -> THP -> TDP -> TTP if the rate of phosphorylation of TDP to TTP were rapid compared with the rate of formation of TDP from TMP in which case the appearance of TDP would only be transient. However, the experiments which they cerried out to investigate this possibilit seemed of doubtful value. Thus, in these experiments, the phosphorylation of TdR by diluted enzyme preparations was examined in order to slow down the initial stages of the reaction, no TDP being detected under these circumstance The conclusion that TDP was not an intermediate in the blosynthesis of TTP fro this data seems very questionable since it is difficult to believe that more dilution of the enzyme preparation would slow down the reaction leading from TDP to TTP to a greater extent then that leading from TMP to TDP. Another explanation which may be offered for the failure to detect TDP, at early time intervals, is that the orude extracts used by Bianchi, Butler, Grathorn and Shooter (1961) contained, in addition to TMP kinese, a diphosphokinase capable of phosphorylation TDF to TTP. The inclusion of an ATP regenerating

- 173 -

system such as that used would tend to drive the reaction to the right and thereby favour the production of TTP at the expense of the accumulation of TDP. The presence of a diphosphokinase soting on TDP has been demonstrated in extracts of Landschutz ascites tumour cells (Figures 24 and 25) and this possibility has also been favoured by Bessman (1963).

From the results given in Section 2.3.2, it is clear that THP and TTP were formed very repidly from TdR and that TTP was formed very repidly from TDP. Under these direumstances, it seemed possible that the formation of TDP from TMP was the rate limiting step in the over-all reaction sequence, in which event TDP would only be expected to appear as a transient intermediat

It seems clear that the sotivity of the thymidine and thymidylate kinese system from Landschutz excites carcinoms is markedly sensitive to accumulation of reaction products and hence the maintenance of a high rate of TTP formation is dependent on its efficient utilisation. This is in full egreement with the findings of Delemore and Prusoff (1962) that specific inhibition of DNA polymerase in Ehrlich assites tumour cells and in celf thymus by IUdR led to a drastic reduction in the radiomotivity recovered in TTP from (¹⁴G) TdR. The negative feedback inhibition of TdR kinase by TTP demonstrated by Ives, Norme and Potter (1962, 1963) is of considerable importance in this connection and will be discussed later (see Section 4.3). In the present experiments, the presence of excess unlabelled TDP and TTP was found to inhibit markedly the formation of TTP from (³²?) TMP (Fig. 87) while it was not certain whether this effect was due to a general mass estion inhibition by both TDP and TTP on TMP kinase or a more specific inhibition by either TDP or TTP. It is of interest to note also that in phosphorylation experiments with (¹⁴3) TMP by extracts of Hela cells, the proportion of total redicactivity recovered in TDP was unaffected by the addition of a 5-fold excess of unlabelled TDP (McAuslan and Joklik, 1962) thus suggesting that the inhibition of TMP kinese observed in the present experiments might be due to TTP alone or to a combination of TTP with TDP.

Little is known as to whether the mechanisms here proposed are compatible with the patterns obtained in the integt cell. Only one investigation is on record in which the phosphorylation of TdR has been studied in vivo with simultaneous determination of the pool sizes of the thymidine nucleotides. The relationship between precursors and products in the spleen and thymus of the rat was studied by Potter and Nygmerd (1963) following the intreperitoncel injection of (14 C) TdR to rate. The results of such experiments showed that in the spleen, the specific activity - time curves for the thymidine derivatives were compatible with the sequence:

TAR \rightarrow TMP \rightarrow (TDP \rightleftharpoons TTP) \rightarrow DNA

1.2.1

Thus it was not possible to distinguish which of the thymidine polyphosphates, TDP or TTP, was the direct precursor of DNA - thymine and no simple precursor-product relationship was established between TMP, TDP and TTP as a result of very repid equilibration of radioactivity between TDP and TTP.

In view of the complicating factors mentioned above and the possible interference from nucleotidases suggested by Bianchi, Butler, Grathorn and Shooter (1961), it seemed clear that significant evidence related to the mechanism of formation of TTP would most readily be obtained if it were possible to separate and characterize the individual reaction steps. As stated elsewhere (Section 3.3.4), it seemed evident that the hypothesis claiming TDP as an intermediate in the formation of TTP from TMP would receive powerful support were it possible to demonstrate unequivocally the presence in the orude assites tumour extracts of a phosphokinase capable of phosphorylating TMP to TDP.

The enzyme fractionstion experiments of Weissman, Smellie and Paul (1960) had indicated the presence of a TMP kinese in cell-free extracts of Ehrlich macites tumour cells and this prediction has been amply substantiated in the present investigation using cell-free extracts of the related Landschul ascites tumour cells (see Sections 3.3.4 and 3.3.5). Thus it was demonstrate that a phosphokinase responsible for the phosphorylation of TMP to TDP was present in the grude cell-free ascites tumour extracts, and several experiments indicated that the THP kinase activity so measured was the property of a discrete protein separable from the TdR and TDP kineses (Figures 32 and 39). Two different types of purification procedure (Section 3.3.6) led to TMP kinese preparations purified 60-fold or 1000-fold over the grude ascites tumour extracts. Such preparations were shown to catalyse the transformation of TMP into TDP while being incepable of forming TMP from TdR or of forming TTP from TDP (Fig. 40). In one procedure, TMP kinese was purified by removing inactive protein by dilute acetic acid precipitation followed by edsorption of further inactive protein by treatment with C X -elumine gel and, finally, by fractionation on a column of Sephader G-100 dextran gel (Table 11)

while this preparation contained TMP kinese purified 1000-fold over the orude extracts, it was shown to contain also a highly active enzyme hydrolysing ATP. This, unfortunately, made the preparation unsuitable as a source of TMP kinese for stoichiometry experiments and for kinetic elucidation of the reaction leading from TMP to TDP. However, the ATPase was shown to be incepeble of dephosphorylating TTP (Table 13) so that the appearance of TDP as the end product of the reaction could not have been caused by a nucleotidase mediated degradation of TTP.

The second procedure led to a TMP kinase preparation purified 60-fold over the crude extracts by acid predipitation (as above) followed directly by fractionation on a column of Sephader G-100 (Table 12). The treatment with Sepheder G-100 dextran gel proved of particular significance in that it provided a means of separating the relevant kinase activities on the besis of the molecular weight of their constituent proteinsl The TMP kinese preparation so obtained was free of TdR kinase, TDP kinase and ATPase, all of which activities appeared in the effluent prior to the TMP kinese activity (Fig. 39). According to Flodin (1962), polymers of molecular weight exceeding 100 000 will pass through a column of Sephader 0-100 unhindered while polymers having a molecular weight of less than 100 000 will exhibit partition coefficients for the gel of between zero and one. An approximate measure of the molecular weight of a particular protein may thus be obtained with the aid of appropriate marker polymers. In the present experiments, the THP kinase was observed to appear in the effluent imperfectly separated from the heem containing band of grude sheep blood heemoglobin and a molecular

- 177 -

weight of 50 000 - 60 000 may therefore be tentatively ascribed to the partially purified TMF kinase. By the same token, the TdR kinase would appear to have a molecular weight exceeding 100 000 while the extremely low levels of TDP kinase activity present at this stage of the purification made its precise position on the chromatograms somewhat more uncertain.

TdR kinese has likewise been purified (4 - 5-fold) by virtue of the greater stability of this enzyme in dilute acid media under which dircumstances the TMP and TDP kinese activities appear to be destroyed (Fig. 38; Table 6). The TdR kinese obtained after removal of inactive protein by dilute acetic acid precipitation was shown to catalyze the formation of TMP from TdR without being capable of phosphorylating TMP to TDP and TTP.

The instability of the thymidine and thymidylate kineses in aqueous solution proved a major obstacle in the characterisation of those enzymes (see Section 3.3.3). However, the evidence was consistent with the view expressed by Meisaman, Smellie and Paul (1960) that the kineses were not equally labile and it was generally found that in order of dedreasing stability were TdR kinese, TMP kinese and TDP kinese. The addition of excess TMP to enzyme solutions was found to protect TMP kinese activity while being without effect on the TdR or TDP kinese activities. This protection of TMP kinese by its substrate originally predicted by Hiatt and Bojarski (1960) was thus confirmed. It is difficult to see what form the essociation of TMP kinese with TMP takes but presumably the specific protection of active groups on the enzyme surface is involved. The formation of a definite enzyme - substrate complex under these circumstances cannot be excluded, but this seems unlikely since the stabilising effect of TMP was observed in the absence of ATP and Mg and whatever molecular event does occur, the TMP - TMP kinase association was shown to be freely reversible (see Fig. 36).

The properties of the TdR kinase of Landschutz ascites carcinoma generally concur with those ascribed to the Thrlich ascites enzyme by Weissman Smellie and Paul (1960) although the TdR kinese in the present investigation did not respond to ammonium sulphate fractionation. However, the partially purified TdR kinese appeared to be different from the TdR kinese of Novikoff hepatoms (Ives, Morse and Potter, 1963) since it was more labile and showed different requirements for ATP and Mg ++ (see Fig. 31). Comparison with the highly purified TdR kinese of E. coli (Okezaki and Kornberg, 1964a) was somewhat inappropriate owing to the exceptionally high reaction velocity of that enzyme; however, the pH optime of the two enzymes appeared to be similar as were the Mg requirements and the stabilisation with 2-mercaptoethanol. As found in the present investigation (see Fig. 32), Okezeki and Kornberg (1964a) showed that it was possible to separate the TdR kinese from TMP and TDP kineses by treatment with dilute sold and the purified E.coli enzyme was shown to catelyse stoichiometrically the reaction:

TdR + ATP - ADP

The TMP kinese had several properties which distinguished it from the non-specific nucleoside monophosphate kinese of Strominger, Heppel and Maxwell (1959). Thus the TMP kinese was very much more labile, had a lower reaction velocity and was destroyed by scetone and ammonium sulphate fractionations. Similarly, the eacites tumour TMP kinese differed considerably from the <u>s</u>. <u>ooli</u> TMP kinese described by Hurwitz (1959) in that soctone and asmonium sulphate fractionations resulted in serious loss of activity. The TMP kinese also appears to be distinct from the thymidylate kinese found in extracts of Novikoff hepatoma (Ives, Morse and Potter, 1963) which had different ATP and Mg⁺⁺ requirements. Only one report is available in the literature of a TMP kinese which stoichiometrically catalyses the reaction:

THP + ATP = TDP + ADP

Bello and Bessman (1963a) have described the purification of a new enzyme induced in E. coli following infection with the T2 bacteriophege. This enzyme was readily separable from the thymidylate kinase of uninfected bacteria but, unlike the latter enzyme which appeared to be apacific for TMP (Hurwitz, 1959), the newly formed TMP kinase was shown to be tri-functional and it was capable of phosphorylating TMP, dGMP and 5-MeOH-dGMP to the corresponding diphosphates, all of which activities appeared to be the property of single protein. Bello and Bessman (1963a, described the kinetic properties of this enzyme in some detail. Thus both ATP and dATP could serve as phosphoryl group donor, Mg or Mn were required for optimal activity and the enzyme had a broad pH optimum in the range 7.0 - 9.3. This latter property clearly distinguished it from the TMP kinese of Landschutz ascites tumour cells which exhibited a perticularly sharp pH optimum at pH 7.6 (Fig. 34), and while the partially purified T2 bacteriophage-infected E. coli TIP kinese remained stable for several months at 4°, the Landschutz ascites tumour enzyme was destroyed in a matter of hours at this temperature unless

the enzyme solutions were supplemented with TMP. The substrate specificity of the TMP kinase was not studied in the present investigation, partly because of possible phosphetese contamination, but mainly because of the low absolute activity of the enzyme. It is somewhat sobering to note that the TMP kinase studied by Bello and Besemen (1963a) had a specific activity in the crude extracts some ten times higher than that of the TMP kinese purified 1000-fold from cell-free extracts of Landschutz escites tumour cells. For similar reasons, the determination of K values for phosphoryl group donor and phosphoryl group acceptor must ewait further purification of the enzy e. but there is reason to believe that the TMP kinese shows absolute specificity for TMP (Weissman, Smellie and Paul, 1960; Hurwitz, 1959). In spite of the lack of kinetic information, the present instance appears to be the only TMP kinase from a mammalian source shown specifically to catalyse the formation of TDP from TMP, with the possible exception of the TMP kinese from calf thymus extracts (Bojarski, 1982) about which little is yet known.

The TDP kinese was not studied in detail but in so far as it was found to be extremely unstable (Sections 3.3.3 and 3.3.5), it resembles the enzyme from regenerating rat liver (Breitman, 1963) and that from extracts of Ehrlich escites tumour cells (Neissman, Smellie and Paul, 1960). The TDP kinese was further characterised by its relatively high remotion velocity in the crude cell-free extracts (Fig. 25). Nucleoside diphosphokineses are generally regarded as exhibiting wide substrate specificity as is illustrated by the recently purified and crystallised enzyme from Browers' yeast. This enzyme (Retliff, Neever, Lardy and Kuby, 1964) was shown to utilise as phosphoryl group donors ATP, GTP, ITP, GTP, dATP and dGTP with ADP as acceptor and GDP, IDP, GDP and UDP as acceptors with dATP and ATP as donors; TDP was not tested with the enzyme. While the existence of a substrate specific TDP kinase in the cell-free extracts of Landschutz ascites carcinome cannot be excluded, it seems rather unlikely in view of the large number of non-specific nucleoside diphosphokinases alluded to in extracts of memoralian, plant and becterial cell (Bessman, 1963).

The demonstration that the extracts of Landschutz ascites tumour cells contain three separable phosphokineses, each responsible for one step in the reaction sequence: TdR -> TMP -> TDP -> TTP does not of itself prove that this is the mechanism which obtains in the ascites tumour extracts or in vivo, it merely shows that all the enzymes required by such a mechanism ere present de facto while alternative machanisms are not precluded. Intensive purification and kinetic characterisation of the different kineses will be necessary in order to elucidate unambiguously the pethway for the formation of TTP from TdR. Nevertheless, the results given in Section 2.3 of kinetic experiments on the formation of TTP from TdR and TMP when augmented by the evidence presented on the separation of the TdR, TMP and TDP kinases and the purification of THP kinase (Section 3.3).make an impressive case for a mechanism such as that proposed by weissman, smallie and Paul (1960) that the phosphorylation of TdR proceeds in a stepwise manner through TMP. TDP and TTP, in that order. The mechanism most consistent with the present evidence is given in Fig. 41.

There remains the question as to which, if any, of the reaction steps

Fig. 41





Fig. 41. The mechanism of formation of thymidine 5'-triphosphate as suggested by experiments with enzymes from Landschutz ascites tumour cells.

> In the presence of the phosphoryl group donor, the phosphory lation of thymidine 5'-triphosphete precursors is essentially non-reversible and the over-all equilibrium fevours production of thymidine 5'-triphosphete. With orule extremts of Landschutz ascites tumour cells, the reverse reactions are cetalyzed by nucleotideses.

in the formation of TTP from TdR (or TMP) is likely to be concerned in the regulation of such synthesis. The present evidence (Section 2.3.2) clearly favours the reaction detalysed by the TMP kinese as the rate limiting step in TTP biosynthesis. One of the ways in which this regulation may operate has been suggested by the different pH optime of the TdR kinese and the overall reaction, on the one hand, and of TMP kinase, on the other (compare Figures 18, 31 (3) and 34). Somewhat similar differences in the pH optime of the TdR and the thymidylate kinases were observed in extracts of Hovikoff hepatoma (Ives, Norse and Potter, 1963). Conflicting evidence is available on this question elsewhere in the literature. Thus Bianchi (1962) has claimed that human tissues contain only small amounts of TdR kinase and that all human tissues convert TMP completely to TTP within 30 minutes; however, those experiments were carried out in the presence of an ATP regenerating system and with the tagit acceptance of the view that the phosphorylation of THP to TTP occurred by pyrophosphate addition and the results could thus be due to the presence of extremely high TDP kinase activities in the cell-free extracts of human louke mic tissue used. The conclusions of Bianchi (1962) were not supported by the experimants of Delemore and Prusoff (1962) in which several types of human leukaemic tissue were examined for their kinese activities. High TdR kinese activity was found by these workers in whole cells and the results were generally consistent with the possibility that the formation of TDP from TNP was rate limiting in the synthesis of TTP from TdR. If THP kinese were the egent responsible for limiting the rate of TTP synthesis then useful confirmation might be expected from the infection of resting cells

- 183 -

with DNA-containing viruses or as a result of a sudden change in memolian cell populations from a resting to a rapidly proliferating state (as in the regeneration of ret liver). As would be expected, an increased rate of production of TTP from TMP has been demonstrated as a result of these events. However, most of the reported investigations have neglected to take any account of intermediate reactions and in only one case has a detailed study of the relative activities of the TMP and TDP kinanes been conducted. The infection of E. coli with TE bacteriophage was shown by Bello and Bessman (1963a) to result in the induction of a new TMP kinese and the consequent rate of TTP production was greatly increased compared with that of uninfected bacteria while the TDP kinage activity remained undisturbed (Bello and Bessman, 1963b, yet the TMP kinase activity of the infected cells was only one-thirtieth of the TDP kinese activity of the uninfected cells. It seems possible to interpret these findings as an indication that high TDP kinase activities were already available in the uninfected calls while an increase in the TMP kinese activity was necessary to sustain the rate of TTP production required by the viral genome, thus also indicating that the TMP kinase activi' was rate limiting in the formation of TTP from TMP in the uninfected cells. The rate limiting effect of TMP kinese in TTP bicaynthesis may be a general phenomenon and if a regulating step were necessary in the formation of TTP from TdR or TMP, it is also the more likely in view of the generally accepted theory that de novo formation of thymidine nucleotides occurs by methylation of dump to TMP.

- 184 -

4.2. Intracellular location of the thymidine and thymidylate kinases

Gurrent concepts on the replication in vivo of DMA seems to imply that such synthesis is restricted to mites located within the nuclei. During the period of interphase when DMA synthesis has been demonstrated by autoradiographic techniques such gynthesis is localised and restricted to nuclear regions at a time when the nuclear membrane remains intext. However, recent studies on cellular morphology have shown that the nuclear membranes of several types of mammalian cells possess porce of measurable size, these findings being not inconsistent with the possibility that relatively large nolecules may traverse the membrane under various circumstances, Indeed, the Jacob-Monod concept of the genetic control of protein synthesis stipulates the passage of metabolically unstable "measurger" RMA from nuclei-localised DMA to cytoplesmic microscomes. In becterie, this type of transfer presents few structural problems as such organisms do not possess definable nuclear membrane structures while in membrane cells such transfer would be governed by the permeability of the membrane berrier.

The current concepts on the synthesis of DNA appear to imply, further, that the formation of decayribonucleoside 5'-triphosphates is a sufficient as well as a necessary condition for such synthesis. The two types of process may therefore be intimately connected in vivo and it seems conceivable that the decayribonucleotide kineses and the DNA polymerase may be structurall linked so as to allow the passage of precursor material through an integrated abain of reactions in the course of DNA replication. This may not represent a true in vivo situation but, in the absence of firm alternatives, it is intriguing to examine the available information on the intracellular location of the thymidine and thymidylate kinases in the light of the DNA polymerase distribution in mammalian cells.

In view of the relatively dependable evidence in favour of DNA synthesi being localised in conjunction with the chromosomes in the nucleus, it seems strikingly anomalous that the enzyme which is thought to be responsible for this process, DNA polymerase, should frequently be found in a highly active form in cytoplasmic extracts of mammalian cells (Smellie, Keir and Davidson, 1959; Bollum and Potter, 1959). One possible explanation for this observati was that this enzyme, as a nuclear constituent, had traversed the nuclear membrane in the course of the preparation of cytoplasmic extracts or that the isolation procedure employed sufficed to rupture a considerable proportion of auclear monbranes. The problem was investigated by Keir, Smellie and Siebert (1962) who compared the DMA nucleotidyltransferase activities in regenerating rat liver nuclei and cytoplasm isolated in non-equeous media. The findings were interpreted as an indication of the existence of DNA polymerase in both nuclei and cytoplasm, the cytoplasmic enzyme activity being augmented by DMA polymerase leaching from the nucleus when such cellular fractionations were performed in aqueous media. The extreme solubility of the DNA polymerese thus indicated (Siebert, 1963) makes it manacessary to stipulate a movement of DNA molecules back and forth from the nucleus for replicative DNA synthesis to occur. These conclusions were further amplified by the work of Smith and Keir (1963) who, working with celf thymus tissue, found that nuclei isolated in non-equeous solvents retained a DNA polymerase of high specific activity

while the corresponding cytoplasmic extracts also contained measurable DNA polymerase activity. Thus it seems probable that the capacity and apparatus for the synthesis of DNA is not restricted to nuclear regions of the cell and that a certain proportion of the cellular DNA polymerase activity occurs in cytoplasmic localities also.

The question then arises as to what function is served by the sytoplasmic DNA polymerase. At least two possible answers have recently presentthemselves. The first of these is the possible involvement of DNA polymereswith the synthesis of the cytoplasmic component of DNA which in memmelian cell apart from chick embryos, appears to be exclusively associated with mitochondrin (Mass and Mass, 1963s, b). Chevremont (1962) has presented sutorediographic and cytochemical evidence for the incorporation of (³H) TdR into mitochondrial structures of certain memmalian cells in tissue culture. However, as pointed out by Mass and Mass (1963b), the total amount of DMA-like material in the mitochondris of, for example, ret liver cells represents no more than 1 per cent of the total DNA content of those cells and, while a possible function for this component cennot be excluded, the present evidence appears to suggest that the mitochondrial DNA fibres are metabolically inert structures consistent with their being vesticial.

The second important possibility has suggested itself from the experiments of Littlefield, MoGovern and Margeson (1963) who studied the changes in the intracellular distribution of DNA polymerase during the cell cycle of mouse fibroblasts. The results of such studies indicated that a proportion of the polymerase activity present in the cell-free cytoplasmic extracts in the G₁ period, became associated with a particulate fraction when the cells entered the 3 period. The particulate fraction was presumably mainly nuclear and the observations thus suggest that cytoplasmic DNA polymerase enters the nucleus prior to, or in the course of, DNA synthesis, becomes particle associated and in this form catalyses DNA replication. However, as pointed out by Smith and Keir (1963), the interpretation of such results is made difficult by the presence in the homogenetes and cell-free extracts of several types of nucleases which may be expected to degrade and fragment the added DNA primer in verious ways and thus influence the measured polymerase activity.

While evidence regarding the intracellular location of the DNA polymerses is thus for from precise, even less is known about the location of the thymidine and thymidylate kineses. The present investigation provided little evidence with a bearing on this problem. Thus, while the kineses wer obtained from cell-free extracts of Landschutz assites tumour cells via a procedure which provided for the recovery of a minly sytoplasmic fraction, no rigorous precentions were taken to exclude possible comptic disruption of some nuclei. It seems clear also that all the reservations regarding the origin of the DNA polymerase in equeous sytoplasmic extracts (Keir, Smellic and Siebert, 1965; Smith and Keir, 1963) apply with equal force to the processibility that the kineses studied were originally derived from the nuclei and appeared in the sytoplasmic extracts by passage scross the nuclear membre on, therefore, not be excluded.

- 168 -

Although most of the evidence from other sources is of uncertain significance, several intriguing possibilities have emerged as to the location and function of the thymidine and thymidylate kinase system in mammalian cells. In most of the studies referred to above, the substrates for DNA polymerese ection were provided in the form of (32p) -labelled deoxyribonucleoside triphosphetes, thus making the presence of kinases an unnecessary condition for incorporation. Several other types of study have been reported, however, in which incorporation of radioactive label into abid insoluble polynucleotides was investigated by the use of (3H) TdR. Behki and Schneider (1963) have examined the distribution of DNA polymerase activity in normal and regenerating rat liver and in Novikoff hepatoma using a polymerase assay which thus depended on the presence of thymidine and thymidylete kineses, and have shown that incorporation of ("H) TdR was particularly effective with hepetome nuclei isolated in non-aqueous solvents. In the case of the other cell types examined, in vitro incorporation was mainly confined to the cytoplasmic fractions. However, regenerating rat liver nuclei were capable of catalysing the rapid incorporation of (32p) THP from (³²P) TTP into acid insoluble polynucleotide material, this observation being interpreted as evidence of the absence of TdR, TMP and TDP kinases from the nuclear regions of such cells. The cytoplasmic origin of the kinases in several other types of mammalian cell has been diroumstantially indicated. Of perticular interest are the experiments of Bach (1962) who allowed HeLa cells to grow in a medium containing (³H) TdR and found that the radioactivity became associated with a microsomal and a nuclear fraction, the label appearin

- 189 -

in the microsomes prior to its detection in the nuclei. Subsequent analysis showed that 10 per cent of the total DNA polymerase activity was associated with the microsomal fraction as measured by incorporation from (³H) doTP. While these experiments provided little direct evidence for the specific location of thymidine and thymidylate kineses, within the cytoplasm, the findings were consistent with the possibility that these enzymes also became associated with perticles at some stage of the cell cycle. A different line of evidence with a bearing on this question has been provided by Billen (1963) who isolated a DNA polymerage - DNA complex from the nucleoprotein component of 5. coli 15 TAU. This complex contained variable quantities of thymidine and thymidyleto kinases as shown by incorporation from (32 P) THP and (3H) TdR. Centrifugation of the crude complex in a sucrose gradient brought about apparent dissociation of the polymerase activity and TdR kinase activity while part of the TMP kinase activity remained with the complex thus suggestin a difference in the "complex affinity" of the two kinases. It is significent also that no dAMP, dGMP or dGMP kinese activities were found in association with the DNA polymerase - DNA complex. The binding of thymidydate kinase in this way is not inconsistent with the conclusion of Firshein (1963) that DNA polymerese may have a much higher affinity for TTP than for dATP, dGTP or doTP. Another type of evidence suggesting that thymidylate kinase may occur bound to cytoplasmic perticles was provided by studies on mouse liver homogenates (Kielley, 1963a). While a considerable proportion of the TMP kinase activity of such homogenates appeared in a high-speed appernatant fraction, the resulting residue fraction upon freezing and thaning several

-5.0

- 190 -

times released TMP kinase activity representing three times the total activity in the original supernatant fraction. It was observed, moreover, that the total TMP kinase activity thus recovered was nearly equal to that found in cell-free extracts of mouse ascites hepetome. This evidence appears to suggest that part of the TMP kinese activity is structurally bound in an inactive form in non-proliferating tissues while in proliferating cells this component is released into the cytoplasm. A somewhat similar conclusion was reached by Bianchi, Brathorn and Shooter (1962) who studied the appearance of the thymidine and thymidylate kinases in vivo in rat liver following partial hepetectomy. If this were true, it should be possible to argue that the transformation of TMP kinase from a bound, inactive form to a soluble, active form occurs in response to specific events in the cell cycle, presumably at the time of DNA synthesis. How, thereafter, the THP and TDP kinases co-operate (with TdR kinase ?) in the formation of TTP is largely a matter of speculation, but the sharp increase in the concentration of intranuclear decxyribonucleotides which occurs in regenerating rat liver et the time of DNA synthesis (Behki and Schneider, 1962) together with the absence of thymidine and thymidylate kinase activities within such muclei (Behki and Schneider, 1963) suggest that TTP may be produced in the cytoplasm and be subsequently transported to intranuclear sites of active DNA synthesis.

Such speculations, however, remain ebstract in the face of the deficiencies in present day cell fractionation techniques and until wastly improved methods become available, there is little prospect of resolving problems of specific intracellular enzyme location in an unequivocal menner.

- 191 -

At present, relatively reliable information on the intracellular location of enzymes is confined to cases where electron microscopy or sutoradiography on be successfully applied.

4.3. The possible involvement of the thrmidine and thrmidylate kineses in the control and regulation of DNA synthesis

The biosynthesis of DNA appears to be a process governed by two different types of control (see Section 1.4), one temporal in concept (Fig. 9) and the other quantitative. Several lines of evidence now available indicate (Lark, 1963) that each type of control operates in response to a different set of circumstances and that it may, therefore, be convenient to separate them sonceptually. In the following discussion, it is proposed to refer to the temporal type as the control of DNA synthesis and to the quantitative type as the regulation of DNA synthesis.

That the formation of TTP may be rate limiting in DNA synthesis has been suggested by several studies in which the efficiency of endogenous decoxyriconucleotide kinese systems were compared in terms of the rates of formation of dATP, dGTP, dGTP and TTP. The findings have revealed striking differences in the ability of the decoxyribonucleotide kineses to phosphorylate dAMP, dGMP, dGMP and TMP, particularly when such studies were conducted on mammalian cells. Thus in normal rat liver, capability to phosphorylate TMP was found to be virtually absent (Ganellakis, Jaffe, Manatevinos and Krakow, 1959; Gray <u>et al.</u>, 1960) in agreement with the present findings (see Table 4) while the same tissue was found capable of efficient production of dATP, dGTP and doTP. Following-partial hepatestomy in the rat, the regenerating tissue was found to contain increased TMP phosphorylating activity whereas the decxyadenylate, decxyguanylate and decxycytidylate kinase activities remined unchanged relative to the activities found in the resting organ (Genellakie, Jaffe, Manstevinos and Krakow, 1959; Gray et al., 1960). Increases in the activity of TAR kinese has also Ween found to be associated with the onset of DNA synthesis in regenerating rat liver (Weissman, Smellie and Paul, 1960; Belts, 1968), the elevation of TdR kinese activity being unaccompanied by any increases in the activities of AdR and GdR kineses (Belts, 1962). The other important point that emerged from these studies was that the activity of the thymidylate kineses remained low, even in rapidly dividing cells, when compared with the enzyme systems catalysing the formation of dATP, dGTP and dGTP. The evidence is thus clearly consistent with the view that TMP phosphorylating activity increases in synchrony with the mitotic index while still remaining effectively rate limiting in providing the ultime presurgors for DWA replication. A similar relationship in the phosphorylating activities acting on dAMP, dGMP and dGMP, on the one hand, and on TMP on the other, has been observed in mouse liver homogenates (Eielley, 1961) and in extracts of Hels cells (Magee, 1962). The relationship does not appear to be universal since studies of this type carried out using bacterial extracts (Besemen, 1959) have indicated that formation of the four deckyribonucleoside 5' -triphosphetes proceeds at approximately the same rate.

In the experiments of Gray <u>et al.</u>, (1960), the failure of extracts of normal rat liver to catalyse the phosphorylation of TMP and the inhibition
by normal liver extracts of the synthesis of TTP by extracts of Ehrlich escites tumour cells were attributed to a non-dielysable, heat sensitive factor present in the liver extracts. These observations were confirmed in the present experiments (Section 3.3.1) and shown to be directed specifically towards the inhibition of the phosphorylation of TdR. No effect was observed on the phosphorylation of GdR (Table 5). The experiments of Gray et al. (1960) showed, moreover, that the rate of phosphorylation of dAMP. dGMP and dGMP by extracts of ascites tumpur cells remained unaffected following the addition of normal rete liver extract and, while these findings were consistent with the view that the inhibition of the formation of TTP from TMP was caused by specific thymidine nucleotidases, the evidence did not permit precise interpretation. An interesting correlation between decry. ribonucleotide kinese activities and dephosphorylating activities in mouse liver and mouse ascites hepetome has recently been provided by Kielley (1963b, This author confirmed the earlier conclusions (Manstevinos and Genellakis, 1959; Ganellakis, Jaffe, Manstavinos and Krakow, 1959; Gray et el., 1960) that non-proliferating tissues contained active kinases transforming dAMP, down and down into the corresponding triphosphetes while being deficient in demonstrable phosphorylation of TMP. With broken cell preparations of hepetoma, the thymidylate kinese activity was shown to be higher than in resting tissue while the decryoytidylate kinase activity was markedly depressed in such a way that TMP kinase still appeared to be effectively rate limiting in the provision of substrates for the DNA polymersse. Measurements of nucleotidase activity indicated that the low TMP kinase activity of normal

- 194 -

mouse liver tissue could to some extent be accounted for by the high THP nucleotidase of that tissue while, in contrast, little or no degradation of the other decryribonucleoside monophosphates was observed. This was taken to be compatible with the findings (Fiele, Fiele, Tobar and MoQuille, 1962, that The nucleotidase resides primerily in the microsomes while the nucleotideses of dAMP, dOMP and dOMP are restricted to mitochondrie where they are latent (Kielley, 1961). The phosphorylation of THP by broken cell preparations of mouse ascites hepatoma (Kielley, 1963b) was characterised by the absence of degradative reactions. However, the initial kinetics of THP utilisation by mouse liver preparations did not indicate that TMP kinase was inherently deficient or lacking in that tissue. This conclusion was cenfirmed by the demonstration that potential thymidylate kinase activities in mouse liver were as high as those of liver assites hepatoms and that a major part of the enzyme in liver was "particle-bound" (Kielley, 1963a). It seems possible that these observations may be related to the increased stability bestowed upon TMP kinnse by its substrate (Section 3.3.4) which event, according to Histt and Bojarski (1961), may provide a means of regulating THP kinese activity in vivo. In their view, THP kinese is continuousl formed in normal liver colls but is inactivated from lack of substrate which normally stabilises the enzyme. Neither Kielley (1963b) using mouse liver nor Weissmon, Smellie and Paul (1960) using rat liver were able to demonstrate the induction of thymidylate kinase following intraperitones1 injection of TdR as suggested by Hiatt and Bojerski (1960) and it seems possible that the stabilisation of thymidylate kinase by TMP may operate in vitro to compensate

- 195 -

for the loss of some other stabilising agent present within the cell.

In addition to the evidence outlined above, there is a considerable weight of evidence from virus infected systems consistent with the view that the activity of the thymidine and thymidylate kinese system is effectively rate limiting in DNA synthesis. In most of the instances quoted below, increased production of TTP (with a consequent increase in the rate of DMA synthesis) was the result of true induction of one or more of the kinase components perticipating in the formation of TTP from TdR. However, as pointed out by Joklik (1962), the more demonstration of an increased concentration of enzyme molecules as a result of viral infection need not imply a general deficiency of that enzyme, only that the required activity was not available in the relatively small region of the cell where replication of viral DN: is to be initiated. Elevation of thymidylate kinase activity has been demonstrated following infection of Hele cells with vaccinis virus in which case this kinase activity was increased without comparable increases in dAMP, dOMP and GGMP kinase activities (Magee, 1962), neither was there evidence of an increase in deoxyribonucleoside kineses. Similarly, following infection of rabbit kidney cells with pseudorables virus, a new thymidylate kinese appeared to be induced resulting in a 6-fold increase in activity over the thymidylate kinese of uninfected cells, while no increases were detected in other deoxyribonucleotide or deoxyribonucleoside kinases (Nohars and Kaplan, 1963). The induction of a new thymidylate kinase in E. coli following infection with the T2 becteriophage (Bello, van Bibber and Beessen, 1961; Bello and Bessman, 1963a) has been described elsewhere (Section 4.1) In that

- 196 -

particular case, however, the TMP kinase is thought to perform a different role from that normally associated with thymidylate kinases in that the newly formed enzyme, following purification, exhibited equal affinity for TMP, dGMP and 5-MeOH-dGMP while the normal TMP kinase was specific for TMP.

The induction of new, active forms of thymidylate kinase would be expected to result in an increase in the efficiency of the <u>de novo</u> pathway leading to TTP, this pathway being dependent of the methylation of dUMP to TMP with subsequent phosphorylation of TMP to TTP (see Fig. 11). However, several instances are on record also of the induction of TdR kinase following viral infection. The over-all effect of such induction would be to elevate the levels of TMP available through the preformed pathway, thus presumbly indicating that production of TMP by methylation of dUMP was insufficient in such cell populations to sustain the required rate of TTP (and DNA) synthesis.

The induction of TdR kinese in Hele cells following infection with the virus of cowpox (McAusland and Joklik, 1968; McAuslan, 1963) was found to be under the control of the viral genome as were the TdR kineses induced in mouse fibroblast cells by infection with wascinia virus (Kit, Piekarski and Dubbs, 1963) or with herpes simplex virus (Kit and Dubbs, 1963). It was found with interest that UdR kinese was simultaneously induced, thus indicating that the newly formed TdR kineses were difunctional as has been demonstrated with the TdR kinese of §.coli (Okasaki and Kornberg, 1964a). In none of the systems quoted above was the induction of TdR kinese accompanied by increases in TMP kinese activity. The weight of evidence and the large number of instances in which induction of TdR kinese has been demonstrated makes the

- 197 -

traditional concept that the phosphorylation of TdR represents a preformed or "salwage" pathway appear decidedly tenuous. Instead, it may well be that some hitherto unknown mechanism allows TdR kinase to participate in a de novo pathway for the synthesis of TTP.

No instances are on record of the induction of a nucleoside diphosphokinese following viral infection of mammalian cells, and the failure to detect an increase in this enzyme may perhaps be taken as enother indication of the highly notive and non-specific nature of nucleoside diphosphokinases in uninfected cells (see Sections 3.1 and 4.1).

All these observations may be regarded as the consequence of control rather than the manifestation of a control mechanism per so. The possibility connot be lightly dismissed, however, that true control or part of the regulation of DNA synthesis in a given cell population resides in the ability of the thymidine and thymidylate kinases of those cells to provide TTP. It is of considerable interest, therefore, to examine in more detail the potentialities of the formation of TTP from its precursors TdR and TMP as a regulator system.

In the view of Devidson (1962), DNA synthesis occurs at oritical concentrations of the decryribonucleoside 5'-triphosphates which are themselves components of a homeostatic mechanism. Pertinent features of such a mechanism have become apparent over recent years and the picture which is emerging is of startling complexity comprising several interconnected regulatory devices, the combined effect of which is to provide sephisticated and flexible control of internucleotide metabolism.

One type of model regulatory mechanism which has assumed particular importance in defining the balance between synthesis and degredation of nucleotides is that termed negative feedback inhibition (see Section 1.4). By this term is meant a situation in which accumulation of an end metabolite causes a specific inhibition of enzymes catalysing an early step in the metholic pathway leading to its production. The first indications that such devices were involved in nucleotide metroolism were given by Yates and Pardee (1956b) who discovered that the first enzyme specific to pyrimidine synthesis de novo, aspertate transcarbamylase, was competitively inhibited by STP, one of its distal products. In the subsequent years, there were described a large number of other negative feedback mechanisms which were shown to intervene at several points in internucleotide metabolism, in general and at critical stages of de novo and preformed pathways leading to decryribonucleoside triphosphates, in particular. Some of the relevant negative feedback mechanisms are given in Fig. 42 from which can be deduced, among other things, the extent to which the de novo pathways for the synthesis of purine (Nyndgaarden and Ashton, 1959; Henderson, 1962) and pyrimidine (Yates and Pardee, 1956s, b; Gerbart and Pardee, 1962) ribonucleotides are specifically regulated by distal metabolites in the form of ribonucleoside monophosphetes and triphosphetes. Among the purine ribonucleotides, several additional feedback mechanisms appear to regulate the cycle of interconversions between IMP, GMP and XMP while histidine inhibits competitively the first enzyme specific to its synthesis, phosphoribosyl-ATP pyrophosphorylase (Magasanik and Karibian, 1960; Ames, Hartman and Jacob, 1963,.

- 199 -



71g. 42. Regulation of internucleotide metabolism by negative feedback control.

> Thick arrows indicate negative feedback control and the thin broken lines denote specific reactions so controlled Only relevant intermediates are included. (For more detailed schemes of internucleotide conversions, see Figures 10 and 11).

> Three types of negative feedback inhibition mechanism may be recognised subject to the same molecular principles:-(a) Feedback inhibitions which regulate the activity of the first enzymes involved in purine and pyrimidine biosynthesis.(b) Feedback inhibitions by decayribonucleoside triphosphetes of the reluctive sequences leading to decayribonucleoside diphosphetes. (c) Feedback inhibition of enzymes involved in the utilization of preformed nucleosides for nucleic acid synthesis (in addition to the mechanisms illustrated, dGTP has been found to exert negative feedback control of 3dR kinnes (Maley and Maley, 1968b)). All negative feedback inhibition mechanisms, apart from those included in the last category, regulate essentially irreversible conversions.

IR-AMP = phosphoribosyl-l-adenosine 5'-monophosphete
IGP : imidszolyigiycerol phosphete
Carbamyl-P = carbamyl phosphete.

While these mechanisms strongly influence the performance of the de novo pethways for the synthesis of ribonucleotides, a further set of feedback mechanisms is directly concerned in regulating the supply of decorribonucleoside 5' -triphosphetes. This regulation is directed towards several critical stages in the passage of metabolites, derived from the ribonucleotid pool, through the series of reactions leading, ultimately, to the substrates for the DNA polymerase (see Figures 10 and 11). One crucial step which involves the departure of metabolites from the ribonucleotide pool is the reductive mechaniam for the conversion of ribonucleotides to decayribonucleotides and which, according to present evidence, appears to take place exclusively at the diphosphete level (Reichard, 1962; Moore and Hurlbert, 1962; Bertani, Engemark and Reichard, 1963; Larsson, 1963,. In vitro studies conducted by Reichard, Ganellekis and Janellakis (1961) with extracts of chick embryos have indicated that such reductive mechanisms may be governed by a singular type of negative feedback inhibition. Of the reducti systems studied, that concerned with the conversion of GDP to dGDP was observed to be inhibited by extremely low concentrations of dATP while that reducing SDP to dODP was strongly inhibited by low concentrations of dATP, dorp, and, to a lesser extent, of TTP but not by dorp. In these experiments, the interplay between the reductive mechanism and DNA synthesis was examined as a function of incorporation of (32p) from (32p) GMP and it became apparent that a sensitive equilibrium existed between the two processes wherein the optimal requirements of one was contradictory to the optimal requirements of the other. Such feedback inhibition may have widespread significance in the regulation of DNA synthesis and it has been connected with the inhibition of DNA synthesis in vivo following the addition of high concentrations of thymidine to tissue culture media and also appears to explain how OdR could relieve systems so inhibited (Morris and Fischer, 1960, 1963). A similar situation appears to obtain in certain bacteria (Lark, 1960).

Experiments on whole Shrlich excites cells have shown that whereas AdR inhibits DNA synthesis and leads to accumulation of dATP, this inhibition may be overcome by addition of GdR (Lenger and Klenow, 1960). The type of inhibitory mechanism involved may be related to the negative feedback inhibits by decoxyribonucleoside triphosphates of the reductive sequence leading to dQDE observed in cell-free systems (Reichard, Ganellakis and Ganellakis, 1960; Morris, Reichard and Fischer, 1963). Nevertheless, inhibition of DNA synthes by AdR has as yet only been observed in media containing very high concentration of decoxyribonucleoside (Klenow, 1968). The fact that accumulation of purine decoxyribonucleoside (Klenow, 1968). The fact that accumulation of purine decoxyribonucleoside (Isanow, 1968). The fact that accumulation of purine decoxyribonucleoside (Isanow, 1968). The fact that accumulation of purine decoxyribonucleoside (Isanow, 1968). The fact that accumulation of purine decoxyribonucleoside (Isanow, 1968). The fact that accumulation of purine decoxyribonucleoside does not appear to occur in normally growing cell populations (Tark, 1963) makes a direct relationship between the two mechanism appear tenuous.

A further indication that the thymidine nucleotides exert an influence over the reactions leading to their formation was obtained by Maley and Maley (1962b, 1963) who discovered that the demmination of dGMP with minoss prepared from rat embryos and chick embryos was strongly inhibited by TTP. Several other important observations were made in conjunction with this discovery. Thus, in experiments on the utilization of GdR by chick embryo minos, dGTP was observed to regulate the activity of DdR kinase (Maley and Maley, 1963) in a manner reminiscent of negative feedback inhibition (Gerhart and Pariee, 1962) while in the presence of TTP, the inhibition of dCMP deeminase was accompanied by a stimulation of OdR kinase activity. In rat embryo mince (Maley and Maley, 1963b), TTP was observed also to inhibit the phosphorylation of UdR. The over-all effect of these interlaced regulatory mechanisms appeared to suggest that TTP might control its own synthesis by regulating the contribution of OdR (and decayoytidine nucleotides) to the thymidine nucleotide pool.

Norking on the assumption that none of the mechanisms given above could provide effective control of TTP synthesis in the presence of a substantial pool of thymidine, Ives, Morse and Potter (1962, 1963) examined the phosphorylation of TdR by cell-free extracts of Nevikoff hepatoms and discovered that TTP exerted negative feedback control of TdR kinese activity. Such inhibition was demonstrated both by endogenously produced TTP (from (¹⁴G) TdR) and by added TTP, neither TMP or TDP being effective. In contrast, thymidylate kinese was not inhibited by TTP under the conditions used, and the inhibition thus seemed to be specifically associated with the competitive inhibition by TTP of the first enzyme (see Section 4.1) in the metabolic pethway leading to its formation. A similar type of regulatory mechanism may operate in regenerating ret liver, from which a partially purified TdR kinese has been shown to be sensitive to added TTP or TDP (Breitman, 1968).

By far the most exhaustive study of the inhibitory effect of TTP on the phosphorylation of TdR has been that conducted by Okazaki and Kornberg (1964b). The use of a highly purified TdR kinase preparation from <u>E. coli</u> (Okazaki and Kornberg, 1964a) in this study permitted kinetic elucidation of many features of the inhibitory mechanism involved.

Thus it was observed (Okazaki and Kornberg, 1964b) that the activity of TdR kinese was specifically ishibited by TTP, THP or TDP being without effect. Several additional properties of the enzyme itself were examined and it was found to entalyse the phosphorylation of UdR at approximately the same rate as TdR, thus confirming an earlier report (Skold, 1960b) that TdR kinase was bi-functional in action. The inhibition of UdR kinase by TTP as observed by Maley and Maley (1962b) was thus also confirmed. An extensive study (Okesaki and Kornberg, 1964b) was made of the acceptor and phosphoryl group donor specificities of the E.coli TdR kinase and such studies revealed that certain deoxyribonucleoside di- and triphosphetos (dGDP, GDP, dATP, dGTP and dCDP) acted as activators, some triphosphates (dATP, dGTP) as both donors and activators and ATP as donor only. It appeared further that inhibition with TTP was not competitive with the phosphoryl group donor ATP. in contrast to the conclusion of Ives, Norse and Potter (1963), or with the activator doDP, but rather with the substrate TdR. Okasaki and Kornberg (1964b) conclude, however, that TTP does not bind to the catalytic site for TdR, on the gounds that no inhibition was observed with TMP or TDP, but rather to an adjacent site close enough to compete sterically with the binding of TdR to its site. This conclusion is remarkably close to that of Gerhart and Pardee (1962) for the molecular mechanism of the feedback inhibition of aspartate cerbanylase by GTP, and it seems quite possible that the provision of a so-called "allosteric site" (Monod, Ghangeux and Jacob, 1963) may have general significance in the regulation of intracellular enzyme systems. In

addition to this property (Okazaki and Kornberg, 1964b), it appeared that the TdR kinese bound, and had, in the absence of activator, sites for 2 molecules of ATP while in the presence of the activator, dGDP, only one molecule of ATP was required, thereby suggesting that the second site might serve an activator function.

The biological significance of this kinetic date, in the view of Okazaki and Kornberg (1964b), lies in the fact that TdR kinese appears to be activated by those decoxyribonucleoside di- and triphosphates which accumulate when the supply to TTP is insufficient and that the activation may operate in effect by lowering the $K_{\rm m}$ of the enzyme for TdR thus favouring its efficien utilisation at low concentrations.

While TdR kinese might thus perticipate in an extremely powerful regulating mechanism governing the production of decoxyribonucleoside 5'triphosphates, it is not yet possible to discover to what extent its potential is fulfilled in the intext cell. Nor is it possible to gamge its significance relative to other negative feedback systems operating as regulators in the production of substrates for the DNA polymerase until those systems have been similarly scrutinised. However, even when solutions have been found to these questions, there remains the essential enigms of the function of the TdR kinase itself. The mejority of the available evidence suggests that TdR is not on the direct pathway for the formation of TTP de novo, but that this pathway utilises instead the methylation of dUMP as an entrance to the thymidime nucleotide pool. The TMP thus formed is presumed to be phosphorylated directly to thymidine polyphosphates followed by the utilisation of

these compounds for DMA synthesis. The smallness of the intracellular thymidine pool (Lark, 1963), when considered together with the extremely poor utilisation of thymine in mammalian cells, suggests that if the TdR kinese functions in its treditional role, as part of a preformed pathway, its contribution to the thymidine nucleotide pool will be insignificent and insufficient to sustain the rate of TTP production required in DNA synthesis. Recently, a mechanism has been suggested which allows the participation of TaR in a normal pethway for producing TTP de novo. By incubating Ehrlich ascites tumour cells in a medium containing (32) -labelled inorganic phosphate, Langen and Lias (1962, 1963) observed that the specific activity of the DNA-thysidylic acid was considerably higher than that in the other decxyribonucleoside monophosphetes recovered from the DNA and they conclude from this that dilution of radioectivity was the less pronounced in the thymidine nucleotide pool and that this constituted the emaller of the decxyribonucleotide pools. However, they (Langen and Liss, 1962) further observed that inhibition of DNA synthesis by fluorourseil (which is presumed to inhibit specifically thymidylate synthetese, could be reversed by adding TdR to the medium. Addition of (38p) -labelled inorganic phosphete to a medium so composed gave DNA-decxyribonucleoside monophosphates of relative specific activities similar to those obtained with cells to which no fluoroure cil or TdR had been added. Langen and Liss (1963) argued from this that TdR itself may be a normal metabolite involved in de novo synthesis of TTP and they visualize a two-component system consisting of TdR kinese in association with TMP nucleotidase as the connecting link with the thyaidine

nucleotides produced via dUMP. The westage of energy resulting from the operation of a "salwage loop" of this type has been pointed out by Brumm, Potter and Biekevitz, (1956) and by Beltz, (1962), but some alternative mechan may well exist which allows the TdR kinnse to realise to the full extent its remarkable potential as a regulator of decayribonucleotide metabolism.

While the thymidine nucleotides are thus intimately concerned in regulating the formation of the ultimate substrates for the DNA polymerase, few of the studies quoted so far have a direct bearing on whether they also contribute to the temporal control of DNA synthesis. In other words, are the thymidine nucleotides concerned with the event (or events) which constitute the initiation of DNA synthesis? Yow attempts have been made to study the alterations in intracellular ensyme patterns in relation to the DNA synthesis cycle in manualian cells. In bacteriophage-infected bacterie, however, <u>de novo</u> synthesis of the various ensymes necessary for the production of specific types of DNA has been demonstrated (see Sections 2.1 (b) and 4.1)

Best known by far are the changes that occur in deczyribonucleotide metabolism following partial hepatectomy in the rat, several gross results of which have been noted above (see p. 193). Thus, increases in several of the enzymes participating in the formation of decxyribonucleoside 5'-triphosphates in rat liver have been demonstrated following such surgical intervention. It has been shown, moreover, that such increases in enzyme activity are discontinuous and linked specifically to events in the DNA synthesis cycle, more particularly, such elevation in synthetic activity has been observed to occur just prior to, or to coincide with, the onset of DNA synthesis. The experiments of Maley and Maley (1960) showed that dCMP deaminase activity, which is very low in normal rat liver (Maley and Maley, 1961b), began to increase 12 hours post partial hepatectomy, followed at 18 hours by demonstrable thymidylate synthetase activity at which time there was also evidence of DNA synthesis. To account for these observations, Maley and Maley (1960) suggested the possibility of a sequential induction of dGMP desminase and thymidylete synthetese, with the desminase becoming elevated prior to and the synthetese concomitant with DNA synthesis. It was suggested further that the product of dGMP desminase action, dUMP, might induce the formation of TMP synthetase, and that the product of this reaction (TMP) might, in its turn, induce the formation of TMP kinase. These suggestions correlate well with findings concerning the appearance of the thymidine and thymidylate kinases during regeneration of rat liver following partial hepatectomy. Thus, Bollum and Potter (1959) found that the highest uptake of (3H) TdR into DNA occurred between 24 and 60 hours following the operation and the experiments of Weissman, Smellie and Paul (1960) revealed elevation in the levels of these enzymes at 20 - 24 hours post partial hepsteetomy, just prior to the onset of DNA synthesis, the TdR, TMP and TDP kinases appearing sequentially, in that order. This timing of the appearance of thymidylete kinese was confirmed by Histt end Bojarski (1960) who observed an increase in the activity of this enzyme at 20 hours post partial hepetectomy. seisemen, Smellie and Paul (1960) suggest that since the increases in activity of these ensymes may only be observed during the growth phase, a sequential induction of the TdR. TMP and TDP kinase might be involved and

- 207 -

it seemed possible also that elevation of each kinase occurred in response to production of its substrate. This suggestion was clearly consistent with the experiments of Histt and Bojerski (1960) which demonstrated the induction of thymidylate kinese in ret liver following intraperitoneel injection of TdR. However, if the production of the thymidine and thymidyle kinases occurs in response to the endogenous formation of their respective substrates, and if this were a necessary as well as a sufficient condition for such increases, it seems logical to conclude with Histt and Bojerski (1960, 1961) that the metabolic event initiating the train of reactions leading to the synthesis of DNA must be manifest at a stage prior to the phosphorylation of TMP. Thus, it would appear that a metabolic pathway is involved proceeding from dGNP and TdR through TMP and TTP to DNA and that this sequence of reactions may be normally repressed. No evidence has yet appeared to show that this type of induction is due to de novo production of enzyme molecules, but rather that enzyme adaptation or positive feedback (see Section 1.4) may be the process involved. The molecular mechanisms which allow the cyclic appearance of thymidine kinase at the end of G periods (Fig. 9) (Hotta and Stern, 1963) may be directly related to the particle-bound form in which such enzymes apparently cocur in resting tissues (Kielley, 1963a; Bianchi, Grathorn and Shooter, 1962). It seems possible to conceive of a mechanism whereby relevant kinases might be released from en inactive, particle-bound complex specifically in response to the accumulation of the appropriate substrate and return to this inactive, latent state following the completion of the S period. Such activation-inactivation

+ 808 -

ayoles, apparently, are not uncommon in mammalian cells (Perdee and Milson, 1963) and suggest possibilities for control quite analogous to induction and repression.

Gurrent concepts on the control of the initiation of DNA synthesis (Tark, 1963; Mamlpe, 1963) do not visualise the thymidine and thymidylate kinnse system as an essential component, but rather that both RNA and protein synthesis are required prior to DNA replication. Moreover, much of the evidence suggests that the controlling factor may be of a structural rather then a catalytic nature and that it may operate, not at the nucleotide level but at the level of the genome itself.

Considerations such as these are somewhat outside the scope of the present work, the object of which was manifestly not to elucidate how intracellular events are regulated along the time axis of the cell's life, but merely to indicate, through the <u>in vitro</u> system, which events in the formatic of TTP could conceivedly provide a regulating step in the synthesis of that compound and thus, in a wider context, influence DNA synthesis. The interpretations presented here can at best be no more than molecular concepts reflecting the potentiality of the ansyme system rather than the manner in which it operates within the intext cell. With these reservations in mind, the results presented are clearly consistent with the view that the thymidine and thymidylate kinase system forms an essential and integral part of the regulatory mechanisms which control the quantitative aspects of DNA replicati

- 209 -

SUMMARY

- 1. The phosphorylation of thymidine by ensymes derived from soluble extracts of Landschuts ascites tumour cells led to the formation of thymidine 5'-memophosphate (TMP), thymidine 5'-diphosphate (TDP) and thymidine 5'-triphosphate (TTP). The formation of TTP from thymidine with ensymes from this source was optimal at pH 7.8 to 8.1 and showed an absolute requirement for Mg^{**} and adenosine 5'-triphosphate (ATP) wit the maximal rate of TTP formation occuring in the presence of 5×10^{-3}] ATP and 5×10^{-3} <u>H</u> MgCl₂.
- 2. The mechanism of formation of TTF from thymidine by soluble extracts of Landschuts ascites tumour cells has been examined using an improved meth for the complete separation of thymidine nucleotides on columns of ECTEDLA cellulose.
- 3. Studies on the time course of formation of TTP from $({}^{3}H)$ -labelled thymidine have invariably shown that TMP and TDP may be detected in the reaction mixtures very soon after the start of the incubation. On no occasion was labelled triphosphate detected earlier than labelled diphosphate. In experiments with thymidine $5^{i}-({}^{32}P)$ -monophosphate, labell diphosphate was again detected before labelled triphosphate.
- When experiments were performed with TDP labelled in either the α- or the β- phosphates with (³²P), TTP was formed very rapidly without the intermediate formation of TMP or any higher phosphates of thymidine than TTP.
 It was observed that TMP and TTP were formed very rapidly from thymidine and that TTP was formed very rapidly from TDP in which circumstances it

seems possible that the formation of TDP is the rate-limiting step in the synthesis of TTP, thus also suggesting the involvement of thymidine kinase, TMP kinase and TDP kinase.

6. Extrasts of normal rat liver were shown to phosphorylate thymidine only at 5 per cent the rate of Landschuts ascites tumour extrasts. THP was not phosphorylated by the liver extracts while decoxycytidine was transformed into phosphorylated derivatives six times the rate with thymidin Inhibition by normal liver extracts of the synthesis of TTP from thymidine and TMP by extracts of Landschuts ascites tumour cells was obser to occur in a manner which suggested specific competitive inhibition of thymidine and thymidylate phosphokinases since no inhibitory effect was observed on the phosphorylation of decxycytidine.

- 7. An investigation was carried out in order to separate and characterise the three phosphokinases in extracts of Landschutz ascites tumour cells A preparation with thymidine kinase alone was obtained by treating orud extracts with dilute acid when TMP kinase and TDP kinase were lost pregressively over a period of 30 minutes. The thymidine kinase so obtain was purified 3 to 4-fold from the orude extracts; it catalysed the phos phorylation of thymidine to TMP with ATP as phosphoryl group donor and optimal activity was obtained in presence of $5 \times 10^{-3} \text{ M HgCl}_2$ and at pH 7.8 to 8.1.
- 8. If fractionation of the crude extracts with dilute acid was performed in the presence of excess TMP, 80 to 150 per cent of the initial TMP kinas (purified 2 to 3-fold) activity was recovered in the supernatant fraction (AS Fraction) following acidification to pH 4.7 while much of the TDP

kinase activity was destroyed. Phosphorylation of THP by this fraction had a narrow pH optimum at 7.6 and an absolute requirement for ATP $(5 \times 10^{-3} \text{ H})$ and Hg⁺⁺ $(5 \times 10^{-3} \text{ H or } 20 \times 10^{-3} \text{ H})$.

- 9. Treatment of the AS Fraction at pH 7.6 with increasing concentrations of Cy-alumina gel led to adsorption of TMP kinase along with much of the protein. If encess TMP was added to the AS Fraction prior to adsorption of the aluming gel, however, most of the protein was adsorbed while 60 to 80 per cent of the TMP kinase remained in solution. Furthe purification on a column of Sephadex G-100 yielded TMP kinase purified 1000-fold. This preparation was devoid of TDP kinase activity but was contaminated with adenosine 5¹-triphosphatase.
- 10. Fractionation of the AS Fraction directly on columns of Sephadex G-100 by gradient elution with phosphate buffer, pH 7.6, yielded three peaks of kinase activity. The first represented thymidine kinase, the second TDP kinase contaminated with adenosine triphosphatase and a small amoun of TMP kinase which was accounted for by slight overlapping of the thir peak of TMP kinase. This third fraction contained TMP kinase purified 60-fold and was shown to catalyse the formation of TDP from TMP. TTP was not detected in the reaction products and it was shown, further, th this failure to detect TTP was not due the presence of 1TF dephosphoryl ating engymes.
- 11. These results pointed to the existence of at least three phosphokinases in extracts of Landschuts ascites tumour cells and strongly support the view that TDP is an intermediate in the formation of TTP from thymidine and TMP in this system.

REFERENCES.

Abrams, R. & Bently, M. (1955a) Arch. Blochem. Blophys., <u>56</u>, 184. Abrams, R. & Bently, N. (1955b) J. Amer. chem. Soc., <u>77</u>, 4179.

Abrams, R., Edmonds, M. & Biswas, R.B. (1962) in "Acides Ribonucléiques et Polyphosphates. Structure, Synthèse et Fonction", vol. 106, p. 323. Colloq. intern. Centre Natl. Recherche Sci. (Strasbeurg).

Abrams, R., Libenson, L. & Edmonds, H. (1960)

Biochin, biophys. Antr, Res. Commun., 3, 272.

Adler, J., Lehman, I.R., Bessman, M.J., Simms, E.S. & Kornberg, A. (1958) Proc. mat. Acad. Sci., Wash., 44, 641.

Adler, M., Weissman, B. & Gutman, A.B. (1958) J. biol. Chem., 230, 717. Agranoff, B.W. & Brady, R.O. (1956) J. biol. Chem., 219, 221. Allende, C. & Backer, E. (1963) Biochim. biophys. Acta, 72, 10. Ames, B.N., Hartman, P.E. & Jacob, F. (1963) J. molec. Biol., 7, 23. Ames, H. & Magasanik, B. (1957) J. biol. Chem., 229, 653. Aposhian, H.V. & Kormberg, A. (1962) J. biol. Chem., 237, 519. Astrachan, L. & Volkin, E. (1958) Biochim. biophys. Acta, 29, 536. Avery, O.T., MacLeod, C.H. & McCarty, H. (1944) J. Exp. Med., 79, 137. Bach, M.E. (1962) Proc. nat. Acad. Sci., Vash., 48, 1031. Baldwin, R.L. & Sheoter, E.M. (1963) J. molec. Biol., 7, 511. Ballio, A., Gasinovi, C. & Serlupi-Grescensi, G. (1956)

Biochim, biophys. Acta, 20, 414.

X.

Beadle, G.W. (1957) in "The Chemical Basis of Heredity", p. 3 (Ed. by HeElroy, W.D. and Glass, B.). Baltimeres John Hepkins Press.

Beaven, G.H., Holiday, E.R. & Johnson, S.A. (1955) in "The Nucleic Acids", vol. 2, chapter 14 (Ed. by Chargaff, E. and Davidson, J.H.). New York: Academic Press.

Bock, W.S. & Lovin, N. (1962) Blochim. blophys. Acts, 55, 245.

Beers, R. F. jr. (1957) Biochem. J., 66, 686.

Behki, R.M. & Schneider, W.C. (1962) Biochim. biophys. Acts, 61, 663.

Behki, R.M. & Schneider, W.G. (1963) Biochim. biophys. Acta, 68, 34.

Bello, L.J. & Bessman, M.J. (1963a) J. Mol. Chem., 238, 1777.

Bello, L.J. & Bessman, M.J. (1963b) Blochim. blophys. Acta, 72, 647.

Bello, L.J., van Bibber, H.J. & Bessman, M.J. (1961) Biochim, biophys.

Acta, 53, 194.

Belts, R.K. (1962) Arch. Biochem. Biophys., 99, 304.

Bendich, A., Pahl, H., Korngold, G., Hosenkrans, H. & Fresco, J. (1958)

J. Amer. chem. Soc., 80, 3949.

Bently, M. & Abrams, R. (1956) Fed. Proc., 15, 218.

Berg, P. & Joklik, W.K. (1954) J. biol. Chem., 210, 657.

Bergkvist, R. (1956) Acta chem. Scand., 10, 1303.

Bertani, E., Haggmark, A. & Heichard, P. (1961) J. biol. Chem., 236, PC67. Bertani, E., Haggmark, A. & Reichard, P. (1963) J. biol. Chem., 238, 3407. Bessman, M.J. (1959) J. biol. Chem., 234, 2735.

Bessman, M.J. (1963) in "Molecular Genetics", part I, p. 1 (Ed. by

Taylor, J.H.). New Yorks Acedemic Press.

Bessman, M.J., Lehman, I.R., Simms, E.S. & Kornberg, A. (1958)

J. biol. Chem., 233, 171.

Bessman, M.J., Lehman, I.R., Adler, J., Zimmerman, S.B., Simms, E.S. & Kornberg, A. (1958) Proc. nat. Acad. Sci., Wash., 44, 633.

Bianchi, P.A. (1962) Biochim. biophys. Acta, 55, 547.

- Bianchi, P.A., Crathorn, A.R. & Shooter, K.V. (1962) Biochim. biophys. Acta, <u>61</u>, 728.
- Bianchi, P.A., Butler, J.A.V., Crathorn, A.R. & Shooter, K.V. (1961) Biochim. biophys.Acts, <u>53</u>, 123.

Biggs, H.G. & Doumas, H. (1963) J. biol. Chem., 238. 2469.

Billen, D. (1963) Biochim. biophys. Acta, 68, 342.

Birnie, G.D. (1959) Ph.D. Thesis, The University, Clasgow.

Birnstiel, M.L., Fleissner, E. & Borek, E. (1963) Science, 1/2, 1577.

Blakley, R.L. (1957) Biochim. biophys. Acta, 24, 224.

Bock, R.M., Ling, N.S., Morell, S.A. & Lipton, S.H. (1956) Arch. Biochem. Biophys., <u>62</u>, 253.

Bojarski, T.B. (1962) Fed. Proc., 21, 383.

Bojarski, T.B. & Hiatt, H.H. (1960) Nature, Lond., 188, 1112.

Bollum, F.J. (1958) J. Amer. chem. Soc., 80, 1766.

Bollum, F.J. (1959) J. biol. Chem., 234, 2733.

Bollum, F.J. (1963) in "Frogress in Nucleic Acid Research", vol. 1, p. 1

(Ed. by Davidson, J.H. and Cohn, W.E.).

New Yorks Academic Press.

Bollum, F.J. & Potter, V.R. (1958) J. blol. Chem., 233, 478. Bollum, F.J. & Potter, V.R. (1959) Cancer Research, 19, 561. Boxer, G.E. & Shonk, C.E. (1958) J. biol. Cham., <u>233</u>, 525. Breitman, T.R. (1963) Biochim. biophys. Acta, <u>67</u>, 153. Bresnick, E. (1963) Biochim. biophys. Acta, <u>67</u>, 425. Brown, G.B. & Roll, P.M. (1955) in "The Nucleic Acids", vol. 2, p. 341

(Ed. by Chargaff, E. and Davidson, J.N.).

New Yorkt Academic Press.

Brunn, A.F., Potter, V.R. & Siekevits, P. (1956) J. biol. Chem., 220, 713. Brenny, S., Jacob, F. & Meselson, H. (1961) Nature, Lond., <u>190</u>, 506. Brunnond, D.C., Stachelin, H. & Ochon, S. (1957) J. biol. Chem., <u>225</u>, 835. Buchanan, J.M., Senne, J.C. & Delluva, A.M. (1948) J. biol. Chem., <u>173</u>, 81. Buchanan, J.M., Flaks, J.G., Hartman, S.C., Levenberg, B., Lukens, L.M.

> & Warren, L. (1957) in "Biochemistry and Biology of Purines", p. 233 (Ed. by Wolstenholme, G.E.W. and O'Conner, C.M.).

Ciba Foundation Symposium. London: J. & A. Churchill Ltd. Burdon, R.H. & Smellie, R.M.S. (1961) Biochim. biophys. Acta, <u>51</u>, 153. Burdon, R.H. & Smellie, R.M.S. (1962) Biochim. biophys. Acta, <u>61</u>, 633. Butler, J.A.V., Laurence, D.J.A., Robins, A.B. & Shooter, K.V. (1959)

Proc. Boy. Soc., A, 250, 1.

Cairns, J. (1963) J. molec. Biol., 6, 208.

Canellakis, E.S. (1956) J. biol. Chem., 221, 315.

Canellakis, E.S. (1957a) Biochin. biophys. Acta, 25, 217.

Canellakis, S.S. (1957b) J. biol. Chem., 227, 79.

Canellakis, E.S. (1962) Annu. Rev. Biochen., 1, 271.

Canellakis, S.S. & Cohen, S.S. (1955) J. biol. Chem., 213, 379 and 385.

Canellakis, E.S., Tuttle, A.L. & Cohen, S.S. (1955) J. biol. Chem., 213, 397.

Canellakis, E.S., & Manstevinos, R. (1958) Biochim. biophys. Acta, 27, 643. Canellakis, E.S., Jaffe, J.J., Manstevinos, R. & Krakow, J.S. (1959)

J. biol. Chem., 234, 2096.

Carutto, R. (1951) J. biol. Chem., 189, 801.

Carter, C.E. & Cohen, L.F. (1955) J. Amer. chem. Soc., 77, 499.

Cavalieri, L.F. & Rosenberg, B.H. (1961) Biophys. J., 1, 317, 323, 337.

Cavalieri, L.F. & Rosenberg, B.H. (1962) Annu. Rev. Biochem., 3. 247.

Cavalieri, L.F., Deutsch, J. & Rosenberg, B.H. (1961) Biophys. J., 1, 301.

Chargaff, E. (1950) Experientia, 6, 201.

Chargaff, E. (1951) Fed. Iroc., 10, 654.

Chargaff, E. (1955) in "The Nucleic Acids", vol. 1, p. 307 (Ed. by

Chargaff, E. and Davidson, J.N.). New York: Academic Press. Chevremont, M. (1962) Biochem. J., 25. 25P.

Chiga, M. & Plaut, G.W.E. (1959) J. biol. Chem., 234, 3059.

Chiga, M. & Plaut, G.W.S. (1962) Biochim. biophys. Acta, 61, 736.

Chung, C.W. & Mahler, H.R. (1959) Blochem. blophys. Res. Commun., 1, 232.

Chung, C.W., Mahler, H.R. & Enrione, N. (1960) J. biol. Chem., 235, 1448.

Cohn, N. (1956) Biochim. biophys. Acta, 20, 92.

Cohn, W.E. (1950) J. Amer. chem. Soc., 72, 1471 and 2811.

Cohn, W.E. (1951) J. cell. comp. Physiol., 38, Suppl. 1, 21.

Cohn, W.E. (1960) J. biol. Chem., 235, 1488.

Cohn, W.E. & Volkin, E. (1953) J. biol. Chem., 203, 319.

Colowick, S.P. & Kalckar, H.H. (1943) J. biol. Chem., 148, 117.

Courtois, J.E. (1947) Biochim. biophys. Acts, 1, 270.

Grosbie, G. (1960) in "The Mucleic Acids", vol. 3, p. 323 (Ed. by

Chargaff, E. and Davidson, J.N.). NewYork: Academic Press. David, S. & Jaymond, J. (1958) Biochim. biophys. Acta, <u>30</u>, 433. Davidson, J.N. (1962) in "The Molecular Basis of Neoplasia", p.420.

Austin: University of Texas Press. Davidson, J.N. & Leslie, I. (1950) Nature, Lond., <u>165</u>, 49. Delamore, I.W. & Prusoff, W.H. (1962) Biochem. Pharmacol., <u>11</u>, 101. Defendi, V. & Manson, L.A. (1963) Nature, Lond., <u>198</u>, 359. Dickens, F. (1936) Biochem. J., <u>32</u>, 1615 and 1626. Dickens, F. & Williamson, D.H. (1956) Biochem. J., <u>64</u>, 567. Doty, P. (1961) Symp. biochem. Soc., <u>21</u>, 8. Doty, P., Boedtker, H., Fresco, J.R., Haselkorn, R. & Litt, M. (1959)

Proo, nat. Acad. Sth., Wanh., <u>45</u>, 482. Dunn, D.B. (1959) Blochim. blophys. Acta, <u>34</u>, 286. Dunn, D.B. (1961) Blochim. blophys. Acta, <u>46</u>, 198. Dunn, D.B. & Smith, J.D. (1958) Blochem. J., <u>66</u>, 627. Dunn, D.B., Smith, J.D. & Spahr, F.F. (1960) J. molec. Biol., <u>2</u>, 113. Edmonds, M., Delluva, A.H. & Wilson, D.W. (1952) J. biol. Chem., <u>197</u>, 251. Edmonds, M., Delluva, A.H. & Wilson, D.W. (1952) J. biol. Chem., <u>197</u>, 251. Edmonds, M., Delluva, A.H. & Wodel, A. (1936) Blochem. J., <u>30</u>, 1380. Elwyn, D. & Sprinson, D.B. (1950) J. biol. Chem., <u>184</u>, 1142. Elwyn, D. & Sprinson, D.B. (1954) J. biol. Chem., <u>207</u>, 467. Embden, G. & Zimmermann, E. (1927) Hoppe-Seyl. Z., <u>167</u>. 114 and 137. Ennis, H.L. & Lubin, M. (1963) Blochim. blophys. Acta, <u>68</u>, 78. Svans, E.A. & Stanford, F.G. (1963a) Nature, Lond., <u>197</u>, 551. Piala, S., Fiala, A., Tobar, G. & MoQuilla, H. (1962)

J. nat. Cancer Inst., 28, 1279.

Fink, K., Cline, R.A., Henderson, R.B. & Fink, R.B. (1956) J. biol. Chem., 221, 425.

Firshein, W. (1963) Biochem, biophys. Res. Commun., <u>11</u>, 187. Flaks, J.G. & Cohen, S.S. (1959) J. biol. Chem., <u>234</u>, 1501. Fleissner, E. & Borek, E. (1962) Proc. nat. Acad. Sei., Wash., <u>48</u>, 1199. Flodin, P. (1962) "Dextran Gels and their Application in Gel Filtration".

Uppeala: AB Pharmacia.

Fox, J.J. & Shugar, D. (1952) Biochim. biophys. Acta, 9, 369.

Friedkin, M. (1950) J. biel. Chem., 184. 437.

Priedkin, M. & Kalckar, H.M. (1961) in "The Enzymes", 2nd ed., vol. 5, p. 237 (Ed.by Boyer, P.D., Lardy, H. and Myrback, K.). New York: Academic Press.

Friedkin, M. & Kornberg, A. (1957) in "The Chemical Basis of Heredity", p. 600 (Ed. by MeElroy, W.D. and Glass, B.). Baltimore: John Hopkins Frees.

Friedkin, M. & Roberts, D. (1954) J. biol. Chem., 207, 245, 257. Friedkin, M. & Roberts, D. (1956) J. biol. Chem., 220, 653. Friedman, H.C. & Vennesland, B. (1958) J. biol. Chem., 213, 1398. Frydman, R.B., Neufeld, E.F. & Hassid, W.Z. (1963)

Biochim, biophys. Acta, <u>77</u>, 332. Furlong, N.B. (1963) Analyt. Biochem., <u>5</u>, 515. Furth, J.J., Hurwitz, J. & Goldmanp, M. (1961) Fed. Proc., <u>20</u>, 363. Gerhart, J.C. & Pardee, A.B. (1962) J. biel. Chem., <u>237</u>, 89. Ghosh, D. & Bernstein, I.A. (1963) Biochim. biophys. Acta, 72, 1.

Glock, G.E. (1955) in "The Mucleic Acids", vol. 2, p. 247 (Ed. by Chargeff, H

and Davidson, J.N.). New York: Academic Press.

Glock, G.E. & McLean, P. (1954) Blochen. J., 56, 171.

Gold, M., Hurwitz, J. & Anders, M. (1963a) Biochem. biophys. Res. Commun.,

11, 103.

Gold, M., Hurwitz, J. & Anders, M. (1963b) Proc. nat. Acad. Sci., Wash., <u>50</u>, 529.

Goldthweit, D.A. (1956) J. biol. Chem., 222, 1051.

Goldthweit, D.A., Peabody, R.A. & Greenberg, G.R. (1954)

J. Amer. chen. Soc., 76, 5258.

Goldthwait, D.A., Peabody, R.A. & Greenberg, G.R. (1956)

J. biol. Chem., 221, 555 and 569.

Goutier, R. (1964) Personal communication.

Gray, E.D., Weissman, S.H., Richards, J., Bell, D., Keir, H.M., Smellie, M.M.

& Davidson, J.N. (1960) Biochim. biophys. Acta, 45, 111.

Green, H. & Cohen, S.S. (1957) J. biol. Chem., 225. 387.

Greenberg, D.M., Nath, R. & Humphreys, G.K. (1961) J. Mol. Chen., 236, 2371.

Greenberg, G. R. (1948) Arch. Biochem., 19, 337.

Greenberg, G.R. (1950) Fed. Froc., 9, 179.

Greenberg, G.R. (1951a) Fed. Proc., 10, 192.

Greenberg, G.R. (1951b) J. biol. Cham., 190, 611.

Greenberg, G.R. (1954) Fed. Foc., 13, 745.

Greenberg, G.R. & Somervilla, R.L. (1962) J. biol. Chem., 237, 89.

Grunberg-Hanage, H. (1962) Annu, Rev. Blochem., 31, 301.

Grunberg-Manago, M. & Ochoa, S. (1955) J. Amer. chem. Soc., 77, 3165. Grunberg-Manago, M., Ortis, P.J. & Ochoa, S. (1956)

Biochia. biophys. Acta, 20, 269.

Guarino, A.J. & Sable, H.Z. (1955) J. Mol. Chem., 215, 515.

Guarino, A.J. & Turegir, G. (1959) Biochim. biophys. Acta, 36, 157.

Haggmark, A. (1963) Acta chem. Scand., 17, 890.

Hall, R.H. (1963) Biochen. biophys. Res. Commun., 13, 394.

Harbers, S. & Heidelberger, C. (1959) Blochin, biophys. Acta, 35, 361.

Harford, G.G. & Kornberg, A. (1958) Fed. Proc., 17, 515.

Hartman, S.G. & Buchanan, J.M. (1959) J. Mol. Chem., 234. 1812.

Hayaishi, 0. & Kornberg, A. (1952) J. biol. Chem., 197, 717.

Hecht, L.I. & Potter, V.E. (1956) Gane r Mesearch, 16, 999.

Hecht, L.I. & Potter, V.R. (1958) Gancer Research, 18, 186.

Hecht, L.I., Fotter, V.R. & Herbert, S. (1954) Blochim. blophys. Acta,

15, 134.

Hecht, L.I., Stephenson, M.L. & Zamecnik, P.G. (1959)

Proc. nat. Acad. Sci., Wash., 45, 505.

Hecht, L.I., Zamecnik, P.C., Stephenson, H.L. & Scott, J.F. (1958)

J. biol. Ches., 233. 954.

Heidelberger, C., Harbers, E., Liebaan, K.C., Takagi, Y. & Potter, V.K. (1956) Biochim. biophys. Acta, 20, 445.

Heinrich, M.R. & Wilson, D.W. (1950) J. biol. Chem., <u>186</u>, 447. Henderson, J.F. (1962) J. biol. Chem., <u>237</u>, 2631. Henderson, J.F. & LePage, G.A. (1958) Chem. Revs., <u>58</u>, 645.

Heppel, L.A. & Hilmoe, H.J. (1951) J. biol. Chem., 188, 665.

Heppel, L.A. & Hilmos, R.J. (1953) J. biol. Chem., 202. 217.

Heppel, L.A., Strominger, J.L. & Maxwell, E.S. (1959)

Biochim. biophys. Acta, 32, 422.

Herbert, E. (1958) J. biol. Chem., 231, 975.

Herebey, A.D. (1953) Cold Spr. Harb. Symp. quant. Biol., 18, 135.

Hershey, A.D. & Burgi, E. (1960) J. molec. Biol., 2, 143.

Hiatt, H.H. & Bojarski, T.B. (1960) Blochem. biophys. Res. Commun., 2, 35.

Hiatt, H.H. & Bejarski, T.B. (1961) Cold Spr. Harb. Symp. quant. Biol., 26, 367.

Hoagland, M.B. (1960) in "The Nucleic Anids", vol. 3, p. 349 (Ed. by

Chargaff, E. and Davidson, J.N.). New York: Academic Press.

Horecker, B.L. (1951) in "Phosphorus Metabolism", vol. 1, p. 117 (Ed. by McElroy, W.D. and Glass, B.).

Baltimore: John Hopkins Press.

Horecker, B.L. & Smyrnietis, P.Z. (1951) J. biol. Chem., <u>193</u>, 371. Horecker, B.L. & Smyrnietis, P.Z. (1952) J. Amer. chem. Soc., <u>74</u>, 2123. Horecker, B.L. & Smyrnietis, P.Z. (1953) Blochim. biophys. Acta, <u>12</u>, 98. Horecker, B.L. & Smyrnietis, P.Z. (1955) J. biol. Chem., <u>212</u>, 811. Horecker, B.L., Smyrnietis, P.Z. & Elenov, H. (1953)

J. biol. Chen., 205, 661.

Horseker, B.L., Smyrniotia, P.Z. & Seeguiller, J.E. (1951)

J. biol. Chem., 193, 383.

Horecker et el. (1956):- Smyrniotis, F.Z. & Homecker, B.L. (1956) J. Mel.
Chem., <u>218</u>, 745; Stumpf, P.K. & Horecker, B.L. (1956) J.
biol. Chem., <u>218</u>, 753; Eurwitz, J., Weissbach, A.,
Horecker, B.L. & Smyrniotis, P.Z. (1956) J. Mol. Chem.,
<u>218</u>, 769; Horecker, B.L., Hurwitz, J. & Weissbach, A.
(1956) J. Mol. Chem., <u>218</u>, 785; Weissbach, A.,
Horecker, B.L. & Hurwitz, J. (1956) J. Mol. Chem., <u>218</u>, 795

Hotta, Y. & Stern, H. (1961) J. blochem. blophys. Cytol., <u>11</u>, <u>31</u>. Hotta, Y. & Stern, H. (1963) Proc. nat. Acad. Sci., Wash., <u>49</u>, 648 and 861. Hurlbert, R.B. & Potter, V.R. (1952) J. biol. Chen., <u>195</u>, 257. Hurlbert, R.B. & Meichard, F. (1955) Acta chem. Scand., <u>9</u>, 251. Hurlbert, R.B., Schmitz, H., Brumm, A.F. & Potter, V.R. (1954)

J. biol. Chem., 209. 23.

Hurwits, J. (1959) J. biol. Cham., 234, 2351.

Hurwitz, J. & Horecker, B.L. (1956) J. biol. Chem., 223, 993.

Hurwitz, J., Bresler, A.E. & Diringer, R. (1960)

Biochem. biophys. Res. Commun., 3, 15. Hurwitz, J., Bresler, A.E. & Kaye, A. (1959)

Biochen. biophys. Res. Commun., 1, 3.

Ives, D.H., Morse, P.A.jr. & Potter, V.R. (1962) Fed. Proc, 21, 383. Ives, D.H., Morse, P.A.jr. & Potter, V.R. (1963) J. biol. Chem., 238, 1467. Jacob, F. & Monod, J. (1961) J. molse. Biol., 3, 318. Jaenicke, L. (1956) Fed. Proc., 15, 281. Joklik, W.K. (1962) Cold Spr. Harb. Symp. quant. Biol., 27, 199. Jones, M.E., Spector, L. & Lipman, F. (1955) J. Amer. chem. Soc., 77, 819. Josse, J., Kaiser, A.D. & Kornberg, A. (1961) J. biol. Chem., 236, 864. Kahan, F.M. & Hurwitz, J. (1962) J. biol. Chem., 237, 3778. Kalekar, H.M. (1943) J. biol. Chem., 141, 127. Kalekar, H.M. (1947) J. biol. Chem., 167, 477. Kalina, M. & Avigad, G. (1963) Biochim. biophys. Acta, 73, 652. Kamen, H.O. & Hurlbert, A.B. (1959) Cencer Research, 19, 654. Keir, H.M. (1962) Biochem. J., <u>85</u>, 265. Keir, H.M. & Smellie, R.M.S. (1959) Biochim. biophys. Acta, <u>35</u>, 405. Keir, H.M. & Smellie, R.M.S. (1963) Biochim. biophys. Acta, <u>68</u>, 589. Keir, H.M., Smallie, R.M.S. & Siebert, G. (1962) Hature, Lond., <u>196</u>, 572. Kellenberger, E. (1960) in "Microbial Genetics", p. **39**, 10th Symp.

Soc. Gen. Microbiol. Cambridge University Press. Mielley, R.K. (1961) Biochim. biophys. Acta, 53, 150. Mielley, R.K. (1963a) Biochem. biophys. Res. Commun., 10, 249. Mielley, R.K. (1963b) Gancer Research, 23, 801. Mit, S. & Dubbs, D.R. (1963) Biochem. biophys. Res. Commun., 11, 55. Mit, S., Piekarski, L.J. & Dubbs, D.H. (1963) J. molec. Biol., 6, 22. Mienow, H. (1962) Biochim. biophys. Acta, 61, 885. Mienow, H. & Lichtler, E. (1957) Biochim. biophys. Acta, 23, 6. Korn, E.D. & Buchanan, J.M. (1955) J. biol. Chem., 217, 183. Kornberg, A. (1957) in "The Chemical Basis of Heredity", p. 579 (Ed. by McElrey, W.D. and Glass, B.). Baltimore: John Hopkins Press.

Kornberg, A. (1960) Science, <u>131</u>, 1503. Kornberg, A. (1962) "Enzymatic Synthesis of DNA". New York: John Wiley & Sc

xxi.

Kornberg, A., & Pricer, W.S. jr. (1951) J. bdol. Chem., <u>193</u>, 481. Kornberg, A., Lehman, I.A. & Simme, S.S. (1956) Fed. Proc., <u>15</u>, 291. Kornberg, A., Lieberman, I. & Simme, E.S. (1954)

J. Amer. chem. Soc., 76, 2027.

Kornberg, A., Lieberman, I. & Simms, E.S. (1955a) J. blol. Chem., 215, 389. Kornberg, A., Lieberman, I. & Simms, E.S. (1955b)

J. biol. Chem., 215, 403 and 417.

Kornberg, A., Zimmerman, S.B., Kornberg, S.A. & Josse, J. (1959)

Proc. nat. Acad. Sci., Wash., 15, 272.

Kornfeld, S. & Glaser, L. (1961) J. biol. Chem., 236, 1791.

Korafeld, S. & Glaser, L. (1962) J. biol. Chem., 237, 3052.

Krakow, J.S., Coutsogeorgopoulos, C. & Canellakis, E.S. (1961)

Biechim. biophys. Acta, 55, 639.

Krebs, H.A. & Hems, R. (1953) Blochim. blophys. Acta, 12, 172.
Kuby, S.A., Mahowald, T.A. & Noltmann, E.A. (1962) Blochemistry, 1, 749.
Kupiecki, F.P. & Goon, M.J. (1957) J. blol. Chem., 229, 743.
Kuriki, T. & Okasaki, R. (1959) Embryologia, 4, 337.
Lagerkvist, U. (1953) Arkiv Kemi, 5, 569.
Lagerkvist, U. (1955) Acta chem. Scand., 9, 1028.
Lajtha, L.G., Oliver, R. & Ellis, F. (1954) Brit. J. Cancer, 8, 387.
Langen, P. & Liss, E. (1962) Blochem, Z., 336, 139.
Langer, L. & Klenov, H. (1960) Blochim. blophys. Acta, 37, 33.
Lanning, M.C. & Cehen, S.S. (1955) J. blol. Chem., 216, 413.

Lark, K.G. (1960) Biochim. biophys. Acta, 45, 121. Lark, K.G. (1963) in "Molecular Genetics", part 1, p. 153

(Ed. by Taylor, J.H.). New York: Academic Press. Lark, K.G. & Maalge, O. (1956) Blochim. biophys. Acta, <u>21</u>, 448. Larsson, A. (1963) J. biol. Chem., <u>238</u>, 3414. Lehman, I.A. (1959) Ann.N. Y. Acad. Sci., <u>81</u>, 745.

Lehman, I.A., Bessman, M.J., Simms, E.S. & Kornberg, A. (1958)

J. biol. Chem., 233. 163.

LePage, G.A. (1957) J. biol. Chem., 226, 135.

Levenberg, B. & Buchanan, J.N. (1957a) J. biol. Chem., 224, 1005. Levenberg, B. & Buchanan, J.M. (1957b) J. biol. Chem., 224, 1019. Levens, P.A. & Jacobs, W.A. (1909) Ber. dtsch. chem. Ges., 42, 1198. Levens, P.A., Mikeska, C.A. & Mori, T. (1930) J. biol. Chem., <u>85</u>, 785. Levinthal, C. & Davison, P.F. (1961) J. molec. Biol., <u>3</u>, 674. Lichtenstein, J., Barner, H.D. & Cohen, S.S. (1960) J. biol. Chem., <u>235</u>, 457 Lieberman, I. (1956a) J. Amer. chem. Soc., <u>78</u>, 251. Lieberman, I. (1956b) J. biol. Chem., <u>222</u>, 765. Lieberman, I. & Kornberg, A. (1954) J. biol. Chem., <u>207</u>, 911. Lieberman, I., Kornberg, A. & Simms, E.S. (1955)

J. biol. Chem., 215, 389, 403 and 429. Littauer, U.Z. & Kornberg, A. (1957) J. biol. Chem., <u>226</u>, 1077. Littlefield, J.W., McGovern A.P. & Margeson, K.B. (1963)

Proc. nat. Acad. Sci., Wash., 49, 102. Long, C. (1955) Biochem. J., 19, 322.

xxiii.

Lovenstein, J.M. (1958) Biochin, biophys. Acta, 28, 206. Lowry, O.H., Rosebrough, N.G., Farr, A.L. & Randall, R.J. (1951) J. biol. Chem., 183, 265. Lukens, L.N. & Buchanan, J.M. (1957) J. Amer. chem. Soc., 79, 1511. Maalse, O. (1963) J. cell. comp. Physiol., 62, Suppl. 1, 131. MacNutt, N.S. (1952) Biochem. J., 50, 384. MacNutt, W.S. (1958) J. biol. Chem., 233, 193. Magasanik, B. & Karibian, D. (1960) J. biol. Chem., 235, 2672. Magee, N.E. (1962) Virology, 17, 604. Mager, J. & Magasanik, B. (1960) J. biol. Chem., 235, 1474. Maley, G.F. & Maley, F. (1959) J. biol. Chem., 234, 2975. Maley, G.F. & Maley, F. (1960) J. biol. Chem., 235, 2968. Haley, G.F. & Maley, F. (1961a) Biochem. biophys. Res. Commun., 5, 439. Maley, G.F. & Maley, F. (1961b) Cancer Research, 21, 1421. Maley, G.F. & Maley, F. (1961c) J. biel. Chem., 236, 1806. Maley, G.F. & Maley, F. (1962a) Fed. Proc., 21, 382. Haley, F. & Maley, G.F. (1962b) Biochemistry, 1, 847. Maley, G.F. & Maley, F. (1963) Biochim. biophys. Acta, 68, 293. Maley, F. & Ochoa, S. (1958) J. biel. Chem., 233, 1538. Handel, L.R. & Borek, E. (1961) Biocham, biophys. Res. Commun., 6, 138. Mandel, L.R. & Borek, S. (1963a) Biochemistry, 2, 555. Mandel, L.R. & Borek, B. (1963b) Blochamistry, 2, 560. Manson, L.A. & Lampen, J.O. (1951) J. Mol. Chem., 191, 95. Manstavinos, R. & Canellakis, E.S. (1959) J. biol. Chem., 234, 628. Manstavinos, R. & Zamenhof, S. (1961) J. biol. Chem., 236, 876.
McAuslan, B.K. (1963) Virology, 20, 162.

McAuslan, B.R. & Joklik, W.K. (1962) Biochem. biophys. Sec. Commun., <u>B</u>, 486. Meselson, M. & Stahl, F.W. (1958) Proc. nat. Acad. Sci., Wash., <u>44</u>, 671. Michelson, A.M. & Todd, A.R. (1953) J. chem. Soc., p. 951. Miller, G.S., Gurin, S. & Wilson, D.W. (1950) Science, <u>112</u>, 654. Miller, R.W., Lukens, L.N. & Buchanan, J.M. (1957)

J. Amer. chem. Soc., 79, 1513.

Mitchell, H.K., Houlahan, M.B. & Nyc, J.F. (1948) J. biol. Chem., <u>172</u>, 525.
Monod, J., Changeux, J.-P. & Jacob, F. (1963) J. molec. Biol., <u>6</u>, 306.
Monod, J., Jacob, F. & Gros, F. (1962) Symp. biochem. Soc., <u>21</u>, 104.
Moore, E.C. & Hurlbert, R.B. (1962) Biochim. biophys. Acta, <u>55</u>, 651.
Moore, E.C. & Reichard, P. (1963) J. biol. Chem., <u>238</u>, FC2244.
Horris, N.R. & Fischer, G.A. (1960) Biochim. biophys. Acta., <u>42</u>, 184.
Horris, N.R. & Fischer, G.A. (1963) Biochim. biophys. Acta, <u>68</u>, 82.
Morris, N.R., Meichard, P. & Fischer, G.A. (1963)

Blochim. blophys. Acta, <u>68</u>, 93.

Nagata, T. (1963) Proc. nat. Acad. Sci., Wash., <u>49</u>, 551. Nakamoto, T. & Weiss, S.B. (1962) Proc. nat. Acad. Sci., Wash., <u>48</u>, 880. Nass, M.M.K. & Nass, S. (1963a) J. Cell Biol., <u>19</u>, 593. Nass, S. & Nass, MM.K. (1963b) J. Cell Biol., <u>19</u>, 613. Noda, L. (1962) in "The Ensymes", 2nd ed., vol. 6, p. 139 (Ed. by Boyer, P.D

Lardy, H. and Myrback, K.). New York: Academic Press. Nohara, H. & Kaplan, A.S. (1963) Biochem. biophys. Res. Commun., <u>12</u>, 189. Ochoa, S. & Heypel, L.A. (1957) in "The Chemical Basis of Heyedity", p. 615

(Ed. by McElroy, W.D. & Glass, B.). Haltimore:

John Hopkins Press.

XX¥.

Proc. nat. Acad. Sci., Wash., 47, 670.

Ochoa, S., Burma, D.B., Kroger, H. & Weill, J.D. (1961)

Okazaki, R. (1959) Biochem. biophys. Res. Commun., 1, 34. Okazaki, R. & Kornberg, A. (1964a) J. biol. Chem., 239, 269. Okazaki, R. & Kornberg, A. (1964b) J. biol. Chem., 239, 275. Okazaki, R., Okazaki, T. & Kuriki, Y. (1959) Biochim. biophys. Acta, 33, 289 Okazaki, R., Okazaki, T., Strominger, J.L. & Michelson, A.M. (1962)

J. biol. Chem., 237. 3014.

Okazaki, T., Strominger, J.L. & Okazaki, R. (1963) J. Bacteriol., <u>86</u>, 118. Ord, M.G. & Stocken, L.A. (1958) Biochim. biophys. Acta, <u>29</u>, 201. Pardee, A.B. & Wilson, A.C. (1963) Cancer Research, <u>23</u>, 1483. Paterson, A.R.P. & LePage, G.A. (1957) Cancer Research, <u>17</u>, 409. Pasur, J.H. & Anderson, J.S. (1963) J. biol. Chem., <u>236</u>, 3155. Pasur, J.H. & Shuey, E.W. (1961) J. biol. Chem., <u>236</u>, 1780. Pontis, H., Degerstedt, G. & Reichard, P. (1961)

Biochin, biophys. Acta, 51, 138.

Potter, R.L. & Nygaard, O.F. (1963) J. biol. Chem., 238, 2150. Potter, R.L., Schlesinger, S., Buettner-Janusch, V. & Thompson, L. (1957)

J. blol. Chem., 226, 881.

Potter, V.R. (1960) "Nucleic Acid Outlines", vol. 1, chapter 9. Minneapolis: Burgess Publishing Company.

Potter, V.R. (1962) in "The Molecular Basis of Neoplasia", p. 367. Austin: The University of Texas Press.

Potter, V.R. (1963) Exp. Cell Res., Suppl. 9, 259.

Potter, V.R. & Auerbach, V.H. (1959) Lab. Invest., 8, 495.

Potter, V.R. & Elvehjelm, C.A. (1936) J. biol. Chem., 114, 495. Prescott, D.M. (1960) Exp. Cell Res., 19, 228. Racker, E. (1951) Nature, Lond., 167, 408. Racker, E. (1952) J. biol. Chem., 196, 347. Ratliff, R.L., Weaver, R.H., Lardy, H.A. & Kuby, S.A. (1964)

J. biol. Chem., 239, 301.

Heichard, P. (1949) Acta chem. Scand., j, 422.
Reichard, P. (1954) Acta, chem. Scand., g, 795 and 1102.
Reichard, P. (1955) Acta chem. Scand., g, 1275.
Reichard, P. (1957) Acta chem. Scand., 11, 523.
Reichard, P. (1958) Biochim. biophys. Acta, 27, 434.
Reichard, P. (1960) Biochim. biophys. Acta, 41, 368.
Reichard, P. (1962) J. biol. Chem., 237, 3513.
Reichard, P. & Estborn, B. (1951) J. biol. Chem., 188, 839.
Reichard, P. & Hanshoff, G. (1956) Acta chem. Scand., 10, 548.
Reichard, P. & Lagerkvist, U. (1953) Asta chem. Scand., 7, 1207.
Reichard, P., Baldesten, A. & Rutberg, L. (1961) J. biol. Chem., 236, 1150.
Reichard, P., Canellakis, Z.N. & Canellakis, E.S. (1961)

J. bdol. Chem., 236, 2514.

Hemy, C.H., Remy, M.J. & Buchanan, J.M. (1955) J. biol. Chem., <u>217</u>. 685.
Rice, S.A. & Doty, P. (1957) J. Amer. chem. Soc., <u>79</u>, 3937.
Roll, P.M., Weinfeld, R. & Carroll, E. (1956) J. biol. Chem., <u>220</u>, 455.
Rose, J.A. & Schweigert, B.S. (1953) J. biol. Chem., <u>202</u>, 635.
Roth, J.S. (1963) J. Theoret. Biol., <u>4</u>, 113.

Rotherham, J. & Schneider, W.C. (1958) J. biol. Chem., 232. 853. Rubenstein, I., Thomas, C.A. jr. & Hershey, A.D. (1961)

Proc. nat. Acad. Sci., Wash., 47, 1113.

Scarano, E. (1958) Biochim. biophys. Acta, 29, 459.

Scarano, B. (1960) J. biol. Chem., 235, 706.

Scarane, S. & Maggio, R. (1959) Arch. Bloches. Biophys., 79, 392.

Scarane, E., Bonaduce, L. & DePetrocellis, B. (1960)

J. biol. Chem., 235, 3556.

- Schachman, H.K., Adler, J., Radding, G.M., Lehman, I.R. & Kernberg, A. (1960 J. biol. Chem., 235, 3242.
- Schachman, H.K., Lehman, I.R., Bessman, M.J., Adler, J., Simms, E.S. & Kornberg, A. (1958) Fed. Proc., 17, 304.
- Schive, W., Ackerman, W.W., Gerdon, M., Getsendaner, M.E. & Sakin, R.E. (1947) J. Amer. chem. Soc., <u>69</u>, 725.

Schmidt, G. (1955) in "The Bucleic Acids", vol. 1, p. 555 (Ed. by

Chargaff, E. and Davidson, J.N.). New York: Academic Press. Schmitz, H. (1954) Biochem. Z., 325, 555.

Schmitz, H., Hurlbert, R.B. & Potter, V.R. (1954) J. biol. Chem., 209, 41. Schmeider, W.C. (1955) J. biol. Chem., 216, 287.

Schneider, W.C. & Hotherham, J. (1961) J. biol. Chem., 236, 2764.

Schulmann, M.P. & Buchanan, J.M. (1952) J. Mol. Chem., 196, 513.

Scott, D.B.M. & Cohen, S.S. (1951) J. cell, comp. Physiol., 38, Suppl. 1, 17 Scott, D.B.M. & Cohen, S.S. (1953) Biochem. J., 55, 33.

Seegniller, J.S. & Horecker, B.L. (1952) J. biol. Chem., 194. 261.

Shemin, D. & Rittenberg, D. (1947) J. biel. Chem., 167, 875.

Shooter, E.M. & Baldwin, R.L. (1962) Fed. Proc., 21, 374. Siebert, G. (1963) Exp. Cell Res., Suppl. 2, 389. Singer, M.F., Heppel, L.A., Ochoa, S. & Mii, S. (1959)

Proc. 3rd Canadian Cancer Conference, p. 41. Singer, M.F., Heppel, L.A. & Hilmoe, R.J. (1960) J. biol. Chem., 235, 738. Skold, O. (1960a) J. biol. Chem., 235, 3273. Skold, O. (1960b) Biochim. biophys. Acts, <u>44.</u> 1. Smellie, R.M.S. (1963) in "Progress in Nucleic Acid Research", vol. 1, p. 27

(Ed. by Davidson, J.N. and Cohn, W.E.).

New York: Academic Press.

Smellie, R.M.S., Keir, H.M. & Davidson, J.N. (1959)

Blochim. blophys. Acta, 35, 389.

Smellie, R.M.S., Gray, E.D., Keir, H.M., Richards, J., Bell, D. &

Davidson, J.N. (1960) Biochim. biophys. Acta, 37, 243.

Smith, M. & Kherana, H.G. (1958) J. Amer. chem. Soc., 80, 1148.

Smith, S.M.J. & Keir, H.M. (1963) Biochim. biophys. Acta, 68, 978.

Somerville, R.L., Ebisunski, K. & Greenberg, G.M. (1959)

Proc, nat. Acad. Sci., Wash., 45, 1250.

Sonne, J.C., Buchanan, J.M. & Delluva, A.H. (1948) J. biol. Chem., <u>173</u>, 69. Sonne, J.C., Lin, I. & Buchanan, J.M. (1953) J. Amer. chem. Soc., <u>75</u>, 1516. Sonne, J.G., Lin, I. & Buchanan, J.M. (1956) J. biol. Chem., <u>220</u>, <u>369</u>. Srinivasan, P.R. & Borek, E. (1963) Proc. nat. Acad. Soi., Wash., <u>49</u>, 529. Stern, H. (1961) J. biophys. biochem. Cytol., <u>9</u>, 271. Stetten, M.R. & Fox, C.L. (1945) J. biol. Chem., <u>161</u>, <u>333</u>. Stevens, A. (1961) J. biol. Chem., 236, PC43.

Straus, D.B. & Goldwasser, E. (1961) J. biol. Chem., 236. 849.

Strominger, J.L., Heppel, L.A. & Maxwell, E.S. (1954)

Arch. Blochen. Biophys., 54, 488.

Stroninger, J.L., Heppel, L.A. & Maxwell, E.S. (1959)

Biochin, blophys. Acta, 32, 412.

Svensson, I., Boman, H.G., Eriksson, K.G. & Kjellin, K. (1963) J. molec. Biol., 7, 254.

Taylor, J.H. (1963) in "Molecular Genetics", part 1, p. 65.

(Ed. by Taylor, J.H.), New York: Academic Press. Tener, G.M. (1961) J. Amer. chem. Soc., <u>83</u>, 159. Terasima, T. Tolmach, L.J. (1963) Exp. Cell Res., <u>30</u>, 344.

Thomas, G.A. jr. (1963) in "Molecular Genetics", part 1, p. 113

(Ed. by Taylor, J.H.). New York: Academic Press.
Totter, J.R. (1954) J. Amer. chem. Soc., <u>76</u>, 2196.
Vendrely, R. & Vendrely, G. (1956) Int. Rev. Gytol., <u>5</u>, 171.
de Verdier, C.-H. & Gould, B.J. (1963) Biochim. biophys. Acta, <u>68</u>, 333.
de Verdier, C.-H. & Potter, V.R. (1960) J. nat. Gancer Inst., <u>24</u>, 13.
Wahba, A.J. & Friedkin, M. (1962) J. biol. Chem., <u>237</u>, 3794.
Wang, T.F. & Lampen, J.O. (1952) J. biol. Chem., <u>194</u>, 784.
Warburg, O., Christian, W. & Griese, A. (1935) Biochem. Z., <u>282</u>, 157.
Warren, L. & Flaks, J.G. (1956) Fed. Froc., <u>15</u>, 379.
Warren, L., Flaks, J.G. & Buchanan, J.M. (1957) J. biol. Chem., <u>229</u>, 627.
Watcon, J.D. & Crick, F.H.C. (1953n) Nature, Lond., <u>171</u>, 735 and 964.

Watson, J.D. & Grick, F.H.C. (1953b) Cold. Spr. Harb. Symp. quant. Biol., 18. 123.

Weed, L.L. (1951) Cancer Research, 11, 470.

Weiss, S.B. & Nakamoto, T. (1961a) J. biol. Chem., 236, PC18.

Weissman, S.H., Smellie, R.M.S. & Paul, J. (1960)

Biochin, biophys, Acta, 45, 101.

Whitfield, P.R., Heppel, L.A. & Markhan, R. (1955) Blochem. J., <u>60</u>, 15. Wilkins, M.H.F. (1957) Symp. blochem. Soc., <u>14</u>, 13. Willstatter, R. & Kraut, H. (1923) Ber. dtsch. chem. Ges., <u>56</u>, 1117. Wright, L.D., Miller, G.S., Skegge, H.R., Huff, J.W., Weed, L.L. &

Wilson, D.W. (1951) J. Amor. chem. Soc., <u>73</u>, 1898. Wu, R. & Wilson, D.W. (1956) J. biol. Chem., <u>223</u>, 195.

Wyatt, G.R. (1950) Nature, Lond., 170, 237.

Wyatt, G.R. (1951) Biochem. J., 18, 584.

Wyatt, G.R. & Cohn, W.E. (1952) Nature, Lond., 170, 1072.

Wyngnarden, J.B. & Ashton, D.M. (1959) J. Mol. Chem., 234, 1494.

Yates, R.A. & Pardee, A.B. (1956a) J. biol. Chem., 221, 743.

Yates, R.A. & Pardse, A.B. (1956b) J. biol. Chem., 221, 957.

Yoshikawa, H. & Sueoka, N. (1963) Proc. aat. Acad. Sci., Wash., 49, 806.

Zamecnik, P.C., Stephenson, M.L. & Hecht, L.I. (1958)

Proc. nat, Acad. Sci., Wash., 44, 73.

Zilverman, M. (1962) Biochem. biophys. Res. Commun., §, 169. Zimmerman, M. (1963) Fed. Proc., 22, 291.